Spectra of Indefinite Linear Operator Pencils



Hasen Mekki Öztürk Department of Mathematics and Statistics University of Reading

A thesis submitted for the degree of Doctor of Philosophy

June 2019

Abstract

In recent years, there has been a rapid growth of interest in spectral properties of non-self-adjoint operators and operator pencils. This thesis is concerned with indefinite self-adjoint linear pencils which lead to a special class of non-self-adjoint spectral problems. These problems are not well understood, and, in general, many sign-indefinite problems which are trivial to state require some highly non-trivial analysis.

We look at indefinite linear pencil problems from the perspective of a two parameter eigenvalue problem. We derive localisation results for real eigenvalues and present several examples. We also use different approaches to obtain estimates of non-real eigenvalues, supported by a large number of numerical experiments. Additionally, these experiments lead to various open questions and conjectures.

Declaration

I confirm that this is my own work and that the use of all material from other sources has been properly and fully acknowledged.

Hasen Mekki Öztürk June 2019

Acknowledgements

I would like to thank my supervisor Michael Levitin for his time and effort during instructive meetings on my research. Without his help and advice, this thesis would not have been produced. Alongside my supervisor, I would like to express my thanks to Sabine Bögli for useful discussions. I am also grateful to Simon Chandler-Wilde and Lyonell Boulton for useful suggestions.

I gratefully acknowledge the financial support received from the Ministry of National Education of the Republic of Turkey. Without their financial support throughout my study, I would not have been able to undertake this research. I would also like to thank the Department of Mathematics and Statistics for the funding I have received.

I would like to thank all members of the Department of Mathematics and Statistic at University of Reading. I am extremely grateful to Jani Virtanen and Titus Hilberdink for their remarkable support through some difficult times. I am thankful for the emotional support and encouragement, in particular, to Peta-Ann King, Ruth Harris, Brigitte Calderon, Kris Aldridge, Birzhan Ayanbayev, Maha Kaouri, Samuel Groth, Andrew Gibbs, Oliver Sutton, Kasia Kozlowska, György Geher, Nicola Thorn, Jemima Tabeart, James Jackaman, Emily Roashan, Dan Derrick, Chris Davies, Philip Maybank, Julia Abery. They have made the past few years much more interesting.

I owe a great deal of thanks to Turgut Anıl Karlıdağ, Ömür Özdemir, Oğuzhan Hazır, Nizamettin Akbulut, Aliya and Yerlik Ayanbayev, and many other friends who made my life so special. I am very grateful to have them in my life.

Finally, but most importantly, I would like to thank my parents and siblings, Hıfzı, Hacıkadın, Mehmed Zahid and Saliha Öztürk for their endless love, support and encouragement.

Contents

1	Introduction							
	tor pencils	3						
	1.2	Indefir	nite linear pencils	4				
	1.3	Key re	esults and outline of the thesis	6				
2	Background							
	2.1	Block	matrices	16				
	2.2	Gersh	gorin circles	20				
		2.2.1	Inclusion regions for the eigenvalues of matrices	20				
		2.2.2	Exclusion regions for the eigenvalues of matrices	23				
		2.2.3	Inclusion regions for the eigenvalues of a linear pencil	24				
		2.2.4	Minimal Gershgorin sets	25				
		2.2.5	Gershgorin sets for block matrices	26				
		2.2.6	A Gershgorin-type result for the spectrum of a two-parameter eigen-					
			value problem	27				
	2.3	2.3 Orthogonal polynomials		28				
	2.3.1 Properties of the Chebyshev polynomials of the second kind							
		2.3.2	Extension of Chebyshev polynomials	32				
		2.3.3	Spectral properties of $H_0^{(n)}$	33				
		2.3.4	Ratios of Chebyshev polynomials of the second kind	35				
3	A tv	vo-para	meter eigenvalue problem for a class of block-operator matrices	38				
	3.1	1 Basics from a linear pencil problem						
	3.2	Introduction to a two-parameter eigenvalue problem						
	3.3 Basics and statements							
		3.3.1	Restrictions and notation	46				
		3.3.2	Statement of the simple Chess Board Theorem	48				
		3.3.3	Statement of the full Chess Board Theorem	49				

		3.3.4	Limit cas	ses					
	3.4	Auxilia	ry results						
	3.5	Proofs	of the main results						
	3.6	Examp	oles						
		3.6.1	Example	91					
		3.6.2	Example	92					
		3.6.3	Example	93					
		3.6.4	Example	94					
		3.6.5	Example	95					
		3.6.6	Example	96					
	3.7	The no	on-real pa	ir-eigenvalues of $\mathbf M$					
		3.7.1	Limiting	cases					
		3.7.2	Example	es					
4	Bou	nds for	r the eige	envalues of A_c 76					
	4.1	Differe	ent repres	entation of the operator pencil \mathcal{A}_c					
		4.1.1	Explicit e	expressions for eigenfunctions of A_c					
		4.1.2	Mapping	z + 1/z					
		4.1.3	\mathcal{A}_c in the	basis of eigenfunctions of H_0					
	4.2	Reform	nulating Conjecture 1.3.3 in different forms						
		4.2.1	Ratios of orthogonal polynomials						
		4.2.2	The area between $ z = 1$ and $ z \pm i ^2 < 2$						
		4.2.3	Intersection of Gershgorin disks 8						
	4.3	Partial	l cases						
		4.3.1	The case $n = 2$						
		4.3.2	.3.2 The case $n = 3$						
	4.4	Forma	l asymptotic expansion as $c \rightarrow 0$						
	4.5	Improv	ved bounds and ways to bound						
		4.5.1	Block-structure estimates						
		4.5.2	Bounds on the ratios of Chebyshev polynomials of the second kind . 1						
		4.5.3	Gershgorin-type localisation						
			4.5.3.1	Application of known results to the pencil problem 115					
			4.5.3.2 Gershgorin-type region for the two-parameter eigenvalue						
				problem					
			4.5.3.3	Gershgorin circles under the mapping $z + 1/z$					

5	Num	nerics and heuristics	126				
	5.1	Dynamics of the eigenvalues of the pencil A_c	126				
	5.2	Heuristics for the non-real pair-eigenvalues of ${\bf M}$	127				
	5.3	Double eigenvalues of A_c	130				
Bibliography 1							

Chapter 1

Introduction

Spectral theory is one of the richest fields in mathematics and one can find its various applications in classical as well as quantum physics, such as in the study of control theory [35], the physics of musical instruments [24, 33], quantum mechanics [39] and signal processing [57], see [74] for more details.

The spectral theory of linear self-adjoint operators was first presented by D. Hilbert in his famous papers (1904-1910). Following this, the abstract concepts of Hilbert spaces and linear operators were developed by von Neumann whose work was used to build quantum mechanics, which was created by Jordan, Schrödinger, Dirac, and others. Since then, self-adjoint operators have been extensively studied in the literature (see [69,75] for a detailed survey). This was in part due to the needs of quantum mechanics, where fundamental properties such as momentum, position, and the Hamiltonian are expressed in this form. Today, many books can be found in this field (e.g. [13,21]).

On the other hand, G. D. Birkhoff initiated the basics on eigenfunction expansions for non-self-adjoint operators between 1908 and 1913. Although his works were written during the same period as Hilbert's papers, the first abstract results in this field appeared in the literature in 1951, written by M. V. Keldys. There are many authors who made important contributions to the field, including Kato, Wolf, Gohberg, Krein and Langer. For more detailed historical information, see [16, 20, 75] and references therein. Non-self-adjoint operators have a wide range of physical applications, they arise, for instance, in the study of biological systems [59], in the exploration of resonance phenomena [56], describing the motion and oscillation of viscous fluids [36], the optical singularities of crystals [9], the optical model of nuclear scattering [8], and others.

Hermitian systems describe time-dependent systems which conserve the energy, while non-Hermitian systems do not. Examples of such non-Hermitian systems appear, for example, in friction problems and inelastic scattering problems. For recent advances and open problems in quantum physics related to non-self-adjoint operators, see [6, 7, 30]. Although the theory of non-self-adjoint operators is still under development, it already has a variety of applications in different fields, and the theory is far from complete.

This thesis will focus on self-adjoint linear operator pencils which lead to a class of non-self-adjoint operators. Before going into the details of this relation, the general theory of operator pencils will be discussed in the next section. Throughout the thesis, \mathcal{H} denotes a finite-dimensional, complex Hilbert space, equipped with inner product $\langle \cdot, \cdot \rangle$ such that $\langle a\mathbf{u}, b\mathbf{v} \rangle = a\overline{b}\langle \mathbf{u}, \mathbf{v} \rangle$, $\mathbf{u}, \mathbf{v} \in \mathcal{H}$, $a, b \in \mathbb{C}$. We will be working with operators acting on \mathcal{H} , and therefore they are bounded. The identity operator on \mathcal{H} will be denoted by I. We denote by $\mathcal{B}(\mathcal{H}_1, \mathcal{H}_2)$ the space of all bounded operators from \mathcal{H}_1 to \mathcal{H}_2 . If $\mathcal{H}_1 = \mathcal{H}_2 = \mathcal{H}$, then we use the short form $\mathcal{B}(\mathcal{H}) = \mathcal{B}(\mathcal{H}, \mathcal{H})$. Capital bold letters (e.g. **X**) represent block matrices, small bold letters (e.g. **u**) represent vectors. A^T and A^* represent the transpose and the adjoint of A, respectively. We denote by $\operatorname{diag}(\lambda_1, \ldots, \lambda_m)$ a diagonal matrix with entries $\lambda_1, \ldots, \lambda_m$.

1.1 Operator pencils

Operator pencils (or *polynomial operator pencils*) are a family of operators depending on a parameter, that is, operator polynomials of the form

$$\mathcal{P} = \mathcal{P}(\lambda) = A_0 + \lambda A_1 + \lambda^2 A_2 + \dots + \lambda^k A_k,$$

where λ is a spectral parameter and A_0, \ldots, A_k are linear operators acting on a finitedimensional Hilbert space \mathcal{H} . If the operator coefficients are self-adjoint; i.e.

$$A_j = (A_j)^*, \quad j = 0, 1, \dots, k,$$

then such a family is called a self-adjoint operator pencil.

A complex number λ_0 is said to be an eigenvalue of the pencil \mathcal{P} if $\mathcal{P}(\lambda_0)$ is not invertible, or equivalently if

$$\mathcal{P}(\lambda_0)\mathbf{u}=0$$

has a non-zero solution $\mathbf{u} \in \mathcal{H} \setminus \{\mathbf{0}\}$. The set of all eigenvalues of the pencil \mathcal{P} is called the spectrum of \mathcal{P} , which will be denoted by $\operatorname{Spec}(\mathcal{P})$. Moreover, $\operatorname{Spec}(\mathcal{P})$ coincides with the set

$$\operatorname{Spec}(\mathcal{P}) = \{\lambda \in \mathbb{C} : \det(\mathcal{P}) = 0\}.$$

If one considers the simplest case $\mathcal{P}(\lambda) = A - \lambda I$ where *A* is a single operator and *I* is the identity operator, then the pencil problem is equivalent to the spectral problem of a single

operator *A*. The spectrum Spec(A) of a linear operator *A* is the set of $\lambda \in \mathbb{C}$ such that the operator $A - \lambda I$ is not invertible, which is the same as that of the pencil $\mathcal{P}(\lambda) = A - \lambda I$.

For a bounded linear operator T on a Hilbert space $\mathcal H,$ we define the operator norm of T as

$$||T|| = \sup\{|\langle T\mathbf{x}, \mathbf{y}\rangle| : \mathbf{x}, \mathbf{y} \in \mathcal{H}, ||\mathbf{x}|| = ||\mathbf{y}|| = 1\}.$$

The spectral theory of operator pencils has a rich and long history. Pioneering works were undertaken by Tamarkin, Keldysh, Krein, Langer, Shkalikov and others (see [45] for a historical survey). In recent years, it has been a very active research area from both a theoretical and a numerical point view. It has many applications, for instance to control theory [15], vibrating structures [60, 73], mathematical physics and quantum mechanics [6,62] and more recently electron waveguide in graphene [22,28,42]. Several other applications of operator pencils can also be found in [45].

1.2 Indefinite linear pencils

This thesis is devoted to the theory of indefinite linear pencils. We begin by comparing the systems which have a definite or indefinite operator *A*. A self-adjoint operator $A : \mathcal{H} \to \mathcal{H}$ is said to be *positive definite* if and only if all its eigenvalues are positive, or equivalently

$$\langle A\mathbf{x}, \mathbf{x} \rangle > 0$$

for every non-zero vector $\mathbf{x} \in \mathcal{H}$. The *negative definite* operators are defined analogously. A linear operator A is called *indefinite* if there exist $\mathbf{x}, \mathbf{y} \in \mathcal{H}$ for which

$$\langle A\mathbf{x}, \mathbf{x} \rangle > 0, \quad \langle A\mathbf{y}, \mathbf{y} \rangle < 0.$$

One often finds a generalised spectral problem for a pair of self-adjoint operators A, B acting in a Hilbert space \mathcal{H} as: find $\lambda \in \mathbb{C}$ such that

$$A\mathbf{u} = \lambda B\mathbf{u}.\tag{1.1}$$

Suppose that B > 0 (i.e. *B* is positive definite), then there exists $\sqrt{B} > 0$ and, by a change of variables,

$$\mathbf{v} = B^{1/2}\mathbf{u} \quad \Leftrightarrow \quad B^{-1/2}\mathbf{v} = \mathbf{u},$$

so that the problem (1.1) becomes

 $A\mathbf{u} = \lambda B\mathbf{u} \quad \Leftrightarrow \quad AB^{-1/2}\mathbf{v} = \lambda B^{1/2}\mathbf{v} \quad \Leftrightarrow \quad B^{-1/2}AB^{-1/2}\mathbf{v} = \lambda \mathbf{v}.$

Now, $B^{-1/2}AB^{-1/2}$ is a product of three self-adjoint operators and it is self-adjoint since $(CDE)^* = E^*D^*C^*$, therefore all its eigenvalues are real.

If A > 0, then by setting $\mu = 1/\lambda$ in (1.1) yields

$$B\mathbf{u} = \mu A\mathbf{u},$$

and then, applying to the same reasoning for $\sqrt{A} > 0$, the spectral problem can be reduced to one for a self-adjoint operator $A^{-1/2}BA^{-1/2}$.

If B < 0 (similarly A < 0), switching the problem to

$$(-A)\mathbf{u} = \lambda(-B\mathbf{u}),$$

and since -B > 0 (or -A > 0), the problem can be modified in the same way and it follows that the spectrum is real.

If both A, B are sign-definite, then the problem (1.1) is called sign-definite as well. If only one of A, B is sign-definite, then (1.1) is called semi-definite (but still reduces to a self-adjoint spectral problem). In the general case when A and B are *sign-indefinite* (or *indefinite*), the same tricks cannot be applied, and in this case (1.1) is called indefinite.

If B is invertible, then the problem (1.1) can be reduced to

$$B^{-1}A\mathbf{u} = \lambda \mathbf{u}.$$

However, in general, B^{-1} and A do not necessarily commute, that is,

$$(B^{-1}A)^* = AB^{-1} \neq B^{-1}A,$$

thus the operator $B^{-1}A$ may not be self-adjoint. Therefore there may be some complex eigenvalues in the spectrum.

The problem (1.1) is the spectral problem for the linear pencil

$$\mathcal{P}\left(\lambda\right) = A - \lambda B,\tag{1.2}$$

where A, B is a pair of self-adjoint operators acting in a Hilbert space \mathcal{H} and $\lambda \in \mathbb{C}$. A pencil of the form (1.2) with both A and B self-adjoint is called a self-adjoint pencil. We are interested in the case when both A and B are sign-indefinite. We always assume that B is invertible so that the spectrum of the pencil $\mathcal{P}(\lambda)$ coincides with that of the operator $B^{-1}A$.

The difficulty is that standard operator theoretical methods available for the self-adjoint case, for instance the classical variational principles and the spectral theorem, do not readily apply and very little information can be deduced about non-real eigenvalues from the general theory, see, for instance, the next result.

Lemma 1.2.1. Suppose that \mathbf{u} is an eigenvector corresponding to the eigenvalue λ of a self-adjoint pencil $A - \lambda B$. If $\lambda \in \mathbb{C} \setminus \mathbb{R}$, then $\langle B\mathbf{u}, \mathbf{u} \rangle = 0$ and thus $\langle A\mathbf{u}, \mathbf{u} \rangle = 0$.

Proof. Multiplying (1.1) with \mathbf{u} from both sides, we get

$$\langle A\mathbf{u},\mathbf{u}\rangle = \lambda \langle B\mathbf{u},\mathbf{u}\rangle$$

We know that $\langle A\mathbf{u}, \mathbf{u} \rangle$ and $\langle B\mathbf{u}, \mathbf{u} \rangle$ are real and therefore taking the imaginary part of both sides we obtain

$$0 = \langle B\mathbf{u}, \mathbf{u} \rangle \mathrm{Im}(\lambda),$$

which implies that $\lambda \in \mathbb{R}$ or $\langle B\mathbf{u}, \mathbf{u} \rangle = 0$. The result follows since we consider $\lambda \in \mathbb{C} \setminus \mathbb{R}$. \Box

The main objective of this thesis is to study the spectral properties of some signindefinite self-adjoint linear operator pencils. There are not many examples of such pencils in the literature. Similar problems appeared in [19,22,42,46] where it was shown that many sign-indefinite problems which are trivial to state, require highly non-trivial analysis. One cannot obtain good bounds and estimates on eigenvalues by purely functional analytic methods.

1.3 Key results and outline of the thesis

Chapter 2 will provide the necessary background from functional analysis. We begin this chapter with a brief overview of block matrices. The statements given in Section 2.1 are already known, however we will use them in Chapter 4 to obtain information about the eigenvalues of a pencil problem.

In Section 2.2 we present a review of the known Gershgorin-type results. Gershgorin's method [25] is a well-known method for the localisation of the eigenvalues of a complex square matrix. His result is usually referred to as the Gershgorin (circle) Theorem, and states that all the eigenvalues of a matrix with complex entries are contained in a union of disks. The centre of the disks are the diagonal entries of the matrix, and the radii are given by the row sum of the absolute values of the non-diagonal entries. Namely, the result is given as follows.

Theorem 1.3.1 (Gershgorin (circle) Theorem, [25,79]). Let $a_{i,j}$ denote the (i, j)-th element of the matrix A, that is, $A = [a_{i,j}] \in \mathbb{C}^{n \times n}$, $n \ge 2$. Then

$$\operatorname{Spec}(A) \subseteq \mathcal{G}(A) := \bigcup_{i=1}^{n} \mathcal{G}_{i}(A),$$

where

$$\mathcal{G}_i(A) := \{ \lambda \in \mathbb{C} : |\lambda - a_{i,i}| \le r_i(A) \},$$
(1.3)

and $r_i(A)$ denotes the *i*-th deleted absolute row sum of A, *i.e.*

$$r_i(A) = \sum_{j \in J_i} |a_{i,j}|, \qquad J_i = \{1, 2, \dots, n\} \setminus \{i\}.$$
(1.4)

Section 2.2 contains a collection of theorems related to Gershgorin-type results on the localisation of eigenvalues. We shall apply these results to a particular problem and illustrate some of them in Section 4.5.3. We will finish this subsection by giving a new result; a Gershgorin-type localisation for a two-parameter eigenvalue problem (see Corollary 2.2.16).

In Section 2.3, we will briefly present the general theory of orthogonal polynomials. We will then discuss some properties of Chebyshev polynomials of the second kind, which are our main interest in this section. We will study the spectral properties of the matrix whose determinant leads to Chebyshev polynomials of the second kind. Our main focus in the last part of this section will be on ratio asymptotic results given by Simon [66]. We will recall some of his results and apply them to a particular problem in Section 4.5.2.

There are also other approaches to sign-indefinte pencils, for instance, Krein spaces, numerical range (NR) and block numerical range (BNR). We conducted some numerical investigations on a linear pencil problem. However, they did not produce much information, therefore we do not pursue them in this thesis and we omitted the plots and the investigations we carried out. A comprehensive introduction to Krein spaces is provided in [5, 14, 27, 34, 40] and to NR and BNR in [43, 47, 61, 63, 75, 76].

In **Chapter 3**, we focus on a two-parameter eigenvalue problem, and most results of this chapter have been published in [41]. The main motivation of this chapter comes from a particular example of a finite-dimensional, sign-indefinite, self-adjoint linear pencil, studied by Davies and Levitin [19]. We consider the linear pencil problem in a more general setting, and under reasonable restrictions, we state localisation theorems for the eigenvalues.

We start the chapter by surveying the work of Davies and Levitin [19], and then introduce and study a two-parameter eigenvalue problem. Namely, let $m, n \in \mathbb{N}$ and N = m+n, then Davies and Levitin [19] consider the linear operator pencil

$$\mathcal{A}_c = \mathcal{A}_c(\lambda) := H_c^{(N)} - \lambda S_{m,n}, \tag{1.5}$$

where

$$H_{c}^{(N)} = \begin{pmatrix} c & 1 & & & \\ 1 & c & 1 & & \\ & \ddots & \ddots & \ddots & \\ & & \ddots & \ddots & \ddots & \\ & & & 1 & c & 1 \\ & & & & 1 & c \end{pmatrix}, \qquad S_{m,n} = \begin{pmatrix} 1 & & & & & \\ & \ddots & & & & \\ & & & 1 & & \\ & & & -1 & & \\ & & & & \ddots & \\ & & & & & -1 \end{pmatrix},$$
(1.6)

with omitted entries equal to zero, and $c \in \mathbb{R}$ is a parameter. The size of both matrices is $N \times N$, and the diagonal matrix $S_{m,n}$ has m plus ones and n minus ones.

The matrix $S_{m,n}$ is indefinite. If either $c \ge 2$ or $c \le -2$, then the matrix $H_c^{(N)}$ is signdefinite. We will clarify this in Section 3.1, and in fact, we will see that a sharper bound exists, i.e. $|c| \ge 2\cos(\pi/(N+1))$. Therefore, the spectrum $\operatorname{Spec}(\mathcal{A}_c)$ is real when $|c| \ge$ 2. However, interesting things occur in relation to the complex eigenvalues of \mathcal{A}_c when |c| < 2. Davies and Levitin [19] obtained the asymptotics as $N \to \infty$ of the complex eigenvalues of \mathcal{A}_c when c = 0 and asymptotic estimates when $c \neq 0$. We continue looking at this example for a fixed N with m = n. There are various ways to treat this system. Davies and Levitin [19] pursued an asymptotic approach for large size matrices (i.e. $N \to \infty$) and posed some conjectures. We have made some progress towards open questions from [19] and we will be looking at this system from a different perspective. The first thing we noted that this linear pencil problem is related to a two-parameter eigenvalue problem.

Let L, V_1 and V_2 be self-adjoint linear operators in some Hilbert space \mathcal{H} . Then a *two-parameter eigenvalue problem* is defined as

$$\mathbf{M}(\alpha,\beta)\mathbf{x} := (\mathbf{L} - \alpha \mathbf{V_1} - \beta \mathbf{V_2})\mathbf{x} = 0, \tag{1.7}$$

where $\alpha, \beta \in \mathbb{C}$ are spectral parameters. A pair (α, β) is called a *pair-eigenvalue* of $\mathbf{M} = \mathbf{M}(\alpha, \beta)$ if there exists a vector $\mathbf{x} \in \mathcal{H} \setminus \{\mathbf{0}\}$ such that (1.7) holds. Such a vector \mathbf{x} is called an *eigenvector* corresponding to the pair-eigenvalue (α, β) . We denote by $\operatorname{Spec}_p(\mathbf{M})$ the set of all pair-eigenvalues of \mathbf{M} . If both $\alpha, \beta \in \mathbb{R}$, then we will call (α, β) a *real pair-eigenvalue* of (1.7). In this thesis we will focus on the case where $\mathcal{H} = \mathcal{H}_1 \oplus \mathcal{H}_2$ for some Hilbert spaces \mathcal{H}_1 and \mathcal{H}_2 , and where \mathbf{L} , \mathbf{V}_1 and \mathbf{V}_2 take the particular forms

$$\mathbf{x} = egin{pmatrix} \mathbf{u} \ \mathbf{v} \end{pmatrix} \in \mathcal{H}_1 \oplus \mathcal{H}_2,$$

and

$$\mathbf{L} = \begin{pmatrix} A & C \\ C^* & D \end{pmatrix}, \qquad \mathbf{V}_1 = \begin{pmatrix} I & 0 \\ 0 & 0 \end{pmatrix}, \qquad \mathbf{V}_2 = \begin{pmatrix} 0 & 0 \\ 0 & I \end{pmatrix}.$$
(1.8)

In here, *A* and *D* are self-adjoint operators in the Hilbert spaces \mathcal{H}_1 , \mathcal{H}_2 , respectively, and *C* is a linear operator from \mathcal{H}_2 to \mathcal{H}_1 , then one can re-write the two-parameter eigenvalue problem (1.7) as

$$\mathbf{M}(\alpha,\beta)\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix} = \begin{pmatrix}A-\alpha & C\\C^* & D-\beta\end{pmatrix}\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix} = \mathbf{0},$$
(1.9)

where $A - \alpha$ and $D - \beta$ are shorthand for $A - \alpha I$ and $D - \beta I$, respectively.

Remark 1.3.2. There may be other points $(\alpha, \beta) \in \mathbb{C}^2$ where M does not have a bounded inverse in general. However, we study the eigenvalues of M and Spec_p stands for point spectrum.

Now, let us investigate the connection between a two-parameter eigenvalue problem (1.9) and a linear pencil problem (1.2). Consider the self-adjoint, linear matrix pencil

$$(\mathbf{L} + cI - \lambda \mathbf{S})\mathbf{x} = \mathbf{0},\tag{1.10}$$

where L is given by (1.8) and

$$\mathbf{S} = \begin{pmatrix} I & \\ & -I \end{pmatrix}.$$

Further λ is a spectral parameter and $c \in \mathbb{R}$ is fixed. Then (1.10) can be re-written as a generalised eigenvalue problem

$$\begin{pmatrix} A+c & C\\ C^* & D+c \end{pmatrix} \begin{pmatrix} \mathbf{u}\\ \mathbf{v} \end{pmatrix} = \lambda \begin{pmatrix} I\\ & -I \end{pmatrix} \begin{pmatrix} \mathbf{u}\\ \mathbf{v} \end{pmatrix},$$
(1.11)

which is equivalent to

$$\begin{pmatrix} A+c-\lambda & C\\ C^* & D+c+\lambda \end{pmatrix} \begin{pmatrix} \mathbf{u}\\ \mathbf{v} \end{pmatrix} = \mathbf{0}.$$
 (1.12)

In addition, the problem (1.11) is also equivalent to the non-self-adjoint problem

$$\begin{pmatrix} A+c & C\\ -C^* & -D-c \end{pmatrix} \begin{pmatrix} \mathbf{u}\\ \mathbf{v} \end{pmatrix} = \lambda \begin{pmatrix} \mathbf{u}\\ \mathbf{v} \end{pmatrix}.$$
 (1.13)

Then it can be seen from (1.12) that any linear pencil problem of the form (1.10) (or a non-self-adjoint problem of the form (1.13)) relates to the two-parameter eigenvalue of the form (1.9) by setting

$$\alpha = \lambda - c; \quad \beta = -\lambda - c. \tag{1.14}$$

Indeed, it can be said that Davies and Levitin [19] considered the two-parameter eigenvalue problem (1.9) in the case when $S = S_{m,n}$, $A = H_0^{(n)}$, $D = H_0^{(m)}$, $(C)_{n,1} = 1$ and all

other entries of *C* are zeros. In Section 3.2, the problem (1.9) is attacked in a more general setting; we consider the case when A, D are self-adjoint operators and *C* is rank one, and we obtain localisation results on the real pair-eigenvalues of the two-parameter eigenvalue problem. The main results (Theorem 3.3.6 and its special case Theorem 3.3.4) will be given in Section 3.3. For simplicity, we omit the statement of our main results here and we give an illustrative example instead. For instance, let

$$A = D = \begin{pmatrix} 1 & 0 \\ 0 & 2 \end{pmatrix}, \quad C = \begin{pmatrix} 1 & 1 \\ 1 & 1 \end{pmatrix},$$

then the pair-eigenvalues of M lie in a union of rectangular regions in the (α, β) -plane. In Figure 1.1, we illustrate the region (light-gray shaded area) which contains the real spectrum of M (blue lines). Boundaries of the region and some special cases will be explained later in this chapter.



Figure 1.1: In the (α, β) -plane, blue lines are the real pair-eigenvalues of M.

The rest of this chapter is devoted to the study of a two-parameter eigenvalue problem. Throughout this chapter, we deal with mainly the real pair-eigenvalues of a twoparameter eigenvalue problem. We focus on the non-real pair-eigenvalues of the problem only in Section 3.7. First, we start Section 3.2 by describing the general concept of a twoparameter eigenvalue problem. Then in Section 3.3 after imposing some restrictions and notation, we state the main results of this chapter concerning the real pair-eigenvalues of M, namely Theorem 3.3.6 and its special case Theorem 3.3.4. In Section 3.4 we briefly discuss some auxiliary results for a special class of problems and subsequently we give the proofs of the main results in Section 3.5. Afterwards, we give several examples to illustrate the special cases in Section 3.6.

We conclude the chapter with a discussion of the non-real pair-eigenvalues of a twoparameter eigenvalue problem. In general, there are infinitely many $\beta \in \mathbb{C}$ for every $\alpha \in \mathbb{C} \setminus \text{Spec}(A)$. Thus we shall limit our attention to those that satisfy

$$\operatorname{Im}(\alpha + \beta) = 0.$$

Therefore, we set

$$\operatorname{Spec}(\mathbf{M}) := \operatorname{Spec}_{p}(\mathbf{M}) \cap \{(\alpha, \beta) \in \mathbb{C}^{2} : \operatorname{Im} \alpha + \operatorname{Im} \beta = 0\}.$$

First, we locate the collision points of the real eigenvalues (or double eigenvalues) in terms of the first derivative of $\beta(\alpha)$ with respect to α and we state the result in Lemma 3.7.3. Another major result of our investigation concerning the non-real pair-eigenvalues of M is given in Theorem 3.7.4. Using asymptotic analysis, we prove that as ||C|| diverges to infinity there exists a family of non-real pair eigenvalues which approximately lies on a circle in the complex plane. The centre and the radius of this circle will be explained in Section 3.7.1, including illustrative examples.

We estimate the real eigenvalues of the pencil A_c by looking at a corresponding twoparameter eigenvalue problem. We also obtain partial information about the non-real paireigenvalues of the problem which we will study in detail in Section 3.7. In addition, we obtain some minimal information about the collisions of the real eigenvalues (i.e. double eigenvalues). Therefore, we need to look at the concrete example again so as to gain more insight about the open problems from [19].

We note that in our publication [41] we focus on the two-parameter eigenvalue problem (1.9) in the case when $\dim \mathcal{H}_1 = \dim \mathcal{H}_2$, whereas in this chapter, we extend our results to a more general setting. In addition, Lemma 3.5.5, Lemma 3.7.1 and Theorem 3.7.4 are new, and do not appear in the paper [41].

In **Chapter 4** we turn our attention to the linear pencil problem (1.5). We shall only consider the case when m = n in this chapter. By rephrasing the work of Davies and Levitin [19], we are able to gain a new perspective. First, we look at the characteristic equation of the pencil A_c in Section 4.1 which can be written in several ways, one of which is using iterative functions that turn out to be related to ratios of Chebyshev polynomials of the second kind U_n . In Section 4.1.1, we shall deduce the characteristic equation of the pencil A_c explicitly and give explicit expressions for the eigenfunctions of the pencil A_c in terms of the Chebyshev polynomials of the second kind.

We will frequently use the following substitution

$$\lambda - c := z + \frac{1}{z}, \quad \lambda + c := w + \frac{1}{w},$$
 (1.15)

where λ is an eigenvalue of the pencil \mathcal{A}_c and $z, w \in \mathbb{C}$. Then each λ corresponds to two values of z and two values of w, which are the solutions of the quadratic equations

$$z^{2} - (\lambda - c)z + 1 = 0,$$

 $w^{2} - (\lambda + c)w + 1 = 0,$

respectively. Using (1.15), Davies and Levitin [19] gave another form of the characteristic equation, and we quote some of their results in Section 4.1.2. In Section 4.1.3, the pencil problem will be re-written in the basis of the eigenfunctions $\{\psi_j\}$ of $H_0^{(n)}$.

Numerical experiments suggest that the eigenvalues of a sign-indefinite self-adjoint pencil often lie on or under a set of curves. Davies and Levitin [19] proved that this is true in some cases, and determined the curve asymptotically which bounds the spectrum of A_c when $c \neq 0$. Nevertheless, the crucial step in the proof was the following conjecture.

Conjecture 1.3.3 ([19, Conjecture 5.3]). Let c > 0 and m = n. If λ is a non-real eigenvalue of A_c , then

$$|\lambda \pm c| < 2, \tag{1.16}$$

and therefore

$$|\operatorname{Re}(\lambda)| < 2 - c. \tag{1.17}$$

Conjecture 1.3.3 is still open. We have verified this conjecture numerically for a wide range of *n* and *c*, and we will give numerical experiments to support this conjecture throughout this thesis. One of these experiments is illustrated in Figure 1.2, showing the complicated interplay between $\max_{\lambda \in \text{Spec}(\mathcal{A}_c) \setminus \mathbb{R}} \{|\lambda + c|, |\lambda - c|\}$ and *c*. We see in the figure that the maximum value of $|\lambda \pm c|$ never exceeds two.

We will look at the pencil problem from different angles and we will see that there are several ways to reformulate Conjecture 1.3.3. To make these reformulations more accessible as a whole, we summarize them as follows.

Theorem 1.3.4. Conjecture 1.3.3 is satisfied if and only if any of the following statements hold:



Figure 1.2: For n = 10, $\max\{|\lambda + c|, |\lambda - c|\}$ (red lines) among all non-real eigenvalues of A_c is drawn as c ranges from 0 to 2 with the step-size 0.001 and the constant line at 2 (blue dotted lines).

(i) Consider a two-parameter eigenvalue problem $\mathbf{M}(\alpha, \beta)$ defined by (1.9) with $A = D = H_0^{(n)}$ and $C = C^*$ with $(C)_{nn} = 1$ and all other entries of C are zeros. Then

Spec(**M**)
$$\setminus \mathbb{R}^2 \subset \{(\alpha, \beta) : |\alpha|, |\beta| < 2\}.$$

(ii) Let $\sigma, \tau \in \mathbb{C}$, $\operatorname{Im}(\sigma) = \operatorname{Im}(\tau) > 0$. If, for some $n \in \mathbb{N}$,

$$U_{n}(\sigma) U_{n}(\tau) + U_{n-1}(\sigma) U_{n-1}(\tau) = 0, \qquad (1.18)$$

then $|\sigma| < 1$ *and* $|\tau| < 1$ *.*

(iii) Let $\widetilde{F}_n : \mathbb{C} \to \mathbb{C}$, $n \ge 1$ denote the sequence of iteratively defined rational functions

$$\widetilde{F}_{n+1}(\zeta) = \zeta - \frac{1}{\widetilde{F}_n(\zeta)}, \qquad \widetilde{F}_1(\zeta) = \zeta, \quad \zeta \in \mathbb{C}.$$

Then any solution $(\sigma, \tau) \in \mathbb{C}^2$ of

$$\widetilde{F}_n(\sigma)\widetilde{F}_n(\tau) = -1, \qquad \operatorname{Im}(\sigma) = \operatorname{Im}(\tau) > 0,$$

has both $|\sigma| < 2$ and $|\tau| < 2$.

(iv) Let $G_n : \mathbb{C} \to \mathbb{C}$, $n \ge 1$ denote the family of meromorphic functions

$$G_n(\xi) := \frac{\xi^{n+1} - \xi^{-n-1}}{\xi^n - \xi^{-n}}.$$

Let $z, w \in \mathbb{C} \setminus \mathbb{R}$ such that |z|, |w| > 1. Then any solutions z, w of

$$G_n(z)G_n(w) = -1,$$

satisfy

$$z, w \in \mathcal{Z}_1 \cup \mathcal{Z}_2 \tag{1.19}$$

where

$$\mathcal{Z}_1 := \left\{ \zeta \in \mathbb{C} : |\zeta| > 1 \text{ and } |\zeta - i| < \sqrt{2} \right\}, \qquad \mathcal{Z}_2 := \left\{ \zeta \in \mathbb{C} : |\zeta| > 1 \text{ and } |\zeta + i| < \sqrt{2} \right\}.$$

(v) Let $A_c = S_{n,n}^{-1} H_c^{(N)}$. Then

$$\operatorname{Spec}(\mathbf{A}_c) \setminus \mathbb{R} \subset \left(\bigcup_{i=1}^n \mathcal{G}_i(\mathbf{A}_c) \right) \bigcap \left(\bigcup_{j=n+1}^{2n} \mathcal{G}_j(\mathbf{A}_c) \right)$$

We prove most of these reformulations in Section 4.2. Only two of these equivalences were deduced by Davies Levitin [19], namely Theorem 1.3.4(ii) and (iii). We prove these reformulations in Section 4.2.1. The rest of the reformulations are new. We give a proof of Theorem 1.3.4(i) in Section 3.7. We will derive Theorem 1.3.4(iv) in Section 4.2.2 and verify Theorem 1.3.4(v) in Section 4.2.3.

In Section 4.3 we give a proof of Conjecture 1.3.3 in the case n = 2 and n = 3. Although we can prove the conjecture explicitly, the proof involves very long computations for even small matrices. In Section 4.4, we describe the asymptotic behaviour of the non-real eigenvalues of the pencil A_c as the parameter c tends to zero.

Our purpose in Section 4.5 is to apply useful methods mentioned in Chapter 2 to our problem. We begin this section with a brief discussion of the pencil problem in a block operator setting. The main objective is to show the difficulty of the problem.

As mentioned in Theorem 1.3.4(ii), Conjecture 1.3.3 can be reformulated as a result involving Chebyshev polynomials of the second kind U_n . This statement is apparently a new statement for Chebyshev polynomials. One approach is to treat the system as the ratios of Chebyshev polynomials of the second kind. It is still an open question to estimate the ratio for orthogonal polynomials of finite index. In Section 4.5.2, first we apply the ratio asymptotic results given in Section 2.3.4 to the ratio $U_n(\zeta)/U_{n-1}(\zeta)$. Afterwards, we investigate various properties of this finite ratio and deduce some bounds on this ratio. Although these bounds are not sufficient to prove the conjecture, we believe that they are still useful. Most of these results in this subsection are joint with Sabine Bögli (unpublished). Although the reformulations given in Theorem 1.3.4(ii) and (iii) were already deduced by Davies and Levitin [19], they haven not studied the ratio of orthogonal polynomials, and we remark that our investigation is a new approach.

Another form of Conjecture 1.3.3 is given in terms of Gershgorin circles as in Theorem 1.3.4(v). When we apply Gershgorin's Theorem to our problem, we will see in Section 4.2.3 that all the eigenvalues of A_c lie in the union of all disks, whereas we claim that the non-real eigenvalues of A_c lie in the intersection of two regions. In Section 4.5.3, we will apply various Gershgorin-type results mentioned in Section 2.2 to our pencil problem. We will see that there is no such Gershgorin-type region which contains only the non-real eigenvalues of a matrix. We tried but failed to obtain such bounds, and therefore we omitted our calculations. Instead, we will investigate and illustrate some of the known results, and we will see that a very few of them give a better bound than Gershgorin-type localisation for a two-parameter eigenvalue problem to our problem, and we look at the Gershgorin's result under the change of variables (1.15).

Chapter 5 focuses on localisation of the non-real eigenvalues from different perspectives. We mention some heuristics based on numerical experiments. In Section 5.1, we explain the dynamics of the eigenvalues of the pencil A_c as the parameter c changes. There is a fascinating dynamics of real eigenvalues colliding to form a pair of complexconjugate eigenvalues emerging into the complex plane, and the reverse collisions of complex eigenvalues. We illustrate these collisions (or jumps) which occur in the first quadrant of the complex plane in Table 5.1. In Section 5.2, we illustrate some interesting phenomena for the non-real pair-eigenvalues of a two parametric eigenvalue problem. In Section 5.3, we investigate double eigenvalues of the pencil A_c . One would see an interesting behaviour if one superimposes the double eigenvalues of A_c for the values of n, see Figure 5.6. In addition, we count the number of double eigenvalues which lie in a certain location. Numerical evidence suggests that for c between 0 and 2, there are altogether $2n^2$ double λ -roots (conjectured), of which 2n are at the origin and the rest are non-zero.

Chapter 2

Background

In this chapter, we discuss some necessary background material from functional analysis. In Section 2.1, we review some properties of block matrices. Most of the results contained in this section are probably well-known, however these ideas will be used in Chapter 4. Section 2.2 contains a review of the well-known Gershgorin-type results, which we will apply in Section 4.5.3 to our problem and give a detailed discussion of their applications. In Section 2.3, we provide a brief overview of orthogonal polynomials before turning to a discussion of Chebyshev polynomials of the second kind. We will introduce some notation and basic terminology, and discuss some of the spectral properties of a matrix. We conclude this section with a brief survey of ratio asymptotics of orthogonal polynomials.

2.1 Block matrices

In this section, we shall discuss some properties of block (operator) matrices. Let

$$\mathcal{H} = \mathcal{H}_1 \oplus \mathcal{H}_2 \tag{2.1}$$

be a direct sum of complex Hilbert spaces $\mathcal{H}_1, \mathcal{H}_2$. A linear operator $\mathbf{X} \in B(\mathcal{H})$ always admits a block (operator) matrix representation

$$\mathbf{X} = \begin{pmatrix} A_{11} & A_{12} \\ A_{21} & A_{22} \end{pmatrix}, \tag{2.2}$$

with linear operators $A_{ij} \in \mathcal{B}(\mathcal{H}_j, \mathcal{H}_i)$, i, j = 1, 2. Most of the time, we will be interested in the special case when $\dim \mathcal{H}_1 = \dim \mathcal{H}_2$, $A_{jj} = A_{jj}^*$, j = 1, 2, and either $A_{21} = -A_{12}$ or $A_{21} = -A_{12}^*$. In the rest of this section, we set **X** as a 2×2 block matrix with respect to the decomposition (2.1) and review some of its spectral properties.

We start by considering a 2×2 matrix, and we will then generalise the result for a 2×2 block matrix. The following lemma establishes certain bounds on the eigenvalues of 2×2 matrices.

Lemma 2.1.1. Let $Y = \begin{pmatrix} a & b \\ -\overline{b} & d \end{pmatrix}$, where $a, d \in \mathbb{R}$, and $b \in \mathbb{C}$. If $\lambda \in \mathbb{C} \setminus \mathbb{R}$ is an eigenvalue of Y, then

$$|a - d| < 2|b|,$$

$$\operatorname{Re}(\lambda) - a| \le |b|,$$

$$\operatorname{Re}(\lambda) - d| \le |b|,$$

$$|\operatorname{Im}(\lambda)| \le |b|.$$

Proof. Set $\beta = a + d$ and $\gamma = ad + |b|^2$, then the elementary formula $\lambda^2 - \beta\lambda + \gamma = 0$ gives

$$\lambda = \frac{a+d}{2} \pm \sqrt{\frac{(a+d)^2}{4} - ad - |b|^2}$$
$$= \frac{a+d}{2} \pm \sqrt{\frac{(a-d)^2}{4} - |b|^2}.$$

The assumption $\lambda \notin \mathbb{R}$ implies that $(a - d)^2/4 - |b|^2 < 0$ and hence |a - d|/2 < |b|. It also implies that $\operatorname{Re}(\lambda) = (a + d)/2$. Therefore

$$|\operatorname{Re}(\lambda) - a| = \left|\frac{a+d}{2} - a\right| = \frac{|a-d|}{2} < |b|,$$

and similarly

$$|\operatorname{Re}(\lambda) - d| = \frac{|a - d|}{2} < |b|,$$

$$|\operatorname{Im}(\lambda)| = \sqrt{|b|^2 - \frac{(a - d)^2}{4}} \implies |\operatorname{Im}(\lambda)| \le |b|.$$

Remark 2.1.2. Lemma 2.1.1 is analogous to Lemma 1.2.7 from [75].

The next result gives some information for the eigenvectors of a certain type of a block matrix.

Lemma 2.1.3. Let $\gamma \in \mathbb{R}$. If $\lambda \in \mathbb{C} \setminus \mathbb{R}$ is an eigenvalue of

$$\mathbf{X}_{\gamma} = \begin{pmatrix} A_{11} - \gamma & A_{12} \\ -A_{12}^* & A_{22} + \gamma \end{pmatrix},$$

where $A_{jj} = A_{jj}^*$, j = 1, 2, then the corresponding eigenvectors $(\mathbf{u}, \mathbf{v})^T$ satisfy $||\mathbf{u}|| = ||\mathbf{v}||$. In addition,

$$|\mathrm{Im}(\lambda)| \le ||A_{12}||.$$

Proof. Consider the spectral problem

$$\begin{pmatrix} A_{11} - \gamma & A_{12} \\ -A_{12}^* & A_{22} + \gamma \end{pmatrix} \begin{pmatrix} \mathbf{u} \\ \mathbf{v} \end{pmatrix} = \lambda \begin{pmatrix} \mathbf{u} \\ \mathbf{v} \end{pmatrix}.$$

Let $\lambda = x + iy$ where $x \in \mathbb{R}$ and $y \in \mathbb{R} \setminus \{0\}$. Then re-writing the problem as

$$\begin{cases} (A_{11} - \gamma)\mathbf{u} + A_{12}\mathbf{v} &= (x + iy)\mathbf{u}, \\ -A_{12}^*\mathbf{u} + (A_{22} + \gamma)\mathbf{v} &= (x + iy)\mathbf{v}, \end{cases}$$

and multiplying the first equation by ${\bf u}$ and the second by ${\bf v}$ gives

$$\begin{cases} \langle A_{11}\mathbf{u},\mathbf{u}\rangle - \gamma \|\mathbf{u}\|^2 + \langle A_{12}\mathbf{v},\mathbf{u}\rangle = (x+iy)\|\mathbf{u}\|^2, \\ -\langle A_{12}^*\mathbf{u},\mathbf{v}\rangle + \langle A_{22}\mathbf{v},\mathbf{v}\rangle + \gamma \|\mathbf{v}\|^2 = (x+iy)\|\mathbf{v}\|^2. \end{cases}$$
(2.3)

Considering the imaginary parts of (2.3), we get

$$\begin{cases} -y \|\mathbf{u}\|^2 + \operatorname{Im}\langle A_{12}\mathbf{v}, \mathbf{u} \rangle &= 0\\ \operatorname{Im}\langle A_{12}^*\mathbf{u}, \mathbf{v} \rangle + y \|\mathbf{v}\|^2 &= 0, \end{cases}$$
(2.4)

and since $\langle A_{12}\mathbf{v}, \mathbf{u} \rangle = \overline{\langle A_{12}^*\mathbf{u}, \mathbf{v} \rangle}$, summing the two equations in (2.4) gives the first result. In addition, we have from (2.4) that

$$|y| = \frac{|\text{Im}\langle A_{12}\mathbf{v}, \mathbf{u}\rangle|}{\|\mathbf{u}\|^2} \le \frac{\|A_{12}\mathbf{v}\|\|\mathbf{u}\|}{\|\mathbf{u}\|^2},$$
(2.5)

and since ||u|| = ||v||, we obtain $|y| \le ||A_{12}||$.

Remark 2.1.4. Note that we have the relation

$$\lambda \in \operatorname{Spec} \begin{pmatrix} A_{11} - \gamma & A_{12} \\ -A_{12}^* & A_{22} + \gamma \end{pmatrix} \quad \Leftrightarrow \quad \gamma \in \operatorname{Spec} \begin{pmatrix} A_{11} - \lambda & A_{12} \\ A_{12}^* & -A_{22} + \lambda \end{pmatrix}.$$

Therefore, it follows that for $\lambda \in \mathbb{C} \setminus \mathbb{R}$, eigenvectors $(\mathbf{u}, \mathbf{v})^T$ corresponding to the real eigenvalues of

$$\begin{pmatrix} A_{11} - \lambda & A_{12} \\ A_{12}^* & -A_{22} + \lambda \end{pmatrix}$$

satisfy $\|\mathbf{u}\| = \|\mathbf{v}\|$.

Remark 2.1.5. Consider Lemma 2.1.3. If, in addition, $\|\mathbf{u}\| = \|\mathbf{v}\| = 1$, then we deduce from (2.5) that

$$|\mathrm{Im}(\lambda)| = |\mathrm{Im}\langle A_{12}\mathbf{v}, \mathbf{u}\rangle|$$

Theorem 2.1.6. Consider

$$\mathbf{X}_0 = \begin{pmatrix} A_{11} & A_{12} \\ -A_{12}^* & A_{22} \end{pmatrix},$$

where $A_{11} = A_{11}^*$ satisfies $\operatorname{Spec}(A_{11}) \subseteq [a_1, a_2]$, and $A_{22} = A_{22}^*$ satisfies $\operatorname{Spec}(A_{22}) \subseteq [d_1, d_2]$. Let $\mathbf{X}_0 \mathbf{f} = \lambda \mathbf{f}$ where $\lambda \in \mathbb{C} \setminus \mathbb{R}$ and $\mathbf{f} = \begin{pmatrix} \mathbf{u} \\ \mathbf{v} \end{pmatrix}$ such that $\|\mathbf{u}\|^2 + \|\mathbf{v}\|^2 = 2$. Then

$$|\mathbf{u}|| = ||\mathbf{v}|| = 1,$$
 (2.6)

and

$$\max(a_1, d_1) - \delta \le \operatorname{Re}(\lambda) \le \min(a_2, d_2) + \delta,$$

where $\delta = |\langle A_{12}\mathbf{v}, \mathbf{u} \rangle|$.

Proof. Equation (2.6) immediately follows from Lemma 2.1.3. Now, re-writing $X_0 f = \lambda f$ in the form

$$A_{11}\mathbf{u} + A_{12}\mathbf{v} = \lambda \mathbf{u},$$
$$-A_{12}^*\mathbf{u} + A_{22}\mathbf{v} = \lambda \mathbf{v},$$

leads to

$$\langle A_{11}\mathbf{u},\mathbf{u}\rangle + \langle A_{12}\mathbf{v},\mathbf{u}\rangle = \lambda,$$
 (2.7)

$$-\langle A_{12}^*\mathbf{u},\mathbf{v}\rangle + \langle A_{22}\mathbf{v},\mathbf{v}\rangle = \lambda.$$
(2.8)

This may be re-written in the form

$$\begin{pmatrix} a & b \\ -\overline{b} & d \end{pmatrix} \begin{pmatrix} 1 \\ 1 \end{pmatrix} = \lambda \begin{pmatrix} 1 \\ 1 \end{pmatrix},$$

where $a = \langle A_{11}\mathbf{u}, \mathbf{u} \rangle$, $b = \langle A_{12}\mathbf{v}, \mathbf{u} \rangle$, $\delta = |b|$, and $d = \langle A_{22}\mathbf{v}, \mathbf{v} \rangle$. The theorem now follows directly from Lemma 2.1.1 and the inclusion $a \in [a_1, a_2]$ and $d \in [d_1, d_2]$.

Remark 2.1.7. Theorem 2.1.6 is similar to Proposition 1.3.9 from [75] where she estimates

Spec
$$(\mathbf{X}_0) \cap \mathbb{R} \subset [\min(a_1, d_1), \max(a_2, d_2)].$$

Our result is somewhat stronger if $\delta = |\langle A_{12}\mathbf{v}, \mathbf{u} \rangle|$ is small.

2.2 Gershgorin circles

One of the most famous methods which can be used for eigenvalue localisation regions was introduced by the mathematician Semyon Aranovich Gershgorin in 1931 [25]. He observed that the spectrum of a complex matrix is contained in a union of disks in the complex plane, which are nowadays called Gershgorin disks. These disks are centred at the diagonal entries of the matrix and their radii are the row (or column) sum of the moduli of non-diagonal entries; see Theorem 1.3.1 for the formal statement of Gershgorin's result. It is an easy tool to obtain bounds on the eigenvalues without assuming any condition on a matrix. It has many applications, for example, in stability theory (see [68]), in locating the roots of polynomials (see [1,50,54,55]) and controller design (see [32,64]).

In general, eigenvalue inclusion regions are useful in eigenvalue perturbation theory (see [71, 81]) and in the computation of pseudospectra (see [72]). A good survey of eigenvalue bounds including their rich and interesting history can be found in [31, Chapter 6], whereas [79] provides more advanced and comprehensive results. In this section, the well-known results about Gershgorin-type eigenvalue localizations will be reviewed.

2.2.1 Inclusion regions for the eigenvalues of matrices

We begin by recalling some notation. The (i, j)-th element of the matrix A is denoted by $a_{i,j}$, i.e. $A = [a_{i,j}] \in \mathbb{C}^{n \times n}$. $r_i(A)$ denotes the *i*-th deleted absolute row sum of A, i.e.

$$r_i(A) := \sum_{j \in J_i} |a_{i,j}|,$$

where $J_i = \{1, 2, ..., n\} \setminus \{i\}$. We now give a proof of Gershgorin Theorem (Theorem 1.3.1).

Proof of Theorem 1.3.1. Let λ be an eigenvalue of A and \mathbf{x} be a corresponding eigenvector. Suppose the *i*-th component of \mathbf{x} has the largest modulus; i.e. $|x_i| = \max_j |x_j|$. Looking at the *i*-th coordinate of the equation $A\mathbf{x} = \lambda \mathbf{x}$, it can be seen that

$$\sum_{j=1}^{n} a_{i,j} x_j = \lambda x_i \quad \Leftrightarrow \quad a_{i,i} x_i + \sum_{j \in J_i} a_{i,j} x_j = \lambda x_i \quad \Leftrightarrow \quad \sum_{j \in J_i} a_{i,j} x_j = (\lambda - a_{i,i}) x_i.$$

Taking the modulus on the both sides of the last equation and applying the triangle inequality and then the maximal property of $|x_i|$ yields

$$|\lambda - a_{i,i}||x_i| = \left|\sum_{j \in J_i} a_{i,j} x_j\right| \le \sum_{j \in J_i} |a_{i,j}||x_j| \le |x_i| \sum_{j \in J_i} |a_{i,j}| = |x_i| r_i(A).$$

Since $|x_i| > 0$, dividing both sides of the inequality by $|x_i|$ gives $|\lambda - a_{i,i}| \le r_i(A)$, i.e. λ lies in a disk with centre $a_{i,i}$ and radius $r_i(A)$. Since it is not clear which *i* each eigenvalue corresponds to, one needs to take the union of all such disks in order to obtain a guaranteed region.

The set $\mathcal{G}_i(A)$ is called the *i*-th Gershgorin disk of *A*, which is a closed disk in the complex plane \mathbb{C} , centred at $a_{i,i}$ with a radius of $r_i(A)$. Then $\mathcal{G}(A)$ is the union of the *n* Gershgorin disk, and it is called the Gershgorin set which is closed and bounded in \mathbb{C} .

Remark 2.2.1. Since the spectra of A and its transpose A^T are identical, an analogous statement is obtained by considering the matrix A^T , with deleted column sums replacing deleted row sums. This will also be true for all the other results in this section unless stated otherwise.

The next well-known result is given by Brauer (1947).

Theorem 2.2.2 ([79, Theorem 2.2]). Let $A = [a_{i,j}] \in \mathbb{C}^{n \times n}$, $n \ge 2$. Then

$$\operatorname{Spec}(A) \subseteq \mathcal{K}(A) := \bigcup_{\substack{i,j=1\\i\neq j}}^{n} \mathcal{K}_{i,j}(A),$$

where

$$\mathcal{K}_{i,j}(A) := \{ \zeta \in \mathbb{C} : |\zeta - a_{i,i}| | \zeta - a_{j,j} | \le r_i(A) r_j(A) \}.$$
(2.9)

Cassini ovals are a family of quartic curves defined as the set of points such that the product of the distances to two fixed points is constant, and the fixed points are called the foci of the Cassini oval. $\mathcal{K}_{i,j}(A)$ is called the (i, j)-th Brauer Cassini oval for the matrix A, whereas $\mathcal{K}(A)$ is called the Brauer set.

Remark 2.2.3. Note that $\mathcal{G}_i(A)$ is sometimes also called the *i*-th Gershgorin circle of *A*. Both (i, j)-th Brauer Cassini oval $\mathcal{K}_{i,j}(A)$ and *i*-th Gershgorin circle $\mathcal{G}_i(A)$ were named with the shape of their boundaries, however both $\mathcal{K}_{i,j}(A)$ and $\mathcal{G}_i(A)$ are regions bounded by a Cassini oval and a circle respectively.

Remark 2.2.4. For an $n \times n$ matrix, there are n(n-1)/2 ovals of Brauer Cassini, while there are *n* Gershgorin disks.

The next result is due to Fan and Hoffman (1954).

Theorem 2.2.5 ([79, Theorem 1.20]). For any matrix $A = [a_{i,j}] \in \mathbb{C}^{n \times n}$, if α is any positive number satisfying ¹

$$\sum_{i=1}^{n} \frac{r_i(A)}{\max_{j \in J_i} |a_{i,j}|} \le \alpha (1+\alpha),$$

then

$$\operatorname{Spec}(A) \subseteq \bigcup_{i=1}^{n} \{ \zeta \in \mathbb{C} : |\zeta - a_{i,i}| \le \rho_i \},\$$

where

$$\rho_i = \alpha \max_{j \in J_i} |a_{i,j}|. \tag{2.10}$$

Melman found new eigenvalue inclusion sets for matrices without assuming any structure in [50, 52, 55], for structured matrices in [49], and for Toeplitz type matrices in [53], however we shall mention only two of them. Before stating the results, we introduce additional notation. Let $J_{i,j} = \{1, ..., n\} \setminus \{i, j\}$, then define

$$\widetilde{r}_{i,j} := \sum_{k \in J_{i,j}}^{n} |a_{i,k}| = r_i - |a_{i,j}| = \sum_{k=1}^{n} |a_{i,k}| - |a_{i,i}| - |a_{i,j}| \qquad (\text{for } i \neq j).$$
(2.11)

Theorem 2.2.6 ([50, Theorem 2.1]). Let $A = [a_{i,j}] \in \mathbb{C}^{n \times n}$. Then

$$\operatorname{Spec}(A) \subseteq \Omega(A) := \bigcup_{i=1}^{n} \left\{ \bigcap_{j \in \widetilde{J}_{i}} \Omega_{i,j}(A) \right\},$$

where \widetilde{J}_i is any non-empty subset of J_i and

 $\Omega_{i,j}(A) := \{ \zeta \in \mathbb{C} : |(\zeta - a_{i,i})(\zeta - a_{j,j}) - a_{i,j}a_{j,i}| \le |\zeta - a_{j,j}|\widetilde{r}_{i,j}(A) + |a_{i,j}|\widetilde{r}_{j,i}(A) \}.$ (2.12)

In addition,

$$\Omega_{i,j}(A) \subseteq \mathcal{G}_i(A) \cup \mathcal{G}_j(A)$$
 for all $i \neq j$.

Recall that the matrix is centrohermitian if it is symmetric with respect to its centre, i.e. its elements satisfy $a_{i,j} = \overline{a}_{n-i+1,n-j+1}$, and it is skew-centrohermitian if $a_{i,j} = -\overline{a}_{n-i+1,n-j+1}$, where $\overline{a}_{i,j}$ is the complex conjugate of $a_{i,j}$.

Theorem 2.2.7 ([49, Theorem 2.6]). Let $A = [a_{i,j}] \in \mathbb{C}^{n \times n}$ such that *n* is even and *A* is skew-centrohermitian. Then

$$\operatorname{Spec}(A) \subseteq \bigcup_{i=1}^{n} \mathcal{V}_{i}(A),$$

where $\mathcal{V}_i(A)$ is defined to be the set

$$\{\zeta \in \mathbb{C} : \left| (\zeta - a_{i,i})(\zeta - \overline{a}_{i,i}) - |a_{i,n-i+1}|^2 \right| \le (|\zeta + \overline{a}_{i,i}| + |a_{i,n-i+1}|) \widetilde{r}_{i,n-i+1}(A) \}.$$
(2.13)
¹If $\max_{j \in J_i} |a_{i,j}| = 0 = r_i(A)$ for some *i*, then $r_i(A) / \max_{j \in J_i} |a_{i,j}|$ is defined to be zero.

2.2.2 Exclusion regions for the eigenvalues of matrices

There are some proper subsets of the Gershgorin set which do not contain any eigenvalue of the matrix. These sets give substantial improvements to determine the location of the eigenvalues of a matrix (cf. [44,51]). Melman [51] gave the following Gersghorin-type set.

Theorem 2.2.8. Let $A = [a_{i,j}] \in \mathbb{C}^{n \times n}$. Then

$$\operatorname{Spec}(A) \subseteq \widetilde{\Omega}(A) = \bigcup_{i=1}^{n} \widetilde{\Omega}_{i}(A),$$

where

$$\widetilde{\Omega}_i(A) := \mathcal{G}_i(A) \setminus \mathcal{L}_i(A), \qquad \mathcal{L}_i(A) := \bigcup_{j \in J_i} \mathcal{L}_{i,j}(A),$$
(2.14)

and

$$\mathcal{L}_{i,j}(A) := \{ \zeta \in \mathbb{C} : |\zeta - a_{j,j}| < 2|a_{j,i}| - r_j(A) \}.$$
(2.15)

Furthermore, $\widetilde{\Omega}(A) \subseteq \mathcal{G}(A)$.

Remark 2.2.9. In [51], there was a typographical error in the definition of $\mathcal{L}_{i,j}(A)$, " \geq " instead of "<". What we state here is the corrected version by Li et al. [44].

According to Melman, this result was based on a previous result obtained by Parodi and Schneider. We refer to [79, p.73-79] for detailed discussion of the Parodi and Schneider results. The next result is given by Li et al. [44] in which they gave two new Brauer-type sets. We mention only one of them.

Theorem 2.2.10 ([44, Theorem 4]). Let $A = [a_{i,j}] \in \mathbb{C}^{n \times n}$. Then

$$\operatorname{Spec}(A) \subseteq \widetilde{\Phi}(A) = \bigcup_{\substack{i,j=1\\i\neq j}}^{n} \widetilde{\Phi}_{i,j}(A),$$

where

$$\widetilde{\Phi}_{i,j}(A) = \mathcal{K}_{i,j}(A) \setminus \widetilde{\mathcal{L}}_i(A), \qquad \widetilde{\mathcal{L}}_i(A) = \bigcup_{s \in J_i} \widetilde{\mathcal{L}}_{s,i}(A),$$
(2.16)

and

$$\hat{\mathcal{L}}_{s,i}(A) = \{ \zeta \in \mathbb{C} : |\zeta - a_{s,s}| (|\zeta - a_{i,i}| + \widetilde{r}_{i,s}(A)) < (|a_{s,i}| - \widetilde{r}_{s,i}(A)) |a_{i,s}| \}.$$
(2.17)

Furthermore, $\widetilde{\Phi}(A) \subseteq \mathcal{B}(A)$.

Remark 2.2.11. It was pointed out in [44] that there are 3n(n-1)/2 Cassini ovals in the set $\tilde{\Phi}(A)$, whereas n(n-1)/2 ones in the set $\mathcal{K}(A)$. Although $\tilde{\Phi}(A)$ usually gives a better bound than the Brauer set $\mathcal{K}(A)$, it takes more time to compute.

2.2.3 Inclusion regions for the eigenvalues of a linear pencil

The first Gerhgorin-type inclusion region for a generalised eigenvalue problem $A\mathbf{x} = \lambda D\mathbf{x}$ appeared in the paper by Stewart and Sun [70, 71]. Let $A = [a_{i,j}] \in \mathbb{C}^{n \times n}$ and $D = [d_{i,j}] \in \mathbb{C}^{n \times n}$.

Theorem 2.2.12. [71, Theorem 2.4] Let (A, D) be a regular matrix pair, i.e. $det(A - \lambda D)$ does not vanish identically. Let

$$S_i(A,D) = \{\zeta \in \mathbb{C} : |a_{i,i} - \zeta d_{i,i}| \le \sum_{j \in J_i} |a_{i,j} - \zeta d_{i,j}|\}, \qquad i = 1, \dots, n.$$
(2.18)

Then

$$\operatorname{Spec}(A - \lambda D) \subset \bigcup_{i=1}^{n} \mathcal{S}_{i}(A, D).$$

Note that ζ appears on both sides of the inequality in the sets S_i so that it may be difficult to compute S_i in some cases. Therefore, Stewart and Sun [71] gave an alternative bound as follows.

Theorem 2.2.13. [71, Corollary 2.5] Let \mathcal{X} denote the chordal metric defined by

$$\mathcal{X}(x,y) = \frac{|x-y|}{\sqrt{1+|x|^2}\sqrt{1+|y|^2}}.$$
(2.19)

Let

$$\widetilde{\mathcal{S}}_{i}(A,D) \equiv \{\zeta \in \mathbb{C} : \mathcal{X}(\zeta, a_{ii}/d_{ii}) \le \varrho_{i}\},$$
(2.20)

where

$$\varrho_i = \sqrt{\frac{\left(\sum_{j \in J_i} |a_{i,j}|\right)^2 + \left(\sum_{j \in J_i} |d_{i,j}|\right)^2}{|a_{i,i}|^2 + |d_{i,i}|^2}}.$$
(2.21)

Then

$$\mathcal{S}_i \subset \widetilde{\mathcal{S}}_i, \qquad i=1,\ldots,n,$$

so that

$$\operatorname{Spec}(A - \lambda D) \subset \bigcup_{i=1}^{n} \widetilde{\mathcal{S}}_{i}(A, D)$$

2.2.4 Minimal Gershgorin sets

In this section, we review the minimal Gershgorin set which was introduced by Varga in [78]. Its theoretical properties were studied in more detail in [79] and the algorithms for computing the set were developed in [38, 80].

Let $\mathbf{x} = (x_1, \dots, x_n)^T \in \mathbb{R}^n$ be any vector with positive components (which we write as $\mathbf{x} > 0$), and set

$$X := \operatorname{diag}(x_1, \ldots, x_n),$$

so that X is non-singular. It is clear that if $A = [a_{i,j}] \in \mathbb{C}^{n \times n}$ then

$$X^{-1}AX = \left[a_{i,j}\frac{x_j}{x_i}\right].$$

Since $\text{Spec}(A) \equiv \text{Spec}(X^{-1}AX)$, applying the Gershgorin's result (i.e. Theorem 1.3.1) to $X^{-1}AX$ gives that *i*-th Gershgorin disk for $X^{-1}AX$ is given by

$$\mathcal{G}_i^{r_{\mathbf{x}}}(A) := \left\{ \zeta \in \mathbb{C} : |\zeta - a_{i,i}| \le r_{\mathbf{x},i}(A) \right\},\tag{2.22}$$

where $r_{\mathbf{x},i}(A)$ is the *i*-th deleted absolute row sum of $X^{-1}AX$, i.e.

$$r_{\mathbf{x},i}(A) := r_i(X^{-1}AX) = \frac{1}{x_i} \sum_{j \in J_i} |a_{i,j}| x_j.$$
(2.23)

Then the associated Gershgorin set is given by

$$\mathcal{G}^{r_{\mathbf{x}}}(A) := \bigcup_{i=1}^{n} \mathcal{G}_{i}^{r_{\mathbf{x}}}(A),$$

and therefore

$$\operatorname{Spec}(A) \subseteq \mathcal{G}^{r_{\mathbf{x}}}(A).$$

Since this inclusion holds for any x > 0 in \mathbb{R}^n , by intersecting over all x > 0 in \mathbb{R}^n , we obtain

$$\operatorname{Spec}(A) \subseteq \mathcal{G}^{\mathcal{R}}(A) := \bigcap_{\mathbf{x} > \mathbf{0}} \mathcal{G}^{r_{\mathbf{x}}}(A).$$
 (2.24)

The set $\mathcal{G}^{\mathcal{R}}(A)$ is called the minimal Gershgorin set for A, relative to the collection of all row sums $r_{\mathbf{x},i}(A)$, where $\mathbf{x} > \mathbf{0}$ in \mathbb{R}^n .

In other words, for any nonsingular matrix X, the spectrum of A and $X^{-1}AX$ are the same whereas their Gershgorin sets are not, and one may obtain better estimates by considering the minimal Gershgorin set of a matrix. Indeed, the minimal Gershgorin set gives the sharpest localisation of the spectrum of a matrix among all Gershgorin-type sets
(cf. [37]). Nevertheless, calculating this set for a given matrix is not easy. Therefore, Varga et al. [80] and Kostic et al. [38] proposed numerical algorithms for finding an approximation set, however the applicability of these algorithms are limited to the small or medium size matrices.

2.2.5 Gershgorin sets for block matrices

Gershgorin's theorem was extended to a more general setting of partioned matrices by Feingold and Varga [23,79] and to finite matrices of bounded operators on Banach spaces by Salas [65]. The results in this section have more general forms which are applicable to bounded and/or infinite dimensional operators but we will be working on the block matrix X given in (2.2).

It is well-known (see [26, p. 84]) that the norm of the resolvent of a normal operator can be calculated in terms of the distance to the spectrum, i.e. if A is a normal operator, then

$$\left(\|(\zeta I - A)^{-1}\|\right)^{-1} = \operatorname{dist}(\zeta, \operatorname{Spec}(A)) = \inf_{\lambda \in \operatorname{Spec}(A)} |\zeta - \lambda|,$$
(2.25)

where $\zeta \in \mathbb{C} \setminus \operatorname{Spec}(A)$.

The next result is the Gershgorin theorem for the block matrix X, which is given by Tretter [75].

Theorem 2.2.14. [75, Theorem 1.13.1 and Remark 1.13.4] Let $A_{11} = A_{11}^*$, $A_{22} = A_{22}^*$ and suppose that $A_{21} = -A_{12}^*$. Then

(i) It follows that

$$\operatorname{Spec}(\mathbf{X}) \subset \{\zeta \in \mathbb{C} : \operatorname{dist}(\zeta, \operatorname{Spec}(A_{11}) \cup \operatorname{Spec}(A_{22})) \leq ||A_{12}|| \}.$$

(ii) In addition, let

$$\min(\operatorname{Spec}(A_{11})) = a_{-}, \qquad \max(\operatorname{Spec}(A_{11})) = a_{+}, \\ \min(\operatorname{Spec}(A_{22})) = d_{-}, \qquad \max(\operatorname{Spec}(A_{22})) = d_{+}.$$

Then

 $\operatorname{Re}(\operatorname{Spec}(\mathbf{X})) \subset [\min\{a_-, d_-\}, \max\{a_+, d_+\}].$

There are similar generalisations of the Gershgorin-type results in the block matrix case, see [23, 79]. We mention only Brauer's result on Cassini ovals which was given by Freingold and Varga [23] as the following.

Theorem 2.2.15. [23, Theorem 6] Let $A_{11} = A_{11}^*$, $A_{22} = A_{22}^*$ and suppose that either $A_{21} = A_{12}^*$ or $A_{21} = -A_{12}^*$. Then

$$\operatorname{Spec}(\mathbf{X}) \subset \left\{ \lambda \in \mathbb{C} : \left(\inf_{\zeta \in \operatorname{Spec}(A_{11})} |\lambda - \zeta| \right) \left(\inf_{\zeta \in \operatorname{Spec}(A_{22})} |\lambda - \zeta| \right) \le \|A_{12}\|^2 \right\}.$$
(2.26)

2.2.6 A Gershgorin-type result for the spectrum of a two-parameter eigenvalue problem

In Section 3.2, the general concept of the two-parameter eigenvalue problem will be investigated in more details. In here, we briefly introduce the problem and then we give the Gershgorin-type localisation for the problem. A two-parameter eigenvalue problem is

$$\mathbf{M}(\alpha,\beta)\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix} = \begin{pmatrix}A-\alpha & C\\C^* & D-\beta\end{pmatrix}\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix} = \mathbf{0},$$
(2.27)

where A, D are self-adjoint operators in the Hilbert spaces \mathcal{H}_1 , \mathcal{H}_2 , respectively, and C is a linear operator from \mathcal{H}_2 to \mathcal{H}_1 . We say that (α, β) is a pair-eigenvalue of \mathbf{M} if there exist a non-zero element $(\mathbf{u}, \mathbf{v})^T \in \mathcal{H}_1 \oplus \mathcal{H}_2$ satisfying (2.27). The set of all pair-eigenvalues of \mathbf{M} will be denoted by $\operatorname{Spec}_p(\mathbf{M})$. The following result is the Gershgorin-type set for the two-parameter eigenvalue problem obtained by applying to Theorem 2.2.15.

Corollary 2.2.16. For the set of pair-eigenvalues of M defined by (2.27) we have

$$\operatorname{Spec}_{p}(\mathbf{M}) \subset \mathcal{M}(A, D),$$

where

$$\mathcal{M}(A,D) := \left\{ (\alpha,\beta) \in \mathbb{C}^2 : \operatorname{dist}(\alpha,\operatorname{Spec}(A)) \operatorname{dist}(\beta,\operatorname{Spec}(D)) \le \|C\|^2 \right\}$$

Proof of Corollary 2.2.16. Recall the relation between the two-parameter eigenvalue problem (2.27) and the non-self-adjoint problem (1.13). Applying Theorem 2.2.15 to the non-self-adjoint problem (1.13), the set on the RHS of (2.26) becomes

$$\{\lambda \in \mathbb{C} : \operatorname{dist}(\lambda, \operatorname{Spec}(A+c)) \operatorname{dist}(\lambda, \operatorname{Spec}(-D-c)) \leq \|C\|^2 \}.$$

Re-arranging the distance function with account of (1.14) gives

$$\operatorname{dist}(\lambda, \operatorname{Spec}(A + c)) = \operatorname{dist}(\lambda - c, \operatorname{Spec}(A)) = \operatorname{dist}(\alpha, \operatorname{Spec}(A)),$$

and similarly

$$\operatorname{dist}(\lambda, \operatorname{Spec}(-D - c)) = \operatorname{dist}(-\lambda - c, \operatorname{Spec}(D)) = \operatorname{dist}(\beta, \operatorname{Spec}(D)).$$

Hence the statement of the corollary follows.

2.3 Orthogonal polynomials

Let η be a positive Borel measure with support \mathcal{I} defined on \mathbb{R} for which all the moments

$$\eta_k = \int_{\mathcal{I}} x^k d\eta(x), \quad k = 0, 1, 2, \dots,$$

are finite. Let $P_n(x)$ be a polynomial of degree n. Then a sequence of polynomials $\{P_n(x)\}_{n=0}^{\infty}$ is orthogonal with respect to η on \mathcal{I} if

$$\langle P_n, P_m \rangle_\eta = \int_{\mathcal{I}} P_n(x) P_m(x) d\eta(x) = h_n \delta_{mn}, \qquad m, n = 0, 1, 2, \dots,$$

where δ_{mn} is the Kronecker delta and h_n is the square of the weighted L^2 -norm of P_n given by

$$h_n := \langle P_n, P_n \rangle_{\eta} = ||P_n||_{\eta}^2 = \int_{\mathcal{I}} (P_n)^2 d\eta(x) > 0.$$

Let $\omega(x)$ denote a weight function which is continuous and positive on (a, b). Assume the measure η is absolutely continuous with respect to Lebesgue measure and $d\eta(x) = \omega(x)dx$ on some interval (a, b), then $\{P_n(x)\}_{n=0}^{\infty}$ is called orthogonal on (a, b) with respect to the positive weight function $\omega(x)$ if

$$\langle P_n, P_m \rangle_{\omega} = \int_a^b P_n(x) P_m(x) \omega(x) dx = h_n \delta_{mn}, \qquad m, n = 0, 1, 2, \dots$$

If $h_n = 1$ for each n = 0, 1, 2, ..., then the sequence of polynomials is called orthonormal. If $P_n(x) - x^n$ has a degree of at most n - 1, then the sequence of polynomials is said to be monic.

Orthogonal polynomials satisfy the three-term recurrence relations that show their relations with continued fractions.

Theorem 2.3.1. [77, Theorem 2.1] Orthogonal polynomials $P_n(x) = \gamma_n x^n + k_n x^{n-1} + \dots$ on the real line satisfy a three-term recurrence relation

$$xP_n(x) = a_{n+1}P_{n+1}(x) + b_n P_n(x) + a_n P_{n-1}(x),$$
(2.28)

with initial conditions $P_{-1} = 0$ and $P_0 = 1$. The recurrence coefficients are given by

$$a_n = \int x P_{n-1}(x) P(x) d\eta(x) = \frac{\gamma_{n-1}}{\gamma_n} > 0,$$

$$b_n = \int x P_n^2(x) d\eta(x) = \frac{k_n}{\gamma_n} - \frac{k_{n+1}}{\gamma_{n+1}}.$$

See [77] for proof of Theorem 2.3.1.

The orthogonal polynomials are closely related to the Jacobi matrices (or Jacobi operators). These matrices are real, symmetric, tridiagonal matrices with positive off-diagonal entries, and given by

$$\mathcal{J}_{n} = \begin{pmatrix} b_{1} & a_{1} & & \\ a_{1} & b_{2} & a_{2} & & \\ & a_{2} & b_{3} & \ddots & \\ & & \ddots & \ddots & a_{n-1} \\ & & & a_{n-1} & b_{n} \end{pmatrix}, \qquad a_{j} > 0, \quad b_{j} \in \mathbb{R}.$$
(2.29)

The Jacobi matrix \mathcal{J}_n has *n* real distinct eigenvalues and the monic polynomial $P_n(x)/\gamma_n$ is the characteristic polynomial of \mathcal{J}_n . In other words, zeros of P_n are the eigenvalues of \mathcal{J}_n .

2.3.1 Properties of the Chebyshev polynomials of the second kind

Chebyshev polynomials are special cases of the Jacobi polynomials, which are a class of classical orthogonal polynomials. We shall focus our attention only on Chebyshev polynomials of the second kind, denoted by $U_n(x)$. The sequence of polynomials $\{U_n(x)\}_{n=0}^{\infty}$ was first studied by the mathematician P. L. Chebyshev (1821-1894). Formally, they are defined by

$$U_n(x) = \frac{\sin[(n+1)\theta]}{\sin\theta}, \quad \text{when } x = \cos\theta.$$
 (2.30)

If $x \in [-1, 1]$, then one can take the range of the corresponding variable θ as $[0, \theta]$.

Recall the trigonometric identity

$$\sin[(n+1)\theta] + \sin[(n-1)\theta] = 2\cos\theta\sin(n\theta).$$
(2.31)

If we combine the definition (2.30) with (2.31), then it follows that $U_n(x)$ satisfies the recurrence relation:

$$U_0(x) = 1,$$

$$U_1(x) = 2x,$$

$$U_{j+1}(x) = 2xU_j(x) - U_{j-1}(x), \quad j \ge 2.$$

(2.32)

It can be seen from (2.30) that the θ -zeros in $[0, \pi]$ of $sin[(n + 1)\theta]$ must correspond to the *x*-zeros in [-1, 1] of $U_n(x)$, i.e.

$$(n+1)\theta = j\pi, \qquad j = 1, 2, \dots, n.$$

Therefore the zeros of $U_n(x)$ are

$$x_j = \cos\left(\frac{j\pi}{n+1}\right), \qquad j = 1, 2, \dots, n.$$
 (2.33)

Remark 2.3.2. Using the recursion formula (2.32), one can show by induction that $U_n(x)$ is a polynomial of degree n in x with the leading coefficient equal to 2^n .

Denote the monic orthogonal polynomial

$$F_n(x) = U_n(x/2),$$
 (2.34)

then the functions $F_j(x)$ satisfy the recurrence relations:

$$F_0(x) = 1,$$

$$F_1(x) = x,$$

$$F_j(x) = xF_{j-1}(x) - F_{j-2}(x) \text{ for } j \ge 2.$$
(2.35)

In addition, using the Jacobi matrix (2.29), it can be seen that the functions $F_j(x)$ obey the determinant identity

$$F_n(x) = U_n(x/2) = \det\left(H_x^{(n)}\right),\,$$

where $H_x^{(n)}$ is the $n \times n$ tridiagonal matrix

Therefore, the eigenvalues of $H_0^{(n)}$ correspond to two times the zeros of $U_n(x)$, that is,

$$\mu_j^{(n)} = 2\cos\left(\frac{\pi j}{n+1}\right), \qquad j = 1, 2, \dots, n.$$
 (2.37)

We enumerate the eigenvalues $\mu_i^{(n)}$ in decreasing order.

In the next result, it will be shown that the Chebyshev polynomials of the second kind $\{U_n(x)\}_{n=0}^{\infty}$ form an orthogonal polynomial system on [-1,1] with respect to the weight function $\omega(x) = (1-x^2)^{1/2}$.

Lemma 2.3.3. $\{U_n(x)\}_{n=0}^{\infty}$ are orthogonal on [-1, 1], *i.e.*

$$\langle U_i, U_j \rangle_w = \begin{cases} \frac{\pi}{2} & \text{if } i = j, \\ 0 & \text{if } i \neq j, \end{cases}$$

with respect to the weight function $w(x) = (1 - x^2)^{1/2}$.

Proof. If $i \neq j$, then using

$$x = \cos \theta, \qquad dx = -\sin \theta d\theta,$$
 (2.38)

we have

$$\begin{aligned} \langle U_i, U_j \rangle_w &= \int_{-1}^1 U_i(x) U_j(x) (1 - x^2)^{1/2} dx, \\ &= \int_{\pi}^0 \frac{\sin[(i+1)\theta]}{\sin\theta} \frac{\sin[(j+1)\theta]}{\sin\theta} (1 - \cos^2\theta)^{1/2} (-\sin\theta) d\theta, \\ &= \frac{1}{2} \int_0^{\pi} (\cos[(i-j)\theta] - \cos[(i+j+2)\theta]) d\theta, \\ &= \frac{1}{2} \left[\frac{\sin[(i-j)\theta]}{(i-j)} - \frac{\sin[(i+j+2)\theta]}{(i+j+2)} \right]_0^{\pi} = 0. \end{aligned}$$

If j = i, then again by (2.38), we find the normalization which corresponds to $U_j(x)$ as follows:

$$\langle U_i, U_i \rangle_w = \int_{\pi}^0 \frac{\sin^2[(i+1)\theta]}{\sin^2 \theta} \sin \theta(-\sin \theta) d\theta,$$

= $\frac{1}{2} \int_0^{\pi} (1 - \cos[2(i+1)\theta]) d\theta,$
= $\frac{1}{2} \left[\theta - \frac{\sin[2(i+1)\theta]}{[2(i+1)]} \right]_0^{\pi},$
= $\frac{\pi}{2}.$

Corollary 2.3.4. Similarly, by the change of variables, we can derive the orthonormal system

$$\left\{\sqrt{\frac{1}{\pi}}F_n(x)\right\}_{n=0}^{\infty},$$

on [-2, 2] with respect to the weight function $w(x) = (1 - (x/2)^2)^{1/2}$.

It is well-known that if a polynomial has a repeated (double) root, then its derivative has the same root. Therefore, the next result will be useful later for finding the double roots of $U_n(x)$.

Lemma 2.3.5. For n > 0 and $|x| \neq 1$,

$$\frac{d}{dz}U_n(x) = \frac{(n+2)U_{n-1}(x) - nU_{n+1}(x)}{2(1-x^2)}.$$
(2.39)

We omit the proof of Lemma 2.3.5 as it is based on the chain rule and the trigonometric product identity $\sin a \cos b = \frac{1}{2}(\sin(a+b) + \sin(a-b))$. In addition, this lemma is an exercise in Chapter 2.5, problem 15, [48]. See [48] for further properties of Chebyshev polynomials of the second kind.

2.3.2 Extension of Chebyshev polynomials

Using (2.32), the expressions of $U_n(x)$ can be evaluated for any real x, however the transformation $x = \cos\theta$ is not possible outside the interval [-1, 1]. If |x| > 1, then we may take the alternative transformation $x = \cosh \Theta$ with Θ in the range $[0, \infty)$. One can verify that the same polynomials as sums of powers of x are generated by

$$U_n(x) = \frac{\sinh(n+1)\Theta}{\sinh\Theta}$$

We extend the Chebyshev polynomials $U_n(x)$ into a polynomial $U_n(\xi)$ of a complex variable ξ by introducing a related complex variable ω such that

$$\xi = \frac{\omega + \omega^{-1}}{2}.$$
(2.40)

Let $|\omega| = r$ and suppose that ω moves on a circle of radius r centred at the origin. In the case r > 1, we have

$$\omega = re^{i\theta} = r\cos\theta + ir\sin\theta,$$
$$\omega^{-1} = r^{-1}e^{-i\theta} = r^{-1}\cos\theta - ir^{-1}\sin\theta,$$

and so, from (2.40),

$$\xi = a\cos\theta + ib\sin\theta. \tag{2.41}$$

where

$$a = \frac{1}{2} (r + r^{-1}), \qquad b = \frac{1}{2} (r - r^{-1}).$$

Hence the map (2.40) transforms a circle into an ellipse centred at the origin, with major semi-axes *a*, minor semi-axes *b*, and with foci at $\xi = \pm 1$.

If r = 1, i.e. ω moves on the unit circle, then we obtain b = 0 and ellipse collapses into the real interval [-1,1]. However, ξ moves the interval twice as ω travels around the unit circle.

Now, unless z lies on the interval [-1,1], the equation $\omega+\omega^{-1}=2\xi$ has two solutions for ω

$$\omega = \omega_{1,2} = \xi \pm \sqrt{\xi^2 - 1}.$$

However, they are inverses of each other. This means that $\omega \mapsto \xi$ is a two-to-one mapping, with branch points at $\xi = \pm 1$. We need to cut the plane along the interval [-1, 1] so as

to define the complex square root $\sqrt{\xi^2 - 1}$ in a way that it lies in the same quadrant as ξ . Therefore, we choose

$$\omega = \omega_1 = \xi + \sqrt{\xi^2 - 1},$$
(2.42)

so that $|\omega| = |\omega_1| \ge 1$ and $|\omega_2| = |\omega_1^{-1}| < 1$. Then ω depends continuously on ξ along any path that does not intersect the interval [-1, 1].

To extend the Chebyshev polynomials of the second kind, we define ξ by (2.40) and assume $|\omega| = r = 1$. We then obtain by (2.41) that

$$\xi = \cos \theta$$

and so, using (2.42),

$$\omega = \xi + \sqrt{\xi^2 - 1} = e^{i\theta},$$

where we fixed a branch of a square root so that $\sqrt{-1} = i$. Hence $U_n(\xi)$ is now a Chebyshev polynomials in a real variable. Then the definition (2.30) gives

$$U_{n-1}(\xi) = \frac{\sin(n\theta)}{\sin\theta} = \frac{\frac{1}{2}(e^{in\theta} - e^{-in\theta})}{\frac{1}{2}(e^{i\theta} - e^{-i\theta})},$$

which leads us to the general definition, for all $\xi \in \mathbb{C}$,

$$U_{n-1}(\xi) = \frac{\omega^n - \omega^{-n}}{\omega - \omega^{-1}}$$
(2.43)

where

$$\xi = \frac{1}{2} \left(\omega + \omega^{-1} \right).$$

Hence, $U_n(\xi)$ is the analytic continuation of $U_n(x)$ to the complex plane since we extended the values of $U_n(x)$ across a branch cut in the complex plane.

2.3.3 Spectral properties of $H_0^{(n)}$

In this section, we want to construct the basis of eigenfunctions of $H_0^{(n)}$ explicitly. Let $\mu_j^{(n)}$ be the eigenvalues of the matrix H_0 , defined by (2.37), and ψ_j be corresponding eigenvectors, i.e.

$$H_0^{(n)} \psi_j = \mu_j^{(n)} \psi_j, \qquad j = 1, 2, \dots, n.$$
 (2.44)

Lemma 2.3.6. For an arbitrary $n \in \mathbb{N}$, an eigenfunction corresponding to the eigenvalue $\mu_i^{(n)}$ of $H_0^{(n)}$ is

$$\boldsymbol{\psi}_{j} = \begin{pmatrix} \psi_{j,1} \\ \psi_{j,2} \\ \vdots \\ \psi_{j,n} \end{pmatrix} \quad \text{with} \quad \psi_{j,k} = \sqrt{\frac{2}{n+1}} \sin\left(\frac{\pi jk}{n+1}\right), \qquad k = 1, \dots, n.$$
 (2.45)

Proof. Recall the *the Chebyshev method* for finding the $n^{\rm th}$ multiple angle formula for the sine function

$$\sin(n+1)x + \sin(n-1)x = 2\cos x \sin(nx).$$
 (2.46)

The equation (2.44) is written in components as

$$\psi_{j,2} = \mu_j \psi_{j,1},$$

 $\psi_{j,k-1} + \psi_{j,k+1} = \mu_j \psi_{j,k}, \quad k = 2, 3, \dots, n-1,$
 $\psi_{j,n-1} = \mu_j \psi_{j,n}.$

Substituting (2.45) and (2.37) into above, it can be seen that all n equations hold by (2.46).

Note that all eigenvalues of H_0 are simple. If $j \neq k$, then

$$\left. \begin{array}{ll} \langle H_0 \boldsymbol{\psi}_j, \boldsymbol{\psi}_k \rangle &= \mu_j \langle \boldsymbol{\psi}_j, \boldsymbol{\psi}_k \rangle \\ \langle \boldsymbol{\psi}_j, H_0 \boldsymbol{\psi}_k \rangle &= \mu_k \langle \boldsymbol{\psi}_j, \boldsymbol{\psi}_k \rangle \end{array} \right\} \Rightarrow (\mu_j - \mu_k) \langle \boldsymbol{\psi}_j, \boldsymbol{\psi}_k \rangle = 0,$$

which implies that the eigenvector basis $\{\psi_j\}_{j=1}^n$ is orthogonal. Since the matrix $H_0^{(n)}$ is Hermitian, the sequence of eigenfunctions $\{\psi_j\}$ of $H_0^{(n)}$ forms an orthonormal basis by the spectral theorem. Therefore, for a given dimension *n*, the norm of ψ_j is given by

$$\|\psi_j\|^2 = \frac{2}{n+1} \sum_{k=1}^n \sin^2 \frac{\pi j k}{n+1} = 1, \qquad j = 1, \dots, n.$$
 (2.47)

Remark 2.3.7. One can show (2.47) by using the Dirichlet kernel

$$1 + 2\sum_{k=1}^{n} \cos(kx) = \frac{\sin\left[(n + \frac{1}{2})x\right]}{\sin\left(\frac{x}{2}\right)},$$
(2.48)

and the trigonometric identity

$$\sin^2\left(\frac{x}{2}\right) = \frac{1 - \cos x}{2}.$$
 (2.49)

2.3.4 Ratios of Chebyshev polynomials of the second kind

In this section, we state the ratio asymptotics given by B. Simon [66] when applied to the Chebyshev polynomials of the second kind. We will then use these results in a linear pencil problem in Section 4.2.1. Recall that F_n is a monic orthogonal polynomial and the ratio asymptotics we will state is based on the ratio

$$\frac{F_{n+1}(\zeta)}{F_n(\zeta)} = \frac{U_{n+1}(\zeta/2)}{U_n(\zeta/2)}.$$
(2.50)

In addition, recall that the zeros of $F_n(\zeta)$ are the eigenvalues of $H_0^{(n)}$ given by (2.37), and $H_0^{(n)}$ is the Jacobi matrix. Simon obtained the following results.

Theorem 2.3.8 ([66, Theorem 2.1]). For all $\zeta \in \mathbb{C} \setminus [-2, 2]$

$$\lim_{n \to \infty} \frac{F_{n+1}(\zeta)}{F_n(\zeta)} = \frac{\zeta + \sqrt{\zeta^2 - 4}}{2},$$
(2.51)

where the branch of the square root is taken with $\sqrt{\cdots} = \zeta + O(1/\zeta)$ near $\zeta = \infty$.

Theorem 2.3.9 ([66, Theorem 2.2]). Suppose that for one ζ_0 with $\text{Im}(\zeta_0) \neq 0$,

$$\lim_{n \to \infty} \frac{F_{n+1}(\zeta)}{F_n(\zeta)}$$

exists and is finite. Then the only function that can occur as ratio asymptotics is the one in (2.51).

Proposition 2.3.10 ([66, Proposition 2.3]).

(i)

$$\frac{F_{n-1}(\zeta)}{F_n(\zeta)} = \sum_{j=1}^n \frac{a_j^{(n)}}{\zeta - \mu_j^{(n)}},$$
(2.52)

where $\alpha_j^{(n)} > 0$ and

$$\sum_{j=1}^{n} a_j^{(n)} = 1.$$
 (2.53)

(ii) If $\text{Im}\zeta > 0$, then

$$0 < -\mathrm{Im}\left(\frac{F_{n-1}(\zeta)}{F_n(\zeta)}\right) \le \frac{1}{\mathrm{Im}\zeta}$$
(2.54)

and

 $\left|\frac{F_{n-1}(\zeta)}{F_n(\zeta)}\right| \le \frac{1}{\mathrm{Im}\zeta}.$

(iii) If $\text{Im}\zeta > 0$, then

$$\operatorname{Im}(\zeta) \le \operatorname{Im}\left(\frac{F_n(\zeta)}{F_{n-1}(\zeta)}\right)$$

Proof. (i) Since F_n is monic, the ratio can be written as

$$\frac{F_{n-1}(\zeta)}{F_n(\zeta)} = \frac{\prod_{k=1}^{n-1} \left(\zeta - \mu_k^{(n-1)}\right)}{\prod_{j=1}^n \left(\zeta - \mu_j^{(n)}\right)}.$$
(2.55)

Since $F_n(\zeta)/F_{n-1}(\zeta)$ has simple poles and tends to zero at infinity, equation (2.52) holds for some $\alpha_i^{(n)}$. Multiplying equation (2.52) by $\zeta - \mu_i^{(n)}$ we have

$$\frac{\prod_{k=1}^{n-1} \left(\zeta - \mu_k^{(n-1)}\right)}{\prod_{\substack{j=1\\j \neq i}}^n \left(\zeta - \mu_j^{(n)}\right)} = a_i^{(n)} + \left(\sum_{\substack{j=1\\j \neq i}}^n \frac{a_j^{(n)}}{\zeta - \mu_j^{(n)}}\right) \left(\zeta - \mu_i^{(n)}\right).$$

Then taking ζ to $\mu_i^{(n)}$, we obtain

$$a_{i}^{(n)} = \frac{\prod_{k=1}^{n-1} \left(\mu_{i}^{(n)} - \mu_{k}^{(n-1)}\right)}{\prod_{\substack{j=1\\j\neq i}}^{n} \left(\mu_{i}^{(n)} - \mu_{j}^{(n)}\right)}.$$
(2.56)

Note that for any $j = 1, \ldots, n-1$

$$\mu_j^{(n)} < \mu_j^{(n-1)} < \mu_{j+1}^{(n)}.$$

Therefore for i = n, all factors on the right hand side of (2.56) are positive, and when we decrease *i* by one, both numerator and denominator each pick up a minus sign. Hence $a_i^{(n)} > 0$. As $\zeta \to \infty$, it can be seen from (2.55) that the ratio equals to $\zeta^{-1} + O(\zeta^{-2})$. Also from the right side of (2.52) we have that

$$\zeta^{-1}\left(\sum_{j=1}^{n} a_{j}^{(n)}\right) + O\left(\zeta^{-2}\right),$$

so (2.53) holds.

(ii)First, note that for any $b \in \mathbb{R}$ and $\zeta \in \mathbb{C}$ with $\operatorname{Im}(\zeta) > 0$

$$0 < \operatorname{Im}\left(\frac{1}{\zeta - b}\right) \le \left|\frac{1}{\zeta - b}\right| \le \frac{1}{\operatorname{Im}(\zeta)}.$$

Then using (2.52) we obtain

$$\operatorname{Im}\left(\frac{F_{n-1}(\zeta)}{F_n(\zeta)}\right) = \operatorname{Im}\left(\sum_{j=1}^n \frac{a_j^{(n)}}{\zeta - \mu_j^{(n)}}\right) \le \sum_{j=1}^n \frac{a_j^{(n)}}{\operatorname{Im}(\zeta)} = \frac{1}{\operatorname{Im}(\zeta)}.$$

(iii) Re-write the recurrence relation (2.35) as

$$z = \frac{F_n(\zeta)}{F_{n-1}(\zeta)} + \frac{F_{n-2}(\zeta)}{F_{n-1}(\zeta)}.$$

Since $\text{Im}(F_{n-2}(\zeta)/F_{n-1}(\zeta)) < 0$ by (2.54), the result follows by considering the imaginary parts of the equation.

Chapter 3

A two-parameter eigenvalue problem for a class of block-operator matrices

This chapter deals mainly with a two-parameter eigenvalue problem. Before we introduce the problem, we start Section 3.1 by reviewing the linear pencil problem studied by Davies and Levitin [19], which is the main motivation of this chapter. In Section 3.2 we introduce a two-parameter eigenvalue problem and we explain the relation between eigenvalue problems at the end of this section. In Section 3.3, after introducing some notation we give the main results of this chapter concerning the real spectrum of the problem. We present some preliminary results in Section 3.4, and then we shall give the proofs of our results in Section 3.5. To support our results, illustrative examples of all different possibilities are given in Section 3.6. Finally, we will discuss the non-real spectrum of the problem in Section 3.7.

3.1 Basics from a linear pencil problem

One interesting matrix example of a finite-dimensional linear self-adjoint pencil that is sign-indefinite was considered by Davies and Levitin [19]. Namely, let $m, n \in \mathbb{N}$ and N = m + n, and consider the linear operator pencil

$$\mathcal{A}_c = \mathcal{A}_c(\lambda) := H_c^{(N)} - \lambda S_{m,n}, \tag{3.1}$$

where the matrix $H_c^{(N)}$ is defined in (2.36) with $c \in \mathbb{R}$, and $S_{m,n}$ is the diagonal matrix

$$S_{m,n} = \begin{pmatrix} 1 & & & & \\ & \ddots & & & \\ & & 1 & & \\ & & -1 & & \\ & & & \ddots & \\ & & & & -1 \end{pmatrix}.$$
 (3.2)

The size of both matrices is $N \times N$, and the diagonal matrix $S_{m,n}$ has m plus ones and n minus ones. The number λ_0 is said to be an eigenvalue of the pencil \mathcal{A}_c if

$$\left(H_c^{(N)} - \lambda S_{m,n}\right)\mathbf{f} = 0 \tag{3.3}$$

has a non-zero solution $\mathbf{f} \in \mathbb{C}^N$. Spec (\mathcal{A}_c) denotes the *spectrum* of the pencil \mathcal{A}_c which is the set of all its eigenvalues.

In most of our statements, we assume that m = n and N = 2n. We will mostly use 2×2 block structures and so we will often work with the half-size matrix $H_0^{(n)}$. In the notation $H_0^{(n)}$, the superscript (n) represents its size. For simplicity, we often drop the superscript (n) in $H_c^{(n)}$ whenever it is clear from the context, i.e. $H_c := H_c^{(n)}$, and we will state it otherwise. The full matrix size is N and will not be omitted. Recall that the eigenvalues of $H_0^{(n)}$ were denoted by $\mu_j^{(n)}$ in (2.37), and for the same reason, we set $\mu_j := \mu_j^{(n)}$. Moreover, we denote for brevity $S := S_{n,n}$.

We begin with the following result on the localisation of eigenvalues of the pencil A_c .

Lemma 3.1.1 ([19, Lemma 2.1]). All the eigenvalues $\lambda \in \text{Spec}(\mathcal{A}_c)$ satisfy

 $|\lambda| < 2 + |c|.$

Proof. Note that $||H_0|| = \mu_1 = 2\cos(\pi/(N+1)) < 2$. Since $\operatorname{Spec}(\mathcal{A}_c) \equiv \operatorname{Spec}(S^{-1}H_c)$, we have

$$|\lambda| \le ||S^{-1}H_c|| = ||H_c|| = ||H_0 + cI|| < 2 + |c|.$$

The matrix *S* is indefinite. Since $\operatorname{Spec}(H_c^{(N)}) \equiv \operatorname{Spec}(H_0^{(N)} + cI)$ and $\operatorname{Spec}(H_0^{(N)}) = \{\mu_1^{(N)}, \ldots, \mu_N^{(N)}\}$, the matrix $H_c^{(N)}$ is indefinite when $c \in (\mu_N^{(N)}, \mu_1^{(N)}) \subset (-2, 2)$. Therefore, the spectrum $\operatorname{Spec}(\mathcal{A}_c)$ may contain non-real eigenvalues when $|c| < \mu_1^{(N)}$. Davies and Levitin [19] pursued an asymptotic approach for large-size matrices (i.e. $N \to \infty$) and they obtained the asymptotics of the complex eigenvalues of \mathcal{A}_c when the parameter *c* is equal to zero. They proved that when c = 0 all eigenvalues of \mathcal{A}_c approximately lie on the same curve, which is independent of *N*, in coordinates ($\operatorname{Re}(\lambda), \operatorname{NIm}(\lambda)$). The curve is explicitly given in the next result.

Theorem 3.1.2 ([19, Theorem 4.2]). Let c = 0, $n = m = N/2 \rightarrow \infty$. The eigenvalues of A_c are all non-real, and those not lying on the imaginary axis satisfy

$$\operatorname{Im}(\lambda) = \pm \frac{Y_0(|\operatorname{Re}(\lambda)|)}{2n} + o(n^{-1}),$$

$$\operatorname{Re}(\lambda) = \pm 2\cos\left(\frac{2\pi k}{2n+1}\right) + o(n^{-1}),$$

where $k = 1, ..., \left| \frac{n}{2} \right|$ and $\lfloor \cdot \rfloor$ denotes the integer part, and

$$Y_0(y) := \sqrt{4 - y^2} \log \left(\tan \left(\frac{\pi}{4} + \frac{1}{2} \arccos \left(\frac{y}{2} \right) \right) \right).$$

If n is even, then there are no other eigenvalues. If n is odd, there are additionally two purely imaginary eigenvalues at

$$\lambda = \pm i \frac{\log(n)}{n} (1 + o(1)).$$

We omit numerical illustrations concerning the asymptotics given in Theorem 3.1.2. Instead, we illustrate the eigenvalues of A_0 for some *n* in Figure 3.1. For a more detailed discussion on the asymptotics of the complex eigenvalues of A_0 , we refer the reader to [19].

We will see in most of our numerical experiments that there is symmetry in the spectrum $Spec(A_c)$.

Lemma 3.1.3 ([19, Lemma 2.1(a) and Lemma 5.1]). Let n = m. Then the spectrum $\text{Spec}(\mathcal{A}_c)$ is invariant under the symmetry $\lambda \to \overline{\lambda}$ and the symmetry $\lambda \to -\lambda$. Moreover $\text{Spec}(\mathcal{A}_c)$ is symmetric with respect to $c \to -c$.

Remark 3.1.4. As mentioned above, Davies and Levitin [19] found the approximate locations of the eigenvalues of A_c when c = 0. As the central focus of this study is the case |c| < 2 and the spectrum of A_c is invariant under the symmetry $c \to -c$, it will be sufficient to consider only the case 0 < c < 2.



Figure 3.1: Spec(A_0) for n = 170 (black diamonds), n = 300 (blue triangles) and n = 700 (red circles). Left: in coordinates (Re(λ), Im(λ)). Right: in coordinates (Re(λ), 2nIm(λ)).

As can be seen from the left side of Figure 3.1, the spectra asymptotically lie on a particular curve in the complex plane $(\text{Re}(\lambda), \text{Im}(\lambda))$. If one redrew the eigenvalues in coordinates $(\text{Re}(\lambda), N\text{Im}(\lambda))$, one would see that they approximately lie on the same curve, as in the right side of Figure 3.1. Nevertheless, the spectral picture is more complicated when $c \neq 0$ (see the left side of Figure 3.2). The spectrum of \mathcal{A}_c contain both real and non-real eigenvalues and it is hard to believe that the non-real eigenvalues for a fixed c lie on a certain curve.



Figure 3.2: Left: Spec(A_c) for c = 1.5, and n = 170 (white diamonds), n = 300 (blue triangles) and n = 700 (red circles) in coordinates ($\operatorname{Re}(\lambda), \operatorname{Im}(\lambda)$). Right: $\bigcup_{n=1}^{600} \operatorname{Spec}(A_c)$ for c = 1.5 in coordinates ($\operatorname{Re}(\lambda), 2n\operatorname{Im}(\lambda)$).

Nonetheless, there is a common bounding curve if one superimposes all the eigenvalues of the pencil \mathcal{A}_c by taking some values of n with imaginary parts scaled by its size N (see the right side of Figure 3.2). Note that the behaviour in the interior differs for different c. Davies and Levitin [19] derived an explicit expression for the curve which bounds the whole spectrum of \mathcal{A}_c , however the crucial step when proving the result was Conjecture 1.3.3. Namely, Davies and Levitin [19] deduced the following asymptotic estimates when c is non-zero as $N \to \infty$.

Theorem 3.1.5 ([19, Theorem 5.4]). Let 0 < c < 2, $n = m = N/2 \rightarrow \infty$. The eigenvalues of A_c that satisfy

$$|\operatorname{Re}(\lambda)| < 2 - c$$

also satisfy

$$|\mathrm{Im}(\lambda)| \le \frac{Y_c(|\mathrm{Re}(\lambda)|)}{2n} + o(n^{-1}),$$

where

$$Y_c(y) := \mathcal{Y}_{c,y}^{-1}\left(\tan\left(\frac{1}{2}\arccos\left(\frac{y-c}{2}\right)\right) \tan\left(\frac{1}{2}\arccos\left(\frac{y+c}{2}\right)\right)\right),$$

and $\mathcal{Y}_{c,y}^{-1}$ is the inverse of the monotonic increasing analytic function $\mathcal{Y}_{c,y}$: $(0,\infty) \to (0,1)$ defined by

$$\mathcal{Y}_{c,y}(x) := \tanh\left(\frac{x}{2\sqrt{4-(y-c)^2}}\right) \tanh\left(\frac{x}{2\sqrt{4-(y+c)^2}}\right).$$

We shall derive the characteristic polynomial of the pencil A_c explicitly in Section 4.1.1. Now, we will have a look at some open questions from [19] and try to make some progress towards them. We paid a lot of attention to try to prove Conjecture 1.3.3. Proving (1.16) may be harder than proving (1.17). In Section 4.2 we reformulate the harder modulus version of Conjecture 1.3.3 in different forms. We emphasise that extensive numerics confirm Conjecture 1.3.3. One numerical evidence can be seen, for instance, in Figure 1.2, where we show that the maximum value of $|\lambda \pm c|$ (red line) never exceeds 2 (blue dotted line) for n = 9 and n = 100. We conducted some experiments taking *n* between 3 and 300, and taking very small step sizes *c*, and we failed to find a counter example to the conjecture. In Figure 3.3, we illustrate two typical pictures. We also prove this conjecture in the case n = 2 and n = 3 in Section 4.3. We will illustrate in various ways with many other numerical experiments that Conjecture 1.3.3 seems true, however we could not prove it for a fixed *n*.

Conjecture 1.3.3 claims that any non-real eigenvalues λ of the pencil \mathcal{A}_c satisfies both $|\lambda + c| < 2$ and $|\lambda - c| < 2$. First, it is easy to see that two of these conditions cannot be broken simultaneously and we will clarify this case, that is both $|\lambda + c|$ and $|\lambda - c|$ cannot be greater than or equal to 2, in Section 4.2.1. Nevertheless, it is not clear why one of these conditions cannot be broken for the non-real eigenvalues of \mathcal{A}_c . When we apply Gershgorin Theorem to \mathcal{A}_c , we will see in Section 4.2.3 that every eigenvalue of \mathcal{A}_c satisfies at least one of $|\lambda + c| < 2$ and $|\lambda - c| < 2$. Therefore, if we replace "and" by "or" in the conjecture, then the claim would be true by Gershgorin Theorem but then the result holds for all eigenvalues of \mathcal{A}_c .

Remark 3.1.6. There are several ways to treat the pencil problem (3.3). One way is to act by A_c on vectors which we will write as

$$\begin{pmatrix} \mathbf{u} \\ \mathbf{v} \end{pmatrix} = \begin{pmatrix} u_1 \\ \vdots \\ u_n \\ v_n \\ \vdots \\ v_1 \end{pmatrix}.$$
 (3.4)



Figure 3.3: $\max\{|\lambda + c|, |\lambda - c|\}$ (red lines) among all non-real eigenvalues of A_c is drawn as c ranges from 0 to 2 with the step-size 0.001 and the constant line at 2 (blue dotted lines). Left: n = 9. Right: n = 100.

With this convention, the spectral problem for the linear pencil A_c can be written in the block-matrix form as

$$\begin{pmatrix} H_0 - \sigma I & B \\ B & H_0 + \tau I \end{pmatrix} \begin{pmatrix} \mathbf{u} \\ \mathbf{v} \end{pmatrix} = \mathbf{0},$$
(3.5)

where

$$\sigma = \lambda - c; \qquad \tau = \lambda + c. \tag{3.6}$$

Here all sub-blocks are $n \times n$, and $B = B^*$ with $B_{nn} = 1$ and all other entries of B are zeros.

Remark 3.1.7. Since S is invertible, the spectrum of the pencil A_c equals that of the matrix

$$\mathbf{A}_c := S^{-1} H_c^{(N)}. \tag{3.7}$$

Using (3.4), the spectral problem for the block matrix A_c can be written as

$$\mathbf{A}_{c}\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix} = \begin{pmatrix}H_{c} & B\\-B & -H_{c}\end{pmatrix}\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix} = \lambda\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix}.$$
(3.8)

The next result is a special case of Lemma 2.1.3 when $A_{11} = H_0$, $A_{22} = -H_0$, $A_{12} = B$ and $\gamma = -c$.

Corollary 3.1.8. Let $\lambda \in \text{Spec}(\mathbf{A}_c) \setminus \mathbb{R}$ and $(\mathbf{u}, \mathbf{v})^T$ be eigenvectors which correspond to the non-real eigenvalues λ of \mathbf{A}_c . Then

$$\|\mathbf{u}\| = \|\mathbf{v}\|$$
 and $|\operatorname{Im}(\lambda)| \le 1$.

Our original motivation was to try to prove Conjecture 1.3.3 but we could not, nevertheless we still gain some insight and obtain new estimates. The first thing we observed when we considered problem (3.5) in a more general setting, that is,

$$\begin{pmatrix} A - \sigma & \kappa C \\ \kappa C^* & D + \tau \end{pmatrix} \begin{pmatrix} \mathbf{u} \\ \mathbf{v} \end{pmatrix} = 0,$$
(3.9)

where $\kappa \in \mathbb{R}$, $A = A^*$, $D = D^*$ and C has rank one. Indeed, problem (3.9) is a special case of a two-parameter eigenvalue problem which we describe in the next section. Namely, Davies and Levitin [19] consider the case when $\kappa = 1$, $A = H_0^{(n)}$, $D = H_0^{(m)}$, $(C)_{n,1} = 1$ and all other entries of C are zeros. In the next section, we shall introduce the general concept of a two-parameter eigenvalue problem first and then explain the relation between eigenvalue problems.

3.2 Introduction to a two-parameter eigenvalue problem

This section deals with the Multiparameter Eigenvalue Problems (MEPs) which are the generalisation of the one-parameter standard eigenvalue problem $L\mathbf{x} = \lambda \mathbf{x}$ and the generalised one-parameter eigenvalue problem $L\mathbf{x} = \lambda V\mathbf{x}$. MEPs can be written in the following abstract form:

$$L\mathbf{x} = \sum_{i=1}^{k} \lambda_i V_i \mathbf{x},$$
(3.10)

where $\lambda_i \in \mathbb{C}$, i = 1, 2, ..., k, are spectral parameters, and L and V_i are self-adjoint linear operators in some Hilbert space \mathcal{H} . Then $\lambda = (\lambda_1, ..., \lambda_k)$ is called a *multi-parametric eigenvalue* (or *k*-tuple, or *eigentuple*) if there exists an $\mathbf{x} \in \mathcal{H} \setminus \{\mathbf{0}\}$, called an *eigenvector*, such that (3.10) holds.

MEPs arise in numerous applications, in particular in mathematical physics when the method of separation of variables is used to solve boundary value problems for partial differential equations. In the 1960s, an abstract algebraic setting for MEPs was introduced by Atkinson [2, 3], and since then various aspects of MEPs have been investigated by several authors, see for instance [4, 10–12, 67] and references therein.

In the rest of this chapter, we consider a special class of two-parameter eigenvalue problems in a block-operator setting. Let L, V_1 and V_2 be self-adjoint linear operators in some Hilbert space \mathcal{H} . Then a *two-parameter eigenvalue problem* is defined as

$$\mathbf{M}(\alpha,\beta)\mathbf{x} := (\mathbf{L} - \alpha \mathbf{V}_1 - \beta \mathbf{V}_2)\mathbf{x} = 0,$$
(3.11)

where $\alpha, \beta \in \mathbb{C}$ are spectral parameters. We will focus on the case where $\mathcal{H} = \mathcal{H}_1 \oplus \mathcal{H}_2$ for some Hilbert spaces \mathcal{H}_1 and \mathcal{H}_2 , and where L, \mathbf{V}_1 and \mathbf{V}_2 take the particular forms

$$\mathbf{L} = \begin{pmatrix} A & C \\ C^* & D \end{pmatrix}, \qquad \mathbf{V}_1 = \begin{pmatrix} I & 0 \\ 0 & 0 \end{pmatrix}, \qquad \mathbf{V}_2 = \begin{pmatrix} 0 & 0 \\ 0 & I \end{pmatrix}, \qquad (3.12)$$

and

$$\mathbf{x} = egin{pmatrix} \mathbf{u} \ \mathbf{v} \end{pmatrix} \in \mathcal{H}_1 \oplus \mathcal{H}_2.$$

In here, A and D are self-adjoint operators in the Hilbert spaces \mathcal{H}_1 , \mathcal{H}_2 , respectively, and C is a linear operator from \mathcal{H}_2 to \mathcal{H}_1 . Hence one can re-write the two-parameter eigenvalue problem (3.11) as

$$\mathbf{M}(\alpha,\beta)\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix} = \begin{pmatrix}A-\alpha & C\\C^* & D-\beta\end{pmatrix}\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix} = \mathbf{0}.$$
(3.13)

We call $(\alpha, \beta) \in \mathbb{C}^2$ a *pair-eigenvalue* of $\mathbf{M} = \mathbf{M}(\alpha, \beta)$ if there exists a non-trivial solution $\mathbf{x} = \begin{pmatrix} \mathbf{u} \\ \mathbf{v} \end{pmatrix} \in \mathcal{H}$ of (3.13). Such a vector \mathbf{x} is called an *eigenvector* corresponding to the pair-eigenvalue (α, β) . We denote by $\operatorname{Spec}_p(\mathbf{M})$ the set of all pair-eigenvalues of \mathbf{M} . If both $\alpha, \beta \in \mathbb{R}$, then we will call (α, β) a *real pair-eigenvalue* of (3.13).

The equation (3.13) can be re-written as

$$(A - \alpha)\mathbf{u} = -C\mathbf{v},\tag{3.14}$$

$$(D-\beta)\mathbf{v} = -C^*\mathbf{u}.$$
(3.15)

If $\alpha \notin \operatorname{Spec}(A)$, then (3.14) can be re-written as $\mathbf{u} = -(A - \alpha)^{-1}C\mathbf{v}$, and substituting this into (3.15) yields

$$(D - C^*(A - \alpha)^{-1}C)\mathbf{v} = \beta \mathbf{v}.$$
(3.16)

This also means that if $\alpha \notin \operatorname{Spec}(A)$ and $\beta(\alpha)$ is an eigenvalue of

$$D - C^* (A - \alpha)^{-1} C,$$

then $(\alpha, \beta(\alpha)) \in \operatorname{Spec}_p(\mathbf{M})$.

The connection between a two-parameter eigenvalue problem (3.13) and a linear pencil problem (1.2) was obtained in Section 1.3 by setting

$$\alpha = \lambda - c; \quad \beta = -\lambda - c. \tag{3.17}$$

With this convention, we see that any linear pencil problem of the form (1.10) (or a non-self-adjoint problem of the form (1.13)) can be transformed into the two-parameter eigenvalue of the form (3.13).

Indeed, it can be said that Davies and Levitin [19] consider the two-parameter eigenvalue problem (3.13) in the case when $A = H_0^{(n)}$, $D = H_0^{(m)}$, $(C)_{n,1} = 1$ and all other entries of *C* are zeros. In this section, we set (α, β) as general spectral parameters of a two-parameter eigenvalue problem M. In this setting, the corresponding change of notations will be

$$\alpha = \sigma; \qquad \beta = -\tau.$$

We shall plot the pair-eigenvalues of M in the (α, β) -plane, whereas σ and τ is fixed as in (3.6) throughout this thesis.

3.3 Basics and statements

In the rest of this chapter, we investigate the real pair-eigenvalues (α, β) of the twoparametric eigenvalue problem (3.13) with starting additional restrictions in Section 3.3.1. We discuss the non-real pair-eigenvalues of M only in Section 3.7. We also give some heuristics for the localisation of the eigenvalues of the pencil A_c in the aspect of a two parametric eigenvalue problem in Section 5.2.

3.3.1 Restrictions and notation

Suppose that \mathcal{H}_1 and \mathcal{H}_2 are finite dimensional, and therefore we are dealing with matrices. Take $\dim(\mathcal{H}_1) = n$ and $\dim(\mathcal{H}_2) = m$. First, we focus our attention to only the real pair-eigenvalues of M. Our main results (Theorem 3.3.6 and its special case Theorem 3.3.4) are stated below.

Remark 3.3.1. Most of the results in the rest of this chapter have been published in [41]. Note that Levitin and Ozturk [41] deal with the case when n = m. What we discuss in here is the expanded version of [41].

Unless otherwise stated, we set

$$i = 1, \dots, n, \quad k = 1, \dots, n-1, \quad j = 1, \dots, m, \quad l = 1, \dots, m-1$$

which will be used as subscripts. The eigenvalues of A and D will be denoted by

$$\alpha_1 \leq \ldots \leq \alpha_n, \qquad \beta_1 \leq \ldots \leq \beta_m,$$

respectively, and their corresponding eigenvectors will be denoted by φ_i and ψ_j , respectively.

In stating most of our results, we restrict our attention to the case where *C* has rank one. More specifically, we choose two vectors $z \in \mathbb{R}^n$ and $w \in \mathbb{R}^m$ (the real one are chosen for simplicity), and set

$$C\mathbf{v} = \kappa \langle \mathbf{v}, \mathbf{w} \rangle \mathbf{z}, \quad \text{for all} \quad \mathbf{v} \in \mathcal{H}_2,$$

where $\|\mathbf{z}\| = \|\mathbf{w}\| = 1$ and $\kappa \ge 0$, and similarly

$$C^*\mathbf{u} = \kappa \langle \mathbf{u}, \mathbf{z} \rangle \mathbf{w}, \quad \text{for all} \quad \mathbf{u} \in \mathcal{H}_1.$$

The entries of the matrix *C* will be $C_{i,j} = \kappa \mathbf{z}_i \mathbf{w}_j$.

Remark 3.3.2. In [41], where $\mathcal{H}_1 = \mathcal{H}_2$, we take $C = \kappa P$, where $\kappa \in \mathbb{R}$, and P is a projection onto a one-dimensional subspace $Z = \text{Span}\{\mathbf{z}\}, \mathbf{z} \in \mathbb{R}, \|\mathbf{z}\| = 1$. Then, in the basis $\{\varphi_j\}, P$ will have the matrix representation $(\langle \mathbf{z}, \varphi_i \rangle \langle \mathbf{z}, \varphi_j \rangle)_{i,j=1}^n$.

Let $\Phi_{X,\lambda}$ denote the eigenspace of a self-adjoint operator *X* corresponding to an eigenvalue λ , simple or multiple. Further denote, for any $\mathbf{f} \in \mathbb{R}^n$ with $\|\mathbf{f}\| = 1$,

$$\Gamma_{X,\mathbf{f}} := \{ \lambda \in \operatorname{Spec}(X) \mid \exists \phi \in \Phi_{X,\lambda} : \phi \neq 0 \text{ and } \langle \mathbf{f}, \phi \rangle = 0 \},$$

$$\widetilde{\Gamma}_{X,\mathbf{f}} := \{ \lambda \in \operatorname{Spec}(X) \mid \langle \mathbf{f}, \phi \rangle = 0 \ \forall \phi \in \Phi_{X,\lambda} \}.$$

Here, X stands for either A or D, and f stands for either z or w. For simplicity, denote

$$\Gamma_A = \Gamma_{A,\mathbf{z}}, \qquad \widetilde{\Gamma}_A = \widetilde{\Gamma}_{A,\mathbf{z}},$$
$$\Gamma_D = \Gamma_{D,\mathbf{w}}, \qquad \widetilde{\Gamma}_D = \widetilde{\Gamma}_{D,\mathbf{w}}.$$

Note that $\widetilde{\Gamma}_{X,\mathbf{f}} \subseteq \Gamma_{X,\mathbf{f}}$. If λ is a simple eigenvalue of X, then $\dim(\Phi_{X,\lambda}) = 1$, and therefore $\lambda \in \Gamma_{X,\mathbf{f}} \iff \lambda \in \widetilde{\Gamma}_{X,\mathbf{f}}$. Also, $\Gamma_{X,\mathbf{f}}$ contains all the multiple eigenvalues of X. This is due to the fact that in any linear subspace of dimension greater than one, one can find a non-trivial linear combination which is orthogonal to a given vector.

The orthogonal projection $P_{\mathbf{f}}$ of \mathcal{H} onto $\mathcal{S} = \operatorname{Span}{\{\mathbf{f}\}}$ is a linear operator such that $P_{\mathbf{f}}\mathbf{y} \in \mathcal{S}$ for all $\mathbf{y} \in \mathcal{H}$, given by

$$P_{\mathbf{f}}\mathbf{y} = \langle \mathbf{y}, \mathbf{f} \rangle \mathbf{f}.$$

Let $Q_{\mathbf{f}} := I - P_{\mathbf{f}}$ be the orthogonal projection onto \mathcal{S}^{\perp} . For a self-adjoint operator $X : \mathcal{H} \to \mathcal{H}$, denote

$$\begin{split} X_{\parallel,\parallel} &= P_{\mathbf{f}} X|_{\mathcal{S}} : \mathcal{S} \to \mathcal{S}, \qquad X_{\perp,\parallel} = P_{\mathbf{f}} X|_{\mathcal{S}^{\perp}} : \mathcal{S}^{\perp} \to \mathcal{S}, \\ X_{\parallel,\perp} &= Q_{\mathbf{f}} X|_{\mathcal{S}} : \mathcal{S} \to \mathcal{S}^{\perp}, \quad X_{\perp,\perp} = Q_{\mathbf{f}} X|_{\mathcal{S}^{\perp}} : \mathcal{S}^{\perp} \to \mathcal{S}^{\perp}. \end{split}$$

 $P_{\mathbf{z}}$ and $P_{\mathbf{w}}$ will represent the orthogonal projection of \mathcal{H}_1 and \mathcal{H}_2 , onto $Z = \text{Span}\{\mathbf{z}\}$ and $W = \text{Span}\{\mathbf{w}\}$, respectively. The eigenvalues of $A_{\perp,\perp}$ and $D_{\perp,\perp}$ will be denoted by

$$\widehat{\alpha}_1 \leq \ldots \leq \widehat{\alpha}_{n-1}, \qquad \widehat{\beta}_1 \leq \ldots \leq \widehat{\beta}_{m-1},$$

respectively, and their corresponding eigenvectors will be denoted by $\hat{\varphi}_k$ and $\hat{\psi}_l$, respectively.

Lemma 3.3.3. The eigenvalues of X and $X_{\perp,\perp}$ always interlace, i.e. let $\lambda \in \text{Spec}(X)$ and $\widehat{\lambda} \in \text{Spec}(X_{\perp,\perp})$, then for any $j \in \{1, \ldots, \dim(X) - 1\}$,

$$\lambda_j \le \widehat{\lambda}_j \le \lambda_{j+1}.$$

3.3.2 Statement of the simple Chess Board Theorem

Assume for the moment that $\Gamma_A = \Gamma_D = \emptyset$, which in particular implies that all the eigenvalues of *A* and *D* are simple. Denote

$$x_0 := -\infty, \qquad x_{2n} := \infty, \qquad x_{2i-1} := \alpha_i, \qquad x_{2k} := \widehat{\alpha}_k,$$

and similarly for β ,

$$y_0 := -\infty, \qquad y_{2m} := \infty, \qquad y_{2j-1} := \beta_j, \qquad y_{2l} := \beta_l$$

Then, the numbers x_0, \ldots, x_{2n} divide the α -line into 2n intervals, finite or infinite, and similarly the numbers y_0, \ldots, y_{2m} divide the β -line into 2m intervals. Combination of these lines divides the (α, β) -plane into rectangles, some of them semi-infinite,

$$\mathcal{R}_{p,q} := r_p \times r_q, \quad r_p := (x_{p-1}, x_p), \quad r_q := (y_{q-1}, y_q),$$

where $p = 1, \ldots, 2n$ and $q = 1, \ldots, 2m$, see Figure 3.4.

Theorem 3.3.4 (The Simple Chess Board Theorem). Let $\Gamma_A = \Gamma_D = \emptyset$. Then all the real pair-eigenvalues (α, β) of M lie on a family of curves $(\alpha, \beta(\alpha))$ with the following properties:

- (a) each curve may pass only through rectangles $\mathcal{R}_{p,q}$ with p + q even;
- (b) each curve may cross from rectangle to rectangle only through the corner points (x_p, y_q) with p + q odd;



Figure 3.4: In the (α, β) -plane, black dotted and red dot-dashed lines in the vertical direction represent $\alpha = \alpha_i$ and $\alpha = \hat{\alpha}_k$, respectively; in the horizontal direction they represent $\beta = \beta_j$ and $\beta = \hat{\beta}_l$, respectively. Here n = 4 and m = 3, and the rectangles $\mathcal{R}_{p,q}$ with p + q even are shaded.

- (c) each curve β(α) is continuous in α except at eigenvalues of A; at each eigenvalue of A exactly one curve blows up in the following sense: β(α) → ±∞ as α → α_i ± 0, α_i ∈ Spec(A);
- (d) each curve $\beta(\alpha)$ is monotone decreasing in α on its domain of continuity; more precisely, we have

$$\frac{d\beta}{d\alpha} = -\kappa^2 \frac{\langle (A-\alpha)^{-2} \mathbf{z}, \mathbf{z} \rangle (\langle (D-\beta)^{-1} \mathbf{w}, \mathbf{w} \rangle)^2}{\langle (D-\beta)^{-2} \mathbf{w}, \mathbf{w} \rangle} < 0.$$
(3.18)

Remark 3.3.5. As α and β are in fact interchangeable, Theorem 3.3.4 can be equivalently reformulated in terms of curves $(\alpha(\beta), \beta)$ with the only modification being that exactly one curve $\alpha(\beta)$ blows up at each eigenvalue of *B* in the sense that $\alpha(\beta) \to \pm \infty$ as $\beta \to \beta_j \pm 0$, $\beta_j \in \text{Spec}(A)$.

3.3.3 Statement of the full Chess Board Theorem

In this section, we assume that either $\Gamma_A \neq \emptyset$ or $\Gamma_D \neq \emptyset$. Denote additionally, for $X : \mathcal{H} \to \mathcal{H}$,

$$\Delta_X := \{\lambda \in \operatorname{Spec}(X) \mid \lambda \in \operatorname{Spec}(X) \cap \operatorname{Spec}(X_{\perp,\perp}) \text{ and } \dim(\Phi_{X_{\perp,\perp},\lambda}) > \dim(\Phi_{X,\lambda})\}.$$

We will state formally an analogue of Theorem 3.3.4 below, but we start with summarising the principle changes: first, we exclude from the dividing mesh the points of $\tilde{\Gamma}_A \setminus \Delta_A$ and $\tilde{\Gamma}_D \setminus \Delta_D$; and secondly, the real pair-spectrum of \mathbf{M} will, in addition to the curves, contain the lines ($\Gamma_A \times \mathbb{R}$) and ($\mathbb{R} \times \Gamma_D$).

More precisely, let x_i , i = 1, ..., s, denote the points of

$$\left((\operatorname{Spec}(A) \cup \operatorname{Spec}(A_{\perp,\perp})) \setminus \widetilde{\Gamma}_A \right) \cup \Delta_A$$

enumerated in increasing order without account of multiplicities, and similarly y_j , j = 1, ..., t, denote the points of the analogue for D enumerated in increasing order without account of multiplicities. Set additionally $x_0 = y_0 = -\infty$, $x_{s+1} = y_{t+1} = +\infty$, and

$$\mathcal{R}_{p,q} = (x_{p-1}, x_p) \times (y_{q-1}, y_q), \quad p = 1, \dots, s+1, \quad q = 1, \dots, t+1.$$

Theorem 3.3.6 (The Full Chess Board Theorem). All the real pair-eigenvalues (α, β) of M lie either on the straight lines $(\Gamma_A \times \mathbb{R}) \cup (\mathbb{R} \times \Gamma_D)$ or on a family of curves $(\alpha, \beta(\alpha))$ with the following properties:

- (a) each curve may pass only through rectangles $\mathcal{R}_{p,q}$ with p + q even;
- (b) each curve may cross from rectangle to rectangle only through the corner points (x_p, y_q) with p + q odd;
- (c) each curve β(α) is continuous in α except at eigenvalues of A not belonging to Γ_A; at each such eigenvalue of A exactly one curve blows up in the following sense:
 β(α) → ±∞ as α → α_i ± 0, α_i ∈ Spec(A);
- (d) each curve $\beta(\alpha)$ is monotone decreasing in α on its domain of continuity with (3.18).

Remark 3.3.7. The equation (3.16) is real analytic, therefore the solutions of (3.16) are also real analytic locally except at the points where the curves cross. We will see in Lemma 3.5.4 that when $\alpha \in \operatorname{Spec}(A) \setminus \Gamma_A$ there are m - 1 corresponding (α, β) paireigenvalues of M which we can locate. In addition, as mentioned in Theorem 3.3.6(c), one curve blows up. Since each curve $\beta(\alpha)$ is monotone decreasing in α , it can be said that each curve lies its individual sequence of black rectangles, plus the one which blows up. Therefore these curves do not intersect as there is only one curve in each rectangle $\mathcal{R}_{p,q}$ with p + q even. Thus, for every $\alpha \in \mathbb{C} \setminus \operatorname{Spec}(A)$, there are m complex values $\beta(\alpha)$, and the corresponding curves can be chosen continuously in α since each curve is locally analytic in α . However, we note that we do not take into account of additional vertical (or horizontal) straight lines which are not in the mesh.

3.3.4 Limit cases

Only in this subsection, we will be denoting \mathbf{M} by \mathbf{M}_{κ} to indicate its dependence on κ . We state that when $\kappa \to 0$, the components of the real pair-eigenvalues of \mathbf{M}_{κ} approach the eigenvalues of A and D, and when $\kappa \to \infty$, they approach the eigenvalues of $A_{\perp,\perp}$ and $D_{\perp,\perp}$. The proof of the next result will be given in Section 3.5.

Theorem 3.3.8. As $\kappa \to 0$, the real pair-eigenvalue spectrum of \mathbf{M}_{κ} converges to

 $\mathcal{E}_{A,D} := (\operatorname{Spec}(A) \times \mathbb{R}) \cup (\mathbb{R} \times \operatorname{Spec}(D))$

in the sense that $\forall \epsilon > 0 \ \exists \kappa_0 > 0$ such that $\forall (\alpha, \beta) \in \operatorname{Spec}_n(\mathbf{M}_{\kappa}) \cap \mathbb{R}^2$ we have

dist $((\alpha, \beta), \mathcal{E}_{A,D}) < \epsilon$

whenever $\kappa < \kappa_0$.

As $\kappa \to \infty$, the real pair-eigenvalue spectrum of \mathbf{M}_{κ} converges to

$$\widetilde{\mathcal{E}}_{A,D} := (\operatorname{Spec}(A_{\perp,\perp}) \times \mathbb{R}) \cup (\mathbb{R} \times \operatorname{Spec}(D_{\perp,\perp}))$$

in the sense that $\forall \epsilon > 0 \ \exists \kappa_0 > 0$ such that $\forall (\alpha, \beta) \in \operatorname{Spec}_p(\mathbf{M}_{\kappa}) \cap \mathbb{R}^2$ we have

dist
$$\left((\alpha,\beta),\widetilde{\mathcal{E}}_{A,D}\right)<\epsilon$$

whenever $\kappa > \kappa_0$.

3.4 Auxiliary results

The statements in this section are for a single matrix X, and mostly very elementary. We shall use them later in the proof of the Chess Board Theorem. Let

$$\lambda_j \in \operatorname{Spec}(X), \quad \phi_j \in \Phi_{X,\lambda_j}, \quad j = 1, \dots, \dim(X),$$

and

$$\widehat{\lambda}_k \in \operatorname{Spec}(X_{\perp,\perp}), \quad \widehat{\phi}_k \in \Phi_{X_{\perp,\perp},\lambda_k}, \quad k = 1, \dots, \dim(X) - 1.$$

We shall frequently use the Fourier representation of the resolvent, for any $f \in \mathbb{R}^n$ with $\|f\| = 1$,

$$(X - \lambda)^{-1} \mathbf{f} = \sum_{j} \frac{\langle \mathbf{f}, \phi_j \rangle}{\lambda_j - \lambda} \phi_j, \qquad \lambda \notin \operatorname{Spec}(X).$$
 (3.19)

We also set

$$R_X(\lambda, \mathbf{f}) := \langle (X - \lambda)^{-1} \mathbf{f}, \mathbf{f} \rangle = \sum_j \frac{|\langle \mathbf{f}, \phi_j \rangle|^2}{\lambda_j - \lambda}.$$
(3.20)

Lemma 3.4.1. Let $\lambda \notin \operatorname{Spec}(X)$ and $\mathbf{f} \neq 0$. Then

$$R_X(\lambda, \mathbf{f}) = 0 \quad \Leftrightarrow \quad \lambda \in \operatorname{Spec}(X_{\perp, \perp}) \quad \text{and} \quad (X - \lambda)^{-1}\mathbf{f} = c\widehat{\phi},$$

where $\widehat{\phi}$ is an eigenfunction of $X_{\perp,\perp}$ corresponding to λ and $c \neq 0$.

Proof. Set $\zeta = (X - \lambda)^{-1} \mathbf{f}$. Then

$$(X - \lambda)\boldsymbol{\zeta} = \mathbf{f} \Leftrightarrow \begin{pmatrix} X_{\perp,\perp} - \lambda & X_{\parallel,\perp} \\ X_{\perp,\parallel} & X_{\parallel,\parallel} - \lambda \end{pmatrix} \begin{pmatrix} \mathbf{q} \\ \mathbf{p} \end{pmatrix} = \begin{pmatrix} \mathbf{0} \\ \mathbf{f} \end{pmatrix},$$
(3.21)

where $\mathbf{q} = Q_{\mathbf{f}}\boldsymbol{\zeta}$ and $\mathbf{p} = P_{\mathbf{f}}\boldsymbol{\zeta}$. Note that $R_X(\lambda, \mathbf{f}) = \langle \boldsymbol{\zeta}, \mathbf{f} \rangle = 0$ iff $\mathbf{p} = P_{\mathbf{f}}\boldsymbol{\zeta} = 0$. Substituting this into (3.21) gives us

$$\begin{cases} (X_{\perp,\perp} - \lambda)\mathbf{q} &= \mathbf{0}, \\ X_{\perp,\parallel}\mathbf{q} &= \mathbf{f}. \end{cases}$$
(3.22)

By the second equation, \mathbf{q} is non-zero, and then by the first equation $\lambda \in \operatorname{Spec}(X_{\perp,\perp})$ and $\mathbf{q} = c\widehat{\phi}$, with $c \neq 0$. Also, we have

$$\mathbf{q} = Q_{\mathbf{f}}\boldsymbol{\zeta} = (I - P_{\mathbf{f}})\boldsymbol{\zeta} = \boldsymbol{\zeta},$$

and so

$$\boldsymbol{\zeta} = (X - \lambda)^{-1} \mathbf{f} = c \widehat{\boldsymbol{\phi}}.$$

Lemma 3.4.2. $\lambda \in \Gamma_X$ if and only if $\lambda \in \text{Spec}(X) \bigcap \text{Spec}(X_{\perp,\perp})$.

Proof. If there exits an $\lambda \in \Gamma_X$, then there is an eigenfunction $\phi \in \Phi_{X,\lambda}$ such that $\langle \mathbf{f}, \phi \rangle = 0$, and therefore $P_{\mathbf{f}}\phi = 0$ and so $Q_{\mathbf{f}}\phi = \phi$. Thus

$$X_{\perp,\perp}\phi = Q_{\mathbf{f}}X\phi = \lambda Q_{\mathbf{f}}\phi = \lambda\phi,$$

and so $\lambda \in \operatorname{Spec}(X) \bigcap \operatorname{Spec}(X_{\perp,\perp})$.

On the other hand, let $\lambda \in \operatorname{Spec}(X) \bigcap \operatorname{Spec}(X_{\perp,\perp})$. Then

$$X\phi = \lambda\phi \quad \Rightarrow \quad \langle X\phi, \widehat{\phi} \rangle = \lambda \langle \phi, \widehat{\phi} \rangle = \lambda \langle Q_{\mathbf{f}}\phi, \widehat{\phi} \rangle.$$
(3.23)

Also, since $\widehat{\phi} \perp \mathbf{f}$,

$$X\widehat{\phi} = X(\widehat{\phi} + 0\mathbf{f}) = X_{\perp,\perp}\widehat{\phi} + X_{\perp,\parallel}\widehat{\phi},$$
(3.24)

therefore

$$\langle X\phi, \widehat{\phi} \rangle = \langle \phi, X\widehat{\phi} \rangle = \langle \phi, X_{\perp,\perp}\widehat{\phi} \rangle + \langle \phi, X_{\perp,\parallel}\widehat{\phi} \rangle$$

$$= \langle Q_{\mathbf{f}} \boldsymbol{\phi}, X_{\perp,\perp} \widehat{\boldsymbol{\phi}} \rangle + \langle P_{\mathbf{f}} \boldsymbol{\phi}, X_{\perp,\parallel} \widehat{\boldsymbol{\phi}} \rangle$$
$$= \lambda \langle Q_{\mathbf{f}} \boldsymbol{\phi}, \widehat{\boldsymbol{\phi}} \rangle + \langle P_{\mathbf{f}} \boldsymbol{\phi}, X_{\perp,\parallel} \widehat{\boldsymbol{\phi}} \rangle$$

which implies by (3.23) that $\langle P_{\mathbf{f}}\phi, X_{\perp,\parallel}\widehat{\phi}\rangle = 0$. Now, if $P_{\mathbf{f}}\phi = 0$, then $\phi \perp \mathbf{f}$ so that $\lambda \in \Gamma_X$. If $X_{\perp,\parallel}\widehat{\phi} = 0$, then we have from (3.24) that $X\widehat{\phi} = \lambda\widehat{\phi}$, and therefore $\widehat{\phi}$ is an eigenfunction of X such that $\widehat{\phi} \perp \mathbf{f}$, so that $\lambda \in \Gamma_X$.

Lemma 3.4.3. If $\lambda \in \text{Spec}(X) \setminus \widetilde{\Gamma}_X$, then $R_X(t, \mathbf{f})$ has a singularity at $t = \lambda$. The function $R_X(t, \mathbf{f})$ changes sign when t passes through an λ_j , or an $\widehat{\lambda}_k$.

If $\lambda \in \widetilde{\Gamma}_X$, then $(X - \lambda)^{-1}\mathbf{f}$ exists, and $R_X(t, \mathbf{f})$ is continuous at $t = \lambda$. It changes sign at this λ if and only if additionally $\lambda \in \Delta_X$.

Proof. If $\lambda_j \in \operatorname{Spec}(X) \setminus \widetilde{\Gamma}_X$, then there exits at least one $\phi_j \in \Phi_{X,\lambda}$ such that $\langle \mathbf{f}, \phi_j \rangle \neq 0$, and it can be seen from (3.20) that $R_X(t, \mathbf{f})$ goes to $\pm \infty$ as $\lambda \to \lambda_j \mp 0$. Furthermore, since $R_X(t, \mathbf{f})$ has zeros at $\lambda = \widehat{\lambda}_k$ by Lemma 3.4.1, and also is a continuous function except at the poles $\lambda = \lambda_j$, it changes sign every time *z* passes through $\widehat{\lambda}_j$ as well.

The second statement follows immediately from (3.20) and the fact that $f \perp \Phi_{X,\lambda}$.

To show the last statement, we consider the following decomposition

$$\mathcal{H} = \Phi_{X,\lambda} \oplus Q_{\mathbf{f}} \Phi_{X,\lambda}^{\perp} \oplus P_{\mathbf{f}} \Phi_{X,\lambda}^{\perp}$$

where $\Phi_{X,\lambda}^{\perp}$ is the orthogonal complement of $\Phi_{X,\lambda}$ with respect to λ . Let $\dim(\Phi_{X,\lambda}) = d$ and therefore $\dim(Q_{\mathbf{f}}\Phi_{X,\lambda}^{\perp}) = \dim(X) - d - 1$ and $\dim(P_{\mathbf{f}}\Phi_{X,\lambda}^{\perp}) = 1$. Let $\widetilde{X} = X|_{\Phi_{X,\lambda}^{\perp}}$, then

$$X = \begin{pmatrix} \operatorname{diag}(\lambda, \dots, \lambda) & 0 & 0\\ 0 & \widetilde{X}_{\perp, \perp} & \widetilde{X}_{\parallel, \perp}\\ 0 & \widetilde{X}_{\perp, \parallel} & \widetilde{X}_{\parallel, \parallel} \end{pmatrix},$$

where the left corner block has dimension *d*. If $\lambda \in \Delta_X$ then $\lambda \in \text{Spec}(X) \cap \text{Spec}(X_{\perp,\perp})$ such that $\dim(\Phi_{X_{\perp,\perp},\lambda}) > \dim(\Phi_{X,\lambda}) = d$. Thus,

$$\lambda \in \Delta_X \Leftrightarrow \lambda \in \operatorname{Spec}(\widetilde{X}_{\perp,\perp}) \text{ and } \lambda \notin \operatorname{Spec}(\widetilde{X}).$$

Therefore $R_{\tilde{X}}(t, \mathbf{f})$ has zeros at $t = \lambda \in \Delta_X$ by Lemma 3.4.1, and the function $R_{\tilde{X}}(t, \mathbf{f})$ changes sign at this λ .

Proof of Lemma 3.3.3. This lemma is known as Cauchy interlacing theorem. The proof is based on the variational principle. In finite dimensional case,

$$\langle X_{\perp,\perp}\mathbf{f},\mathbf{f}\rangle = \langle Q_{\mathbf{f}}X\mathbf{f},\mathbf{f}\rangle = \langle X\mathbf{f},Q_{\mathbf{f}}\mathbf{f}\rangle = \langle X\mathbf{f},\mathbf{f}\rangle,$$

since $Q_{\mathbf{f}} = Q_{\mathbf{f}}^*$ and $Q_{\mathbf{f}}\mathbf{f} = \mathbf{f}$, so the forms coincide, however the space is different. According to min-max theorem,

$$\lambda_j = \min_{U_1} \{ \max_{\mathbf{f}} \{ \langle X\mathbf{f}, \mathbf{f} \rangle \mid \mathbf{f} \in U_1 \setminus \{0\} \} \mid \dim(U_1) = j \},$$
$$\widehat{\lambda}_j = \min_{U_2} \{ \max_{\mathbf{f}} \{ \langle X_{\perp,\perp}\mathbf{f}, \mathbf{f} \rangle \mid \mathbf{f} \in U_2 \setminus \{0\} \} \mid \dim(U_2) = j \}$$

We have the subspaces U_1 and U_2 of the same dimension, however U_1 contains a bigger collection of subspaces (it may additionally include elements from $P_f f$). Therefore, by taking min over bigger set may be smaller, i.e.

$$\lambda_j \leq \widehat{\lambda}_j.$$

However, $\lambda_j = \hat{\lambda}_j$ if and only if U_1 contains elements only from \mathbf{f}_{\perp} .

Looking at the other direction,

$$\begin{aligned} \lambda_{j+1} &= \max_{U_1} \{ \min_{\mathbf{f}} \{ \langle X \mathbf{f}, \mathbf{f} \rangle \mid \mathbf{f} \in U_1 \setminus \{0\} \} \mid \dim(U_1) = n - j \}, \\ \widehat{\lambda}_j &= \max_{U_2} \{ \min_{\mathbf{f}} \{ \langle X_{\perp,\perp} \mathbf{f}, \mathbf{f} \rangle \mid \mathbf{f} \in U_2 \setminus \{0\} \} \mid \dim(U_2) = n - j \}, \end{aligned}$$

and for the same reason, by taking \max over bigger set may be bigger, i.e.

$$\widehat{\lambda}_j \le \lambda_{j+1}$$

3.5 Proofs of the main results

We proceed to the proof of Theorem 3.3.6; Theorem 3.3.4 follows from Theorem 3.3.6 immediately as a special case.

We first derive the characteristic equation of (3.13).

Theorem 3.5.1. If $\alpha \notin \text{Spec}(A)$ and $\beta \notin \text{Spec}(D)$, then the characteristic equation of (3.13) for $\beta(\alpha)$ is

$$\kappa^2 \langle (A - \alpha)^{-1} \mathbf{z}, \mathbf{z} \rangle \langle (D - \beta)^{-1} \mathbf{w}, \mathbf{w} \rangle = 1.$$
(3.25)

Proof. Re-writing the equation (3.16) as

$$(D - \beta)\mathbf{v} = C^*(A - \alpha)^{-1}C\mathbf{v}.$$

and then using the information that C is rank one, we obtain

$$(D-\beta)\mathbf{v} = C^*(A-\alpha)^{-1}C\mathbf{v},$$

$$= C^* (A - \alpha)^{-1} \kappa \langle \mathbf{v}, \mathbf{w} \rangle \mathbf{z}$$
$$= \kappa^2 \langle \mathbf{v}, \mathbf{w} \rangle \langle (A - \alpha)^{-1} \mathbf{z}, \mathbf{z} \rangle \mathbf{w},$$

which implies

$$\mathbf{v} = \kappa^2 \langle \mathbf{v}, \mathbf{w} \rangle \langle (A - \alpha)^{-1} \mathbf{z}, \mathbf{z} \rangle (D - \beta)^{-1} \mathbf{w}.$$

Now since the term $\kappa^2 \langle \mathbf{v}, \mathbf{w} \rangle \langle (A - \alpha)^{-1} \mathbf{z}, \mathbf{z} \rangle$ is a constant, we can fix it as

$$\kappa^2 \langle \mathbf{v}, \mathbf{w} \rangle \langle (A - \alpha)^{-1} \mathbf{z}, \mathbf{z} \rangle = 1,$$
 (3.26)

by setting

$$\mathbf{v} := (D - \beta)^{-1} \mathbf{w}. \tag{3.27}$$

Substituting (3.27) into (3.26), we arrive at (3.25).

Corollary 3.5.2. Suppose $\alpha \notin \text{Spec}(A)$ and $\beta \notin \text{Spec}(D)$. Then the equation (3.25) is equivalent to the following one:

$$\kappa^2 R_A(\alpha, \mathbf{z}) R_D(\beta, \mathbf{w}) = \kappa^2 \left(\sum_{i=1}^n \frac{|\langle \mathbf{z}, \boldsymbol{\varphi}_i \rangle|^2}{\alpha_i - \alpha} \right) \left(\sum_{j=1}^m \frac{|\langle \mathbf{w}, \boldsymbol{\psi}_j \rangle|^2}{\beta_j - \beta} \right) = 1.$$
(3.28)

Proof. The result yields by substituting $R_A(\alpha, \mathbf{z})$ and $R_D(\beta, \mathbf{w})$ into (3.25).

The next lemma shows that $(\Gamma_A \times \mathbb{C}) \cup (\mathbb{C} \times \Gamma_D) \subset \operatorname{Spec}_p(\mathbf{M})$, strengthening in fact the claim of Theorem 3.3.6.

Lemma 3.5.3. *If* $\alpha \in \Gamma_A$ *, then*

$$(\alpha, \beta) \in \operatorname{Spec}_p(\mathbf{M})$$
 for all $\beta \in \mathbb{C}$.

Similarly, if $\beta \in \Gamma_D$, then

$$(\alpha, \beta) \in \operatorname{Spec}_p(\mathbf{M})$$
 for all $\alpha \in \mathbb{C}$.

Proof. We prove the first of these statements, the second is similar. Let $\alpha \in \Gamma_A$, and let $\varphi \in \Phi_{A,\alpha}$ such that $\langle \varphi, \mathbf{z} \rangle = 0$. An immediate check shows that

$$egin{pmatrix} \mathbf{u} \ \mathbf{v} \end{pmatrix} = egin{pmatrix} oldsymbol{arphi} \ \mathbf{0} \end{pmatrix}$$

is a pair-eigenvector of M for a pair-eigenvalue (α, β) with an arbitrary $\beta \in \mathbb{C}$.

In Lemma 3.5.3 we show what happens when $\alpha \in \Gamma_A$ or $\beta \in \Gamma_D$; our next result shows which points (α, β) may lie in $\operatorname{Spec}_p(\mathbf{M})$ when α is an eigenvalue of A outside of Γ_A .

Lemma 3.5.4. Let $\alpha \in \operatorname{Spec}(A) \setminus \Gamma_A$, and $\beta \notin \Gamma_D$. Then $(\alpha, \beta) \in \operatorname{Spec}_p(\mathbf{M})$ if and only if $\beta = \widehat{\beta} \in \operatorname{Spec}(D_{\perp,\perp})$. Similarly, if $\beta \in \operatorname{Spec}(D) \setminus \Gamma_D$, and $\alpha \notin \Gamma_A$, then $(\alpha, \beta) \in \operatorname{Spec}_p(\mathbf{M})$ if and only if $\alpha = \widehat{\alpha} \in \operatorname{Spec}(A_{\perp,\perp})$.

Proof. Once more, we only prove the first statement. Let $\alpha \in \text{Spec}(A) \setminus \Gamma_A$. Let us re-write (3.14), (3.15) as

$$(A - \alpha)\mathbf{u} = -\kappa \langle \mathbf{v}, \mathbf{w} \rangle \mathbf{z}, \tag{3.29}$$

$$(D - \beta)\mathbf{v} = -\kappa \langle \mathbf{u}, \mathbf{z} \rangle \mathbf{w}.$$
 (3.30)

Multiplying (3.29) by $\varphi \in \Phi_{A,\alpha}$, we get

$$\langle (A-\alpha)\mathbf{u}, \boldsymbol{\varphi} \rangle = \langle \mathbf{u}, (A-\alpha)\boldsymbol{\varphi} \rangle = 0 = -\kappa \langle \mathbf{v}, \mathbf{w} \rangle \langle \mathbf{z}, \boldsymbol{\varphi} \rangle.$$

Since $\alpha \notin \Gamma_A$, we have $\langle \mathbf{z}, \boldsymbol{\varphi} \rangle \neq 0$, and so $\langle \mathbf{v}, \mathbf{w} \rangle = 0$ (and so $C\mathbf{v} = \mathbf{0}$), and by (3.29),

$$\mathbf{u} = a\boldsymbol{\varphi},\tag{3.31}$$

where the constant *a* may or may not be zero. Substituting now (3.31) into (3.30)

$$(D - \beta)\mathbf{v} = -\kappa a \langle \boldsymbol{\varphi}, \mathbf{z} \rangle \mathbf{w}, \qquad (3.32)$$

and since $\mathbf{v} \perp \mathbf{w}$, the equation (3.32) becomes

$$(D_{\perp,\perp} - \beta)\mathbf{v} + D_{\perp,\parallel}\mathbf{v} = -\kappa a \langle \boldsymbol{\varphi}, \mathbf{z} \rangle \mathbf{w}.$$
(3.33)

Now, applying the projections Q_w and P_w to the result, we obtain

$$D_{\perp,\perp}\mathbf{v} = \beta \mathbf{v},\tag{3.34}$$

$$D_{\perp,\parallel} \mathbf{v} = -\kappa a \langle \boldsymbol{\varphi}, \mathbf{z} \rangle \mathbf{w}. \tag{3.35}$$

If $\beta \notin \operatorname{Spec}(D_{\perp,\perp})$, then by (3.34), $\mathbf{v} = \mathbf{0}$, and thus a = 0, and so $\mathbf{u} = \mathbf{0}$, and $(\alpha, \beta) \notin \operatorname{Spec}_n(\mathbf{M})$, proving the "only if" part of the statement.

If $\beta = \hat{\beta} \in \text{Spec}(D_{\perp,\perp})$, and $\hat{\psi} \in \Phi_{D_{\perp,\perp},\hat{\beta}}$, we choose $\mathbf{v} = b\hat{\psi}$; we claim that we may choose constants a, b such that $a^2 + b^2 \neq 0$ to satisfy (3.35). After multiplying (3.35) by \mathbf{w} , it becomes

$$b\langle D_{\perp,\parallel}\widehat{\psi}, \mathbf{w}\rangle = -\kappa a \langle \varphi, \mathbf{z}\rangle.$$
(3.36)

The scalar product on the right-hand side is non-zero by our assumption $\alpha \notin \Gamma_A$. The scalar product on the left-hand side is non-zero since otherwise $D_{\perp,\parallel}\hat{\psi}$, which acts on the space parallel to w, is zero and by Lemma 3.5.3 $\hat{\beta} \in \operatorname{Spec} D$ with an eigenfunction $\hat{\psi}$, and therefore $\beta \in \Gamma_D$ by Lemma 3.4.2, again contradicting our assumptions. Thus we can always choose a, b with $a^2 + b^2 \neq 0$ in order to satisfy (3.36).

We can now prove our main result.

Proof of the full Chess Board Theorem. The eigenvalues inside $(\Gamma_A \times \mathbb{R}) \cup (\mathbb{R} \times \Gamma_D)$ have been already accounted for by Lemma 3.5.3, so we will be working outside this set.

Recall the characteristic equation (3.28). Since it needs to be satisfied, $R_A(\alpha, \mathbf{z})$ and $R_D(\beta, \mathbf{w})$ have to have the same sign for real pair-eigenvalues. It can be seen from (3.20) that $R_A(\alpha, \mathbf{z})$ is positive when $\alpha < \alpha_1$, and by Lemma 3.4.3, it only changes sign every time when α passes through x_p , $p = 1, \ldots, s$. Similarly, $R_D(\beta, \mathbf{w})$ is positive when $\beta < \beta_1$ and it only changes sign every time when β passes through y_q , $q = 1, \ldots, t$. Thus the only allowed regions for real α and β are when $(\alpha, \beta) \in \mathcal{R}_{p,q}$ with p + q even, proving, with account of Lemma 3.5.4, the statements (a) and (b).

Statement (c) follows immediately from (3.25) and Lemma 3.4.3.

To prove (d), we differentiate the characteristic equation (3.25) with respect to α , arriving at

$$\kappa^2 \langle (A-\alpha)^{-2} \mathbf{z}, \mathbf{z} \rangle \langle (D-\beta)^{-1} \mathbf{w}, \mathbf{w} \rangle + \kappa^2 \langle (A-\alpha)^{-1} \mathbf{z}, \mathbf{z} \rangle \langle (D-\beta)^{-2} \mathbf{w}, \mathbf{w} \rangle \frac{d\beta}{d\alpha} = 0,$$

so that

$$\frac{d\beta}{d\alpha} = -\frac{\langle (A-\alpha)^{-2}\mathbf{z}, \mathbf{z} \rangle \langle (D-\beta)^{-1}\mathbf{w}, \mathbf{w} \rangle}{\langle (A-\alpha)^{-1}\mathbf{z}, \mathbf{z} \rangle \langle (D-\beta)^{-2}\mathbf{w}, \mathbf{w} \rangle},$$

and re-arranging with account of (3.25), we can re-write β' as in (3.18).

To see that $\beta' < 0$, we observe from (3.25) that $R_A(\alpha, \mathbf{z}) \neq 0$ and $R_D(\beta, \mathbf{w}) \neq 0$. Also,

$$\langle (A-\alpha)^{-2}\mathbf{z}, \mathbf{z} \rangle = \langle (A-\alpha)^{-1}\mathbf{z}, (A-\alpha)^{-1}\mathbf{z} \rangle = \| (A-\alpha)^{-1}\mathbf{z} \|_{\mathcal{A}}$$

which is always positive by (3.19), and similarly $\langle (D - \beta)^{-2} \mathbf{w}, \mathbf{w} \rangle > 0$, and therefore $d\beta/d\alpha < 0$.

Proof of Theorem 3.3.8. Let $\alpha \notin \text{Spec}(A)$ and $\beta \notin \text{Spec}(D)$. For a self-adjoint operator A, we have by Cauchy-Schwartz, if $\|\mathbf{z}\| = 1$,

$$|\langle (A - \alpha)^{-1} \mathbf{z}, \mathbf{z} \rangle| \le ||(A - \alpha)^{-1} \mathbf{z}|| ||\mathbf{z}|| \le ||(A - \alpha)^{-1}|| = \frac{1}{\operatorname{dist}(\alpha, \operatorname{Spec}(A))},$$
 (3.37)

and similarly for D. By the characteristic equation (3.25), we have, as $\kappa \to 0$, that either

$$R_A(\alpha, \mathbf{z}) = \langle (A - \alpha)^{-1} \mathbf{z}, \mathbf{z} \rangle \to \infty$$

or

$$R_D(\beta, \mathbf{w}) = \langle (D - \beta)^{-1} \mathbf{w}, \mathbf{w} \rangle \to \infty,$$

which implies by (3.37) that either $dist(\alpha, Spec(A)) \rightarrow 0$ or $dist(\beta, Spec(D)) \rightarrow 0$. In other words, we have either α tends to an eigenvalue of A and β is arbitrary, or β tends to an eigenvalue of D and α is arbitrary. Then the first statement follows.

Similarly, if $\kappa \to \infty$, then again by (3.25) we have that either

$$R_A(\alpha, \mathbf{z}) = \langle (A - \alpha)^{-1} \mathbf{z}, \mathbf{z} \rangle \to 0$$

or

$$R_D(\beta, \mathbf{w}) = \langle (D - \beta)^{-1} \mathbf{w}, \mathbf{w} \rangle \to 0,$$

which implies by Lemma 3.4.1 that either $\alpha \to \widehat{\alpha} \in \text{Spec}(A_{\perp,\perp})$ or $\beta \to \widehat{\beta} \in \text{Spec}(D_{\perp,\perp})$. Then the second statement follows.

We note that our result holds under the hypothesis of the full Chess Board theorem since any point in Γ_A or Δ_A is always contained in $\operatorname{Spec}(A) \cap \operatorname{Spec}(A_{\perp,\perp})$, and similarly for D.

To finish the proof of the main theorem, it remains to show that the real pair-eigenvlaues can only lie in rectangles $\mathcal{R}_{p,q}$ with p + q even (except mesh lines). Actually we already mentioned in Remark 3.3.7 which rectangles are permitted and which are excluded. Nevertheless, we will show this case for the sake of completeness. As we know already by Lemma 3.5.4 that the eigenvalue curves can only cross through the mesh vertices, it is sufficient to indicate one rectangle with p + q even which contains at least one real pair-eigenvalue. This will then determine which rectangles are permitted because of the crossings. In the next result, we show that for any $\kappa \neq 0$, one can always find a pair-eigenvalue in the region $\mathcal{R}_{2n,2m}$ (or $\mathcal{R}_{s+1,t+1}$ when we have an eigenvalue with multiplicity greater than one).

Lemma 3.5.5. Let α_n and β_m be the largest eigenvalue of A and D, respectively, which are in the mesh. For any $\alpha > \alpha_n$, $\exists \beta = \beta(\alpha) > \beta_m$ such that $(\alpha, \beta) \in \text{Spec}_p(\mathbf{M})$.

Proof. Take $\alpha > \alpha_n$. Then the resolvent $(A - \alpha)^{-1}$ exists and is a negative operator. Moreover,

$$(\alpha, \beta) \in \operatorname{Spec}_{n}(\mathbf{M}) \Leftrightarrow \beta \in \operatorname{Spec}(T(\alpha)),$$

where $T(\alpha) = D - C^*(A - \alpha)^{-1}C$ is Hermitian. Thus, to prove the statement it remains to show that $T(\alpha)$ has an eigenvalue greater than β_m . Let \mathbf{v}_m be a normalised eigenvector of D corresponding to β_m . Then

$$\langle T(\alpha)\mathbf{v}_m, \mathbf{v}_m \rangle = \langle D\mathbf{v}_m, \mathbf{v}_m \rangle - \langle (A-\alpha)^{-1}\mathbf{w}_m, \mathbf{w}_m \rangle$$

where $\mathbf{w}_m = C \mathbf{v}_m$. If $\mathbf{w}_m \neq 0$, then by (3.20), we have

$$-\langle (A-\alpha)^{-1}\mathbf{w}_m, \mathbf{w}_m \rangle = \sum_{j=1}^n \frac{|\langle \mathbf{w}_m, \mathbf{u}_j \rangle|^2}{\alpha - \alpha_j} = \kappa^2 |\langle \mathbf{v}_m, \mathbf{w} \rangle|^2 \sum_{j=1}^n \frac{|\langle \mathbf{z}, \mathbf{u}_j \rangle|^2}{\alpha - \alpha_j},$$

which is always positive since $\alpha > \alpha_n$. Hence $\langle T(\alpha)\mathbf{v}_m, \mathbf{v}_m \rangle > \beta_m$. In that case, by variational principle $T(\alpha)$ has an eigenvalue $\beta(\alpha) > \beta_m$. Suppose $\mathbf{w}_m = 0$. Then

$$\mathbf{w}_m = C\mathbf{v}_m = \kappa \langle \mathbf{v}_m, \mathbf{w} \rangle \mathbf{z} = 0 \iff \mathbf{v}_m \perp \mathbf{w} \iff \beta_m \in \widetilde{\Gamma}_D$$

However, if $\beta_m \in \widetilde{\Gamma}_D$ then β_m will not be in the mesh. Due to our construction, we do not take into account of the set $\widetilde{\Gamma}$ in the mesh, and therefore $\mathbf{w}_m \neq 0$.

3.6 Examples

3.6.1 Example 1

As we mentioned earlier, the main motivation of this section comes from the particular non-self-adjoint problem which was considered in [19], with corresponding change of no-tations. Re-write the pencil problem as:

$$\begin{pmatrix} H_0^{(n)} - \alpha & \kappa C \\ \kappa C^* & H_0^{(m)} - \beta \end{pmatrix} \begin{pmatrix} \mathbf{u} \\ \mathbf{v} \end{pmatrix} = 0,$$
(3.38)

where $(C)_{n,1} = 1$ and all other entries of *C* are zeros. We therefore set $\mathbf{z} = (0, \dots, 0, 1)^T$ and $\mathbf{w} = (1, 0, \dots, 0)^T$. When $\kappa = 1$, the problem (3.38) is equivalent to the pencil \mathcal{A}_c problem with (3.17), nevertheless we plot the spectrum for different values of κ .

The eigenvalues of $H_0^{(n)}$ are given by (2.37) and the eigenvalues of $\left(H_0^{(n)}\right)_{\perp,\perp}$ are given by the same formula with *n* replaced by n-1. We shall return to the investigation of the non-real pair-eigenvalues of this problem in Section 3.7.

Note that $\Gamma_{H_0^{(n)}} = \emptyset$ and the general spectral picture in the (α, β) -plane including the rectangular mesh can be seen in Figures 3.6 and Figure 3.11. We also illustrate in Figure 3.7 the real pair-eigenvalues by taking a wide range of κ values and superimposing them. We see that the results of the simple Chess Board Theorem hold. We also illustrate the limit cases in Figure 3.5 where we see that real pair-eigenvalues lie in the border of rectangular regions.



Figure 3.5: Spec_p(**M**) when $A = H_0^{(5)}$ and $D = H_0^{(4)}$. Left: with $\kappa = 10^{-5}$. Right: with $\kappa = 10^5$.



Figure 3.6: Spec_p(M) when $A = D = H_0^{(4)}$ with $\kappa = 0.4$ (red curves), $\kappa = 1$ (blue curves) and $\kappa = 2$ (orange curves).

3.6.2 Example 2

This example illustrates the case when $\Gamma = \varnothing.$ Let

$$A_2 = \operatorname{diag}(-1, 2), \quad D_1 = \operatorname{diag}(1, 3, 4).$$
 (3.39)

Set

$$\mathbf{z} = \frac{1}{\sqrt{2}} \begin{pmatrix} 1\\1 \end{pmatrix}, \quad \mathbf{w} = \frac{1}{\sqrt{3}} \begin{pmatrix} 1\\1\\1 \end{pmatrix}.$$
 (3.40)



Figure 3.7: The superimposition of $\operatorname{Spec}_p(\mathbf{M})$ when $A = D = H_0^{(4)}$ for the values of κ from 0.001 to 10 with the step-size of 0.1.

Then $\Gamma_{A_2} = \Gamma_{D_1} = \emptyset$. Also

$$\operatorname{Spec}((A_2)_{\perp,\perp}) = \left\{\frac{1}{2}\right\}, \quad \operatorname{Spec}((D_1)_{\perp,\perp}) = \left\{\frac{1}{3}(8 \pm \sqrt{7})\right\}.$$

The spectral picture is illustrated in Figure 3.8(left), and we see that the simple Chess Board Theorem (Theorem 3.3.4) holds.



Figure 3.8: Left: $\operatorname{Spec}_p(\mathbf{M})$ with $A = A_2$, $D = D_1$, and $\kappa = 1$. Right: $\operatorname{Spec}_p(\mathbf{M})$ with $A = A_3$, $D = D_1$, and $\kappa = 3/4$.

Remark 3.6.1. In the rest of the examples, we set $D = D_1$ and w, as given in (3.39) and (3.40), thus $\Gamma_{D_1} = \emptyset$. We will consider different *A* and z in order to illustrate each special
case, therefore these changes will be seen in the vertical direction only.

3.6.3 Example 3

This example illustrates the case when $\Gamma \neq \emptyset$ and $\widetilde{\Gamma} = \emptyset$. Consider

$$A_3 = \operatorname{diag}(1,1) \quad \text{and} \quad \mathbf{z} = \begin{pmatrix} 1\\ 0 \end{pmatrix}.$$
 (3.41)

Then $\Gamma_{A_3} = \{1\}$ and $\widetilde{\Gamma}_{A_3} = \emptyset$. Also

$$\text{Spec}((A_3)_{\perp,\perp}) = \{1\}.$$

The spectral picture is illustrated in Figure 3.8(right). We see that $\operatorname{Spec}_p(\mathbf{M})$ has an additional straight line at $\alpha = 1$, and there is also a blow-up at $\alpha = 1$. This line is included in the mesh since \mathbf{z} is orthogonal to one eigenvector but $\mathbf{z} \not\perp \Phi_{A_{3,1}}$.

3.6.4 Example 4

This example illustrates the case when $\Gamma = \widetilde{\Gamma} \neq \varnothing$. Consider

$$A_4 = \operatorname{diag}(1,4)$$
 and $\mathbf{z} = \begin{pmatrix} 1\\ 0 \end{pmatrix}$.

Then $\Gamma_{A_4} = \widetilde{\Gamma}_{A_4} = \{4\}$. Also

$$\text{Spec}((A_4)_{\perp,\perp}) = \{4\}.$$

The spectral picture is shown in Figure 3.9(left). It can be seen that there is an additional straight line passing through $\alpha = 4$ which is not included in the mesh since A_4 has simple eigenvalues and $\mathbf{z} \perp \Phi_{A_4,4}$.

3.6.5 Example 5

This example illustrates the case when $\Gamma,\,\widetilde{\Gamma}\neq\varnothing$ and $\Gamma\neq\widetilde{\Gamma}.$ Consider

$$A_5 = \operatorname{diag}(-1, -1, 3, 3)$$
 and $\mathbf{z} = \begin{pmatrix} 0 \\ 0 \\ 0 \\ 1 \end{pmatrix}$.

Then $\Gamma_{A_5} = \{-1, 3\}$ and $\widetilde{\Gamma}_{A_5} = \{-1\}$. Also

$$\operatorname{Spec}((A_5)_{\perp,\perp}) = \{-1, 3\},\$$



Figure 3.9: Left: Spec_p(\mathbf{M}) with $A = A_4$, $D = D_1$, and $\kappa = 3/2$. Right: Spec_p(\mathbf{M}) with $A = A_5$, $D = D_1$, and $\kappa = 4/3$.

where the eigenvalue -1 has multiplicity two. The spectral picture is shown in Figure 3.9(right). As expected, there are two additional vertical straight lines: at $\alpha = -1$, where there is no blow-up and the line is not included in the mesh since $z \perp \Phi_{A_5,-1}$; and at $\alpha = 3$, where there is a blow-up and the line is included in the mesh since $z \not\perp \Phi_{A_5,3}$.

3.6.6 Example 6

Finally, this example illustrates the case when $\Delta \neq \emptyset$. Consider

$$A_6 = \text{diag}(1, 2, 2, 3)$$
 and $\mathbf{z} = \frac{1}{\sqrt{2}} \begin{pmatrix} 1\\0\\0\\1 \end{pmatrix}$. (3.42)

Then $\Gamma_{A_6} = \widetilde{\Gamma}_{A_6} = \{2\}$. Also

$$\operatorname{Spec}((A_6)_{\perp,\perp}) = \{2\},\$$

where the eigenvalue 2 has multiplicity three. The spectral picture is shown in Figure 3.10(right). Since $\mathbf{z} \perp \Phi_{A_6,2}$, there is no blow-up at $\alpha = 2$. Nevertheless, this line is included in the mesh as $\Delta_{A_6} = \{2\}$, i.e. $\dim(\Phi_{(A_6)_{\perp,\perp},2}) > \dim(\Phi_{A_6,2})$.

3.7 The non-real pair-eigenvalues of \mathbf{M}

We now turn our attention to the non-real pair eigenvalues of a two-parameter eigenvalue problem (3.13). Generally speaking, for every $\alpha \in \mathbb{C} \setminus \text{Spec}(A)$, we have either *m* corresponding $\beta \in \mathbb{C}$ which occurs under the restriction of the simple Chess Board Theorem,



Figure 3.10: Spec_p(\mathbf{M}) with $A = A_6$, $D = D_1$, and $\kappa = 5/3$.

or infinitely many $\beta \in \mathbb{R}$ which occurs whenever Γ , $\widetilde{\Gamma}$ or Δ is not equal to the empty set. We therefore limit our attention to the pair-eigenvalues subject to the additional restriction

$$\operatorname{Im}(\alpha + \beta) = 0, \tag{3.43}$$

which is equivalent to introducing the additional restriction $c \in \mathbb{R}$ in the non-self-adjoint problem, see (3.17). For this reason, we introduce the following notation:

$$\widetilde{\operatorname{Spec}}(\mathbf{M}) := \operatorname{Spec}_{n}(\mathbf{M}) \cap \{(\alpha, \beta) \in \mathbb{C}^{2} : \operatorname{Im}(\alpha) + \operatorname{Im}(\beta) = 0\}$$

In addition, for brevity, we will work under the restrictions of the simple Chess Board Theorem.

Lemma 3.7.1. The spectrum $\widetilde{\text{Spec}}(\mathbf{M})$ is invariant under the symmetry $(\alpha, \beta) \to (\overline{\alpha}, \overline{\beta})$.

Proof. The result is true for any $\mathbf{M}(\alpha, \beta) = \mathbf{L} - \alpha \mathbf{V}_1 - \beta \mathbf{V}_2$ with self-adjoint coefficients \mathbf{L} , \mathbf{V}_1 and \mathbf{V}_2 : If (α, β) is a pair-eigenvalue of $\mathbf{M}(\alpha, \beta)$, then $\mathbf{L} - \alpha \mathbf{V}_1 - \beta \mathbf{V}_2$ is not invertible, therefore $(\mathbf{L} - \alpha \mathbf{V}_1 - \beta \mathbf{V}_2)^* = \mathbf{L} - \overline{\alpha} \mathbf{V}_1 - \overline{\beta} \mathbf{V}_2$ is not invertible and $(\overline{\alpha}, \overline{\beta})$ is a pair-eigenvalue of $\mathbf{M}(\alpha, \beta)$.

A general spectral picture of the non-self-adjoint problem considered in [19] is illustrated in the (α, β) -plane in Figure 3.11. Red curves depict the real parts of non-real pair-eigenvalues $\operatorname{Re}\beta(\operatorname{Re}\alpha)$ such that (3.43) holds, which keeps all $(\alpha, \beta) \in \mathbb{R}^2$ in the picture (shown in blue) and also some non-real pair-eigenvalues. In general, as can be seen from the examples which were considered in Section 3.6 there is no additional symmetry. Nevertheless, in our special example, we have the following result. **Corollary 3.7.2.** Consider a two-parameter eigenvalue problem \mathbf{M} with $A = D = H_0^{(n)}$, $C = C^*$ with $(C)_{nn} = 1$ and all other entries of C are zeros. Then, in addition to the result of Lemma 3.7.1, the spectrum $\widetilde{\text{Spec}}(\mathbf{M})$ is invariant under the symmetry $(\alpha, \beta) \rightarrow (\beta, \alpha)$ and the symmetry $(\alpha, \beta) \rightarrow (-\beta, -\alpha)$.

Proof. We know by Lemma 3.1.3 that $\text{Spec}(\mathcal{A}_c)$ is invariant under the symmetries $\lambda \to -\lambda$ and $c \to -c$. We see by (3.17) that these symmetries correspond to $(\alpha, \beta) \to (\beta, \alpha)$ and $(\alpha, \beta) \to (-\beta, -\alpha)$, respectively.



Figure 3.11: $\widetilde{\text{Spec}}(\mathbf{M})$ with $A = D = H_0$ and C = B for n = 5.

The real and non-real eigenvalue curves $\lambda(c)$ may collide, with two possible types of collisions: those when two real eigenvalues collide and produce a complex conjugate pair, called Type-A, and those when a pair of complex conjugate eigenvalues collide and become real, called Type-B, see Figure 3.12 for equivalents in (α, β) -plane.

We gave an equivalent statement of Conjecture 1.3.3 in terms of a two-parameter eigenvalue problem in Theorem 1.3.4(i). We now turn to the proof of the result.

Proof of Theorem 1.3.4(i). Using the relation (3.17), we see that $\lambda \in \mathbb{C} \setminus \mathbb{R}$ if and only if $\alpha, \beta \in \mathbb{C} \setminus \mathbb{R}$. Consider the two-parameter eigenvalue problem $\mathbf{M}(\alpha, \beta)$ with $A = D = H_0^{(n)}$, $C = C^*$ with $(C)_{nn} = 1$ and all other entries of *C* are zeros. We know that any pencil problem of the form (1.10) and any two-parameter eigenvalue problem of the form (1.9)



Figure 3.12: Left: the collisions in the $(\text{Re}(\lambda), \text{Im}(\lambda))$ -plane. Right: the collisions in the $(\text{Re}(\alpha), \text{Re}(\beta))$ -plane.

are equivalent by (1.14). Therefore this two-parameter problem $\mathbf{M}(\alpha, \beta)$ and (3.5) are likewise equivalent with $\alpha = \sigma$ and $\beta = -\tau$. If c > 0, then

$$|\alpha| = |\lambda - c| < 2, \qquad |\beta| = |\lambda + c| < 2,$$

and if c < 0, then the inequalities will still hold by Corollary 3.7.2.

Unfortunately, we could not find much information about the non-real pair-eigenvalues which satisfy such bounds. Nevertheless, we found the collision locations in terms of derivative of $\beta(\alpha)$ with respect to α , and this holds not only for our pencil problem but also for any two-parameter eigenvalue problem with rank one *C*. Our first main result in this section is the following.

Lemma 3.7.3. The collisions in $\operatorname{Spec}(M)$ occur at the points where

$$\frac{d\beta(\alpha)}{d\alpha} = -1$$

Proof. Recall a two-parameter eigenvalue problem with the change of notation (3.17):

$$\begin{pmatrix} A - \lambda + c & C \\ C^* & D + \lambda + c \end{pmatrix} \begin{pmatrix} \mathbf{u} \\ \mathbf{v} \end{pmatrix} = \mathbf{0},$$

which can be written equivalently as

$$\mathbf{M}_{\lambda}\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix} = \begin{pmatrix} A-\lambda & C\\ C^{*} & D+\lambda \end{pmatrix}\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix} = -c\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix}.$$

Considering $c(\lambda)$ and taking the derivative in each side of the spectral problem with respect to λ , we obtain

$$\begin{split} \mathbf{M}_{\lambda}^{'}\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix} + \mathbf{M}_{\lambda}\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix}^{'} &= -c^{'}\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix} - c\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix}^{'} \Leftrightarrow (\mathbf{M}_{\lambda} + c)\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix}^{'} &= \left(-c^{'} - \mathbf{M}_{\lambda}^{'}\right)\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix}\\ \Leftrightarrow (\mathbf{M}_{\lambda} + c)\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix}^{'} &= \left(-c^{'} - \begin{pmatrix}-I\\I\end{pmatrix}\right)\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix}, \end{split}$$

which is solvable if and only if the left hand side of the last equation is orthogonal to $\begin{pmatrix} u \\ v \end{pmatrix}$. Therefore multiplying by $\begin{pmatrix} u \\ v \end{pmatrix}$ and bearing in mind $\|u\| = \|v\|$ by Corollary 3.1.8, we obtain

$$c'(\|\mathbf{u}\|^2 + \|\mathbf{v}\|^2) = \left\langle \begin{pmatrix} I \\ -I \end{pmatrix} \begin{pmatrix} \mathbf{u} \\ \mathbf{v} \end{pmatrix}, \begin{pmatrix} \mathbf{u} \\ \mathbf{v} \end{pmatrix} \right\rangle$$
$$= \|\mathbf{u}\|^2 - \|\mathbf{v}\|^2$$
$$= 0.$$

We show that the critical points occur when $\frac{dc}{d\lambda} = 0$; then using (3.17) gives

$$\frac{dc}{d\lambda} = \frac{\frac{dc}{d\alpha}}{\frac{d\lambda}{d\alpha}} = \frac{-\frac{1}{2} - \frac{1}{2}\beta'}{\frac{1}{2} - \frac{1}{2}\beta'} = 0 \quad \Leftrightarrow \quad \beta' = -1.$$

3.7.1 Limiting cases

If κ tends to zero, then the problem (3.13) becomes disjoint and the spectral picture can be understood easily. On the other hand, a very interesting phenomenon occurs when κ gets large; the spectral picture becomes more complicated. For instance, for some values of κ , non-real pair-eigenvalue collisions may occur, see Figure 3.13.



Figure 3.13: $\widetilde{\text{Spec}}(\mathbf{M})$ when $A = D = H_0$ with n = 5 for $\kappa = 1.37$. Left: in the $(\operatorname{Re}(\alpha), \operatorname{Re}(\beta))$ -plane. Right: in the $(\operatorname{Re}(\alpha), \operatorname{Im}(\alpha))$ -plane.

We know that if $\alpha \in \mathbb{R} \setminus (\operatorname{Spec}(A) \cup \operatorname{Spec}(A_{\perp,\perp}))$, then there are *m* corresponding real β 's such that

$$(\alpha, \beta_{\kappa}^{(1)}(\alpha)), \ldots, (\alpha, \beta_{\kappa}^{(m)}(\alpha)) \in \widetilde{\operatorname{Spec}}(\mathbf{M}).$$

Theorem 3.3.8 implies that as $\kappa \to \infty$, there are (m-1) corresponding β 's such that

$$\beta_{\kappa}^{(1)}(\alpha), \ldots, \beta_{\kappa}^{(m-1)}(\alpha) \to \operatorname{Spec}(D_{\perp,\perp}),$$

and m^{th} one goes to $\pm \infty$ where \pm depends on which block α lies in. For instance, as $\alpha \rightarrow \alpha_n + 0$, then $R_A(\alpha) \rightarrow -\infty$, and by (3.28) we obtain $R_D(\beta) \rightarrow -0$, and therefore $\beta(\alpha) \rightarrow +\infty$.

In addition to the real pair-eigenvalues, there are some non-real pair-eigenvalues and we cannot do the asymptotics for all of them. Nevertheless, if we restrict ourselves to $\widetilde{\text{Spec}}(\mathbf{M})$, there is one more family for which we will show the asymptotics as $\kappa \to \infty$. In the rest of this section, we shall analyse in more detail this family of pair-eigenvalues as κ tends to infinity.



Figure 3.14: $\widetilde{\text{Spec}}(\mathbf{M})$ when $A = D = H_0$ with n = 5 for $\kappa = 5$. Left: in the $(\text{Re}(\alpha), \text{Re}(\beta))$ -plane. Right: in the $(\text{Re}(\alpha), \text{Im}(\alpha))$ -plane.

It can be seen in Figure 3.14 that for sufficiently large κ , α 's lie on a circle in the complex plane. Is it actually true that α 's and/or β 's lie on a circle? Can we figure out what the circle is? Is it always the case or only for this particular example? In the next theorem, which is one of the main results of this section, we will show that it is indeed the case; namely, if we take two circles of sufficiently large radius κ , centred at $A_{\parallel,\parallel}$ in the α -plane and $D_{\parallel,\parallel}$ in the β -plane, then each point in the first circle will be approximately the

first component of the large eigenvalue-pair, and each point in the second circle will be approximately the second component of the same eigenvalue-pair. Note that $A_{\parallel,\parallel}, D_{\parallel,\parallel} \in \mathbb{R}$ and can be computed as

$$A_{\parallel,\parallel}\mathbf{z} = \langle A\mathbf{z}, \mathbf{z} \rangle \mathbf{z}, \quad D_{\parallel,\parallel}\mathbf{w} = \langle D\mathbf{w}, \mathbf{w} \rangle \mathbf{w},$$

which are some constants proportional to z and w, respectively.

Theorem 3.7.4. As $\nu = \frac{1}{\kappa} \to 0$, there exist a family of pair-eigenvalues $(\alpha, \beta) \in \widetilde{\text{Spec}}(\mathbf{M})$ such that

$$|\alpha - A_{\parallel,\parallel}|^2 = \kappa^2 + O(\nu^2),$$
 (3.44)

$$\left|\beta - D_{\parallel,\parallel}\right|^2 = \kappa^2 + O\left(\nu^2\right),$$
 (3.45)

or more precisely, for any $t \in \mathbb{R}$, $(\alpha_t, \beta_t) \in \widetilde{\operatorname{Spec}}(\mathbf{M})$, where

$$\alpha_t = e^{it}\nu^{-1} + a_1 + O(\nu), \quad \beta_t = e^{-it}\nu^{-1} + b_1 + O(\nu), \tag{3.46}$$

and a_1 and b_1 , which are in general depending on t, satisfy

$$\operatorname{Re}\left((a_1 - A_{\parallel,\parallel})e^{-it}\right) = 0, \quad \operatorname{Re}\left((b_1 - D_{\parallel,\parallel})e^{it}\right) = 0.$$

The next result will be useful in the proof of Theorem 3.7.4.

Lemma 3.7.5. Let $c \in \mathbb{C} \setminus \{0\}$ arbitrary constant. Then the system

$$\begin{pmatrix} c & 1\\ 1 & 1/c \end{pmatrix} \begin{pmatrix} p\\ q \end{pmatrix} = \begin{pmatrix} s\\ t \end{pmatrix}$$

has a non-trivial solution if and only if

$$ct - s = 0.$$

Proof. Re-writing as a system of two equations

$$cp + q = s,$$

 $p + (1/c)q = t,$

and leaving *p* alone in the second equation and substituting into the first equation gives

$$c(t - (1/c)q) + q = s \iff ct - s = 0.$$

Proof of Theorem 3.7.4. First, we prove the expansions given in (3.46). Then will deduce the formulas (3.44) and (3.45). Take

$$\alpha = \kappa a, \quad \beta = \kappa b, \quad \nu = \frac{1}{\kappa},$$

where $a, b \in \mathbb{C}$ and we let $\nu \to 0$. Then the equation (3.14) and (3.15) can be re-written as

$$\begin{aligned} A\mathbf{u} + \kappa \langle \mathbf{v}, \mathbf{w} \rangle \mathbf{z} &= \alpha \mathbf{u} \\ D\mathbf{v} + \kappa \langle \mathbf{u}, \mathbf{z} \rangle \mathbf{w} &= \beta \mathbf{v} \end{aligned} \right\} \iff \begin{cases} \nu A\mathbf{u} + \langle \mathbf{v}, \mathbf{w} \rangle \mathbf{z} &= a\mathbf{u} \\ \nu D\mathbf{v} + \langle \mathbf{u}, \mathbf{z} \rangle \mathbf{w} &= b\mathbf{v} \end{cases} \end{aligned}$$

and we are seeking a solution of the form

$$\begin{pmatrix} \mathbf{u} \\ \mathbf{v} \end{pmatrix} = \sum_{k=0}^{\infty} \begin{pmatrix} \mathbf{u}_k \\ \mathbf{v}_k \end{pmatrix} \nu^k, \qquad \begin{pmatrix} a \\ b \end{pmatrix} = \sum_{k=0}^{\infty} \begin{pmatrix} a_k \\ b_k \end{pmatrix} \nu^k.$$

At order $O(\nu^0)$, we have

$$\begin{cases} \langle \mathbf{v}_0, \mathbf{w} \rangle \mathbf{z} = a_0 \mathbf{u}_0 \\ \langle \mathbf{u}_0, \mathbf{z} \rangle \mathbf{w} = b_0 \mathbf{v}_0 \end{cases}$$

$$(3.47)$$

which implies that

$$\mathbf{u}_0 = c_0 \mathbf{z},$$
$$\mathbf{v}_0 = d_0 \mathbf{w},$$

where c_0 , d_0 are constants. Substituting these into (3.47) gives us

$$\left.\begin{array}{c} d_0 = a_0 c_0 \\ c_0 = b_0 d_0 \end{array}\right\}$$

and therefore

 $a_0b_0 = 1.$ (3.48)

Since we can always re-scale the eigenfunctions, we set $c_0 = 1$ and write the first terms as

$$\begin{pmatrix} \mathbf{u}_0 \\ \mathbf{v}_0 \end{pmatrix} = \begin{pmatrix} \mathbf{z} \\ a_0 \mathbf{w} \end{pmatrix}.$$

Using the additional restriction $Im(a_0 + b_0) = 0$, we write

$$a_0 = a_{0,r} + ia_{0,i}$$

 $b_0 = b_{0,r} - ia_{0,i}$

and substituting these into (3.48) yields

$$\left. \begin{array}{c} a_{0,r} = b_{0,r} \\ a_{0,r}b_{0,r} + a_{0,i}^2 = 1 \end{array} \right\}$$

which implies

$$a_0 = e^{it}, \quad b_0 = e^{-it}$$
 (3.49)

for arbitrary $t \in \mathbb{R}$. At order $O(\nu^1)$, we have

$$\begin{aligned} A\mathbf{u}_0 + \langle \mathbf{v}_1, \mathbf{w} \rangle \mathbf{z} &= a_0 \mathbf{u}_1 + a_1 \mathbf{u}_0 \\ D\mathbf{v}_0 + \langle \mathbf{u}_1, \mathbf{z} \rangle \mathbf{w} &= b_0 \mathbf{v}_1 + b_1 \mathbf{v}_0 \end{aligned} \} \iff \begin{cases} A\mathbf{z} + \langle \mathbf{v}_1, \mathbf{w} \rangle \mathbf{z} &= e^{it} \mathbf{u}_1 + a_1 \mathbf{z} \\ e^{it} D\mathbf{w} + \langle \mathbf{u}_1, \mathbf{z} \rangle \mathbf{w} &= e^{-it} \mathbf{v}_1 + e^{it} b_1 \mathbf{w} \end{aligned}$$
(3.50)

Now separating vectors orthogonal and parallel to z and w in (3.50), we get

$$A_{\parallel,\perp}\mathbf{z} = e^{it}Q_{\mathbf{z}}\mathbf{u}_1 \tag{3.51}$$

$$(A_{\parallel,\parallel} - a_1)\mathbf{z} - e^{it}P_{\mathbf{z}}\mathbf{u}_1 + \langle \mathbf{v}_1, \mathbf{w} \rangle \mathbf{z} = 0$$
(3.52)

$$e^{it}D_{\parallel,\perp}\mathbf{w} = e^{-it}Q_{\mathbf{w}}\mathbf{v}_1 \tag{3.53}$$

$$(e^{it}D_{\parallel,\parallel} - e^{it}b_1)\mathbf{w} - e^{-it}P_{\mathbf{w}}\mathbf{v}_1 + \langle \mathbf{u}_1, \mathbf{z} \rangle \mathbf{w} = 0$$
(3.54)

Now, write (3.52) and (3.54) as a system

$$\begin{pmatrix} -e^{it} & 1\\ 1 & -e^{-it} \end{pmatrix} \begin{pmatrix} \langle \mathbf{u}_1, \mathbf{z} \rangle\\ \langle \mathbf{v}_1, \mathbf{w} \rangle \end{pmatrix} = \begin{pmatrix} a_1 - A_{\parallel,\parallel}\\ e^{it}(b_1 - D_{\parallel,\parallel}) \end{pmatrix}.$$
(3.55)

By Lemma 3.7.5, the system (3.55) has a non-trivial solution if and only if

$$(a_1 - A_{\parallel,\parallel})e^{-it} + (b_1 - D_{\parallel,\parallel})e^{it} = 0.$$

Separating real and imaginary parts as $a_1 = a_{1,r} + ia_{1,i}$ and $b_1 = b_{1,r} - ia_{1,i}$, we obtain

$$\Leftrightarrow (a_{1,r} - A_{\parallel,\parallel} + ia_{1,i})(\cos t - i\sin t) + (b_{1,r} - D_{\parallel,\parallel} - ia_{1,i})(\cos t + i\sin t) = 0 \Leftrightarrow \begin{cases} (a_{1,r} - A_{\parallel,\parallel} + b_{1,r} - D_{\parallel,\parallel})\cos t + (a_{1,i} + a_{1,i})\sin t = 0 \\ (a_{1,i} - a_{1,i})\cos t + (-a_{1,r} + A_{\parallel,\parallel} + b_{1,r} - D_{\parallel,\parallel})\sin t = 0 \end{cases} \Leftrightarrow \begin{cases} (a_{1,r} - A_{\parallel,\parallel} + b_{1,r} - D_{\parallel,\parallel})\cos t + 2a_{1,i}\sin t = 0 \\ (a_{1,r} - A_{\parallel,\parallel} - b_{1,r} + D_{\parallel,\parallel})\sin t = 0 \end{cases}$$
(3.56)

From the second equation of (3.56), we obtain

$$a_{1,r} - A_{\parallel,\parallel} = b_{1,r} - D_{\parallel,\parallel}, \tag{3.57}$$

and substituting this into the first equation of (3.56) we obtain

$$2(a_{1,r} - A_{\parallel,\parallel})\cos t + 2a_{1,i}\sin t = 0 \quad \Leftrightarrow \quad \operatorname{Re}[(a_1 - A_{\parallel,\parallel})e^{-it}] = 0,$$
(3.58)

and similarly

$$2(b_{1,r} - D_{\parallel,\parallel})\cos t - 2a_{1,i}\sin t = 0 \quad \Leftrightarrow \quad \operatorname{Re}[(b_1 - D_{\parallel,\parallel})e^{it}] = 0.$$
(3.59)

Now, substituting a_0 and a_1 into α , we have

$$\alpha = \nu^{-1}a = a_0\nu^{-1} + a_1 + O(\nu^1)$$
$$= e^{it}\nu^{-1} + a_1 + O(\nu),$$

and similarly, substituting b_0 , b_1 into β , we have

$$\beta = \nu^{-1}b = b_0\nu^{-1} + b_1 + O(\nu^1)$$
$$= e^{-it}\nu^{-1} + b_1 + O(\nu),$$

where a_1 and b_1 satisfies (3.58) and (3.59), respectively.

We now prove (3.44), the second statement (3.45) is similar. Using the expansion for α in (3.46) we get

$$\begin{aligned} |\alpha - A_{\parallel,\parallel}|^2 &= |\kappa a - A_{\parallel,\parallel}|^2 \\ &= |\nu^{-1}(a_0 + \nu a_1) - A_{\parallel,\parallel}|^2 + O(\nu^2) \\ &= |\nu^{-1}e^{it} + a_1 - A_{\parallel,\parallel}|^2 + O(\nu^2) \\ &= |\nu^{-1}|^2 |e^{it} + \nu(a_1 - A_{\parallel,\parallel})|^2 + O(\nu^2) \\ &= \nu^{-2} \left[1 + 2\nu \operatorname{Re}[(a_1 - A_{\parallel,\parallel})e^{-it}]\right] + O(\nu^2) \\ &= \nu^{-2} + O(\nu^2) \\ &= \kappa^2 + O\left(\frac{1}{\kappa^2}\right), \end{aligned}$$

where we used (3.58) and $|x + y|^2 = |x|^2 + |y|^2 + 2\text{Re}(\overline{x}y)$.

We see from Figure 3.14 that when A = D, the large non-real pair-eigenvalue seems to occur on the line $\operatorname{Re}(\alpha) \cong \operatorname{Re}(\beta)$ which is equivalent to $\operatorname{Re}(\lambda) = 0$ in the non-self-adjoint problem. We now formally state this result as a corollary.

Corollary 3.7.6. Let A = D. Then there exists a pair eigenvalue $(\alpha, \beta) \in \widetilde{\text{Spec}}(\mathbf{M})$ such that

$$\operatorname{Re}(\alpha) = \operatorname{Re}(\beta) + O(\kappa^{-2}),$$

i.e.

$$\alpha = \overline{\beta} + O(\kappa^{-2}).$$

Proof. This follows immediately from (3.49) and (3.57).

3.7.2 Examples

In this section, we apply the limit results to the examples given in Section 3.6. Most examples produce similar plots so we only mention three of them here. In each graph, we also plot a circle of radius κ , centred at $A_{\parallel,\parallel}$ in the α -plane and $D_{\parallel,\parallel}$ in the β -plane, and both indicated by a black circle.

First, consider $A = D = H_0$, and take n = 7. Then $(H_0)_{\parallel,\parallel} = 0$ and the behaviour of the non-real eigenvalues in the α -plane can be seen in Figure 3.15. Since we take A = D, we plot the figures only in the α -plane as figures in the β -plane are identical. The figure illustrates that one family of non-real pair-eigenvalues approaches a circle as κ increases, whereas the rest of the non-real eigenvalues get smaller and eventually become real.



Figure 3.15: $\widetilde{\text{Spec}}(\mathbf{M}) \setminus \mathbb{R}^2$ (red curves) in the α -plane with $A = D = H_0$ for n = 7. Left: $\kappa = 2$ and $\mathcal{D}(0, 2)$ (black dashed circle). Right: $\kappa = 8$ and $\mathcal{D}(0, 8)$ (black dashed circle).

Now, we consider some two-parameter eigenvalue examples given in Section 3.6. As a second illustration, consider Example 3, i.e. take $A = A_3$ with $\mathbf{z} = (1,0)^T$ as in (3.41), and $D = D_1$ with $\mathbf{w} = 1/\sqrt{3}(1,1,1)^T$ as in (3.39). Then

$$(A_3)_{\parallel,\parallel} = 1, \quad (D_1)_{\parallel,\parallel} = 8/3.$$

The limiting non-real eigenvalue behaviour for large κ can be seen as expected in Figure 3.16.

Third, we consider Example 6, i.e. take $A = A_6$ with $\mathbf{z} = 1/\sqrt{2}(1,0,0,1)^T$ as in (3.42), and $D = D_1$ with $\mathbf{w} = 1/\sqrt{3}(1,1,1)^T$ as in (3.39). Then $(A_6)_{\parallel,\parallel} = 2$. The limiting non-real eigenvalue behaviour can be seen in Figure 3.17.



Figure 3.16: $\widetilde{\text{Spec}}(\mathbf{M}) \setminus \mathbb{R}^2$ (red curves) with $A = A_3$ and $D = D_1$ for $\kappa = 10$. Left: $\mathcal{D}(1, 10)$ (black dashed circle) in the α plane. Right: $\mathcal{D}(8/3, 10)$ (black dashed circle) in the β -plane.



Figure 3.17: $\widetilde{\text{Spec}}(\mathbf{M}) \setminus \mathbb{R}^2$ (red curves) with $A = A_6$ and $D = D_1$ for $\kappa = 10$. Left: $\mathcal{D}(2, 10)$ (black dashed circle) in the α plane. Right: $\mathcal{D}(8/3, 10)$ (black dashed circle) in the β -plane.

Chapter 4

Bounds for the eigenvalues of A_c

In this chapter, we turn our attention to the linear pencil problem (3.3) and try to answer some open questions about the non-real eigenvalues of A_c . These questions were first raised by Davies and Levitin [19]. We will try to treat the problem (3.3) differently in order to gain meaningful information. We start by re-writing the characteristic equation of A_c in numerous ways in Section 4.1. In addition, Conjecture 1.3.3 can be reformulated in various ways and we mention three of them in Section 4.2. Section 4.3 contains a full proof of the conjecture when n = 2 and n = 3. We will see below that the conjecture is non-trivial in these cases. Section 4.4 uses asymptotic methods to locate the non-real eigenvalues of A_c when c goes to zero.

Finally, Section 4.5 applies the useful tools introduced in Chapter 2 to our pencil problem. We study finite ratios of Chebyshev polynomials of the second kind $U_n(\zeta)$ in Section 4.2.1. We shall see that known ratio asymptotic estimates mentioned in Section 4.5.2 are not sufficient when applied to the ratio $U_{n+1}(\zeta)/U_n(\zeta)$. Therefore we will look at the finite ratios of $U_n(\zeta)$ in more detail to obtain better estimates. The results we will obtain are still inadequate to answer the open questions, nonetheless we believe that these estimates are useful. We conclude the chapter in Section 4.5.3 with a detailed discussion of the well-known Gershgorin-type results when applied to our particular problem.

4.1 Different representation of the operator pencil A_c

Our goal in this section is to treat the linear pencil problem (3.3) in various ways by writing several variations of the characteristic equation.

4.1.1 Explicit expressions for eigenfunctions of A_c

The problem (3.5) can be re-written as

$$(H_0 - \sigma I)\mathbf{u} = -B\mathbf{v},\tag{4.1}$$

$$(H_0 + \tau I)\mathbf{v} = -B\mathbf{u}.\tag{4.2}$$

Recall the relation $F_j(\zeta) = U_j(\zeta/2)$, where U_j is the Chebyshev polynomial of the second kind, and that the function F_j satisfies the recurrence relation (2.35).

Lemma 4.1.1. Let $\zeta \notin \text{Spec}(H_0)$, and let w solve

$$\left(H_0^{(n)} - \zeta I\right) \mathbf{w} = \begin{pmatrix} 0\\ \vdots\\ 0\\ 1 \end{pmatrix}.$$
(4.3)

Then

$$\mathbf{w} = \mathbf{w}(\zeta) = -\frac{1}{F_n(\zeta)} \begin{pmatrix} F_0(\zeta) \\ F_1(\zeta) \\ \vdots \\ F_{n-2}(\zeta) \\ F_{n-1}(\zeta) \end{pmatrix} = -\frac{F_{n-1}(\zeta)}{F_n(\zeta)} \begin{pmatrix} F_0(\zeta)/F_{n-1}(\zeta) \\ F_1(\zeta)/F_{n-1}(\zeta) \\ \vdots \\ F_{n-2}(\zeta)/F_{n-1}(\zeta) \\ 1 \end{pmatrix}$$

with

$$\|\mathbf{w}(\zeta)\|^2 = \frac{\mathcal{F}_n(\zeta)}{|F_n(\zeta)|^2},$$

where

$$\mathcal{F}_n(\zeta) := \sum_{j=0}^{n-1} |F_j(\zeta)|^2.$$
(4.4)

Proof. The equation (4.3) is written in components as

$$w_2 - \zeta w_1 = 0$$

 $w_{j-1} - \zeta w_j + w_{j+1} = 0,$ for $j = 2, ..., (n-1),$
 $w_{n-1} - \zeta w_n = 1.$

Substituting $w_j = -\frac{F_{j-1}(\zeta)}{F_n(\zeta)}$ we see that all *n* equations hold by (2.35).

Corollary 4.1.2. Let \mathbf{u} , \mathbf{v} , σ , τ solve (4.1)-(4.2), and let σ , $\tau \notin {\mu_n, \ldots, \mu_1}$. Then

$$\mathbf{u} = -v_n \mathbf{w}(\sigma) = \frac{v_n}{F_n(\sigma)} \begin{pmatrix} F_0(\sigma) \\ F_1(\sigma) \\ \vdots \\ F_{n-1}(\sigma) \end{pmatrix},$$
$$\mathbf{v} = -u_n \mathbf{w}(-\tau) = \frac{u_n}{F_n(\tau)} \begin{pmatrix} (-1)^n F_0(\tau) \\ (-1)^{n+1} F_1(\tau) \\ \vdots \\ -F_{n-1}(\tau) \end{pmatrix}$$

Proof. Immediate from (4.1)-(4.2) and Lemma 4.1.1.

Lemma 4.1.3. Let \mathbf{u} , \mathbf{v} , σ , τ solve (4.1)-(4.2), and in addition let σ , $\tau \notin \{\mu_n, \ldots, \mu_1\}$ and $\|\mathbf{u}\| = 1$. Then $\lambda = (\sigma + \tau)/2$ is a simple eigenvalue of \mathcal{A}_c , and

$$F_{n-1}(\sigma)F_{n-1}(\tau) + F_n(\sigma)F_n(\tau) = 0,$$
(4.5)

.

where

$$\sigma = \lambda - c, \qquad \tau = \lambda + c,$$

Moreover

$$|v_n|^2 = \frac{|F_n(\sigma)|^2}{\mathcal{F}_n(\sigma)} = \frac{|F_{n-1}(\tau)|^2}{\mathcal{F}_n(\tau)},$$
(4.6)

$$u_n|^2 = \frac{|F_n(\tau)|^2}{\mathcal{F}_n(\tau)} = \frac{|F_{n-1}(\sigma)|^2}{\mathcal{F}_n(\sigma)}.$$
(4.7)

Additionally,

$$\mathbf{u} = \frac{e^{is}}{\sqrt{\mathcal{F}_n(\sigma)}} \begin{pmatrix} F_0(\sigma) \\ F_1(\sigma) \\ \vdots \\ F_{n-1}(\sigma) \end{pmatrix}, \qquad \mathbf{v} = \frac{e^{it}}{\sqrt{\mathcal{F}_n(\tau)}} \begin{pmatrix} (-1)^n F_0(\tau) \\ (-1)^{n+1} F_1(\tau) \\ \vdots \\ -F_{n-1}(\tau) \end{pmatrix}$$

for some $s, t \in \mathbb{R}$.

Proof. By Corollary 4.1.2,

$$u_n = v_n \frac{F_{n-1}(\sigma)}{F_n(\sigma)}, \qquad v_n = -u_n \frac{F_{n-1}(\tau)}{F_n(\tau)},$$

which implies (4.5). The eigenvalue is geometrically simple since the eigenspace is onedimensional (with u_n or v_n being a simple parameter). By Lemma 2.1.3, we know that

 $\|\mathbf{u}\| = \|\mathbf{v}\|$. The first equalities in each of (4.6) and (4.7) are then obtained from the normalising conditions

$$\|\mathbf{v}\|^{2} = |u_{n}|^{2} \|\mathbf{w}(-\tau)\|^{2} = |u_{n}|^{2} \frac{\mathcal{F}_{n}(\tau)}{|F_{n}(\tau)|^{2}} = 1,$$
$$\|\mathbf{u}\|^{2} = |v_{n}|^{2} \|\mathbf{w}(\sigma)\|^{2} = |v_{n}|^{2} \frac{\mathcal{F}_{n}(\sigma)}{|F_{n}(\sigma)|^{2}} = 1.$$

The second ones are obtained from writing normalising conditions as

$$\|\mathbf{v}\|^{2} = |v_{n}|^{2} \sum_{j=1}^{n} \frac{|v_{j}|^{2}}{|v_{n}|^{2}} = |v_{n}|^{2} \frac{\mathcal{F}_{n}(\tau)}{|F_{n-1}(\tau)|^{2}} = 1,$$

and similarly for \mathbf{u} . [Remark: the second inequalities in (4.6) and (4.7) can be also deduced from the first ones and (4.5).] The expressions for \mathbf{u} and \mathbf{v} result from a substitution.

Remark 4.1.4. It would be nice to obtain additional restrictions on σ and τ from the fact that the quantities in (4.6) and (4.7) should be less than one. Unfortunately this is not the case - the numerics of the contour plot of the RH sides in (4.6) and (4.7) show that the curves on which $\frac{|F_n(\tau)|^2}{\mathcal{F}_n(\tau)} = 1$ protrude (by a small amount) into the domain $|\tau| > 2$ (or $\operatorname{Re}(\tau) > 2$) for $n \ge 4$. These curves are illustrated in Figure 4.1 as n increases. Note that this is because the construction above does not use the crucial condition $\operatorname{Im}(\sigma) = \operatorname{Im}(\tau)$.



Figure 4.1: Contour plots of $\frac{|F_n(\tau)|^2}{\mathcal{F}_n(\tau)} = 1$ (blue curves) in the $(\operatorname{Re}(\tau), \operatorname{Im}(\tau))$ -plane and $\operatorname{Re}(\tau) = \pm 2$ (red dashed lines) for n = 3, n = 4 and n = 5 from left to right.

4.1.2 Mapping z + 1/z

Let $\lambda \in \text{Spec}(\mathcal{A}_c)$. We use the following substitution

$$\lambda - c = \sigma := z + \frac{1}{z}, \qquad \lambda + c = \tau := w + \frac{1}{w}, \tag{4.8}$$

where z and w are some complex numbers. Note that each λ corresponds to two values of z (which are inverses of each other) and two values of w (which are also inverses of each other). These values are the solutions of the quadratic equations

$$z^{2} - (\lambda - c)z + 1 = 0, \qquad (4.9)$$

$$w^{2} - (\lambda + c)w + 1 = 0, \qquad (4.10)$$

respectively. If λ is non-real, then we define z and w to be the unique solutions of (4.9) and (4.10), respectively, which satisfy

$$|z| > 1, \qquad |w| > 1.$$
 (4.11)

Define the meromorphic function $f_n: \mathbb{C}^2 \to \mathbb{C}$ by

$$f_n(z,w) = (z^{n+1} - z^{-n-1})(w^{n+1} - w^{-n-1}) + (z^n - z^{-n})(w^n - w^{-n}).$$
(4.12)

The next result is the characteristic polynomial of the pencil A_c , deduced in [19], as an explicit complex polynomial equation in two variables.

Lemma 4.1.5 ([19, Lemma 2.4]). Let z, w, λ be related by (4.9)-(4.10). If $\sigma = z + z^{-1} \neq \pm 2$ and $\tau = w + w^{-1} \neq \pm 2$, then

$$\det(H_c^{(2n)} - \lambda S) = (-1)^n \frac{f_n(z, w)}{(z - z^{-1})(w - w^{-1})}.$$

Remark 4.1.6. Lemma 4.1.5 is not the exact statement of Lemma 2.4 in [19]. To be precise, Davies and Levitin [19] constructed the characteristic polynomial of a linear pencil $H_0^{(N)} - \lambda S_{m,n;\sigma,\tau}$ where $S_{m,n;\sigma,\tau}$ is the diagonal matrix

$$S_{m,n;\sigma,\tau} = \begin{pmatrix} \sigma I_m & \\ & \tau I_n \end{pmatrix},$$

and what we state here is the special case when m = n and $-\tau$ instead of τ .

Let $\mathcal{D}(o, r)$ be the disk of radius r around a point o, that is,

$$\mathcal{D}(o, r) := \{ \zeta \in \mathbb{C} : |\zeta - o| < r \},\$$

and let $\partial \mathcal{D}(o, r)$ be the boundary of the disk $\mathcal{D}(o, r)$. In addition, define the families of meromorphic function $G_n : \mathbb{C} \to \mathbb{C}$ by

$$G_n(\xi) := \frac{\xi^{n+1} - \xi^{-n-1}}{\xi^n - \xi^{-n}}.$$
(4.13)

Now, we quote another result from [19] which relates the eigenvalues of the pencil A to the function f_n and G_n .

Theorem 4.1.7 ([19, Theorem 2.3]). Let $\lambda \in \mathbb{C} \setminus \{-2 - c, -2 + c, 2 - c, 2 + c\}$ and let λ , *z*, *w* be related by (4.8)-(4.11). Then

(a) λ is an eigenvalue of the pencil A_c if and only if

$$f_n(z,w) = 0.$$
 (4.14)

(b) If λ is an eigenvalue of A_c , then it is real if and only if both z and w lie in the set

 $\partial \mathcal{D}(0,1) \cup (\mathbb{R} \setminus \{0\}).$

(c) If $\lambda \notin \mathbb{R}$ then λ is an eigenvalue of \mathcal{A}_c if and only if

$$G_n(z)G_n(w) = -1.$$
 (4.15)

Remark 4.1.8. Looking at the imaginary parts of (4.8), we see that

$$\operatorname{Im}\lambda = \operatorname{Im}\left(z + \frac{1}{z}\right) = \operatorname{Im}\left(w + \frac{1}{w}\right),$$

and re-arranging this we obtain

Im
$$\lambda = \text{Im}(z)\left(1 - \frac{1}{|z|^2}\right) = \text{Im}(w)\left(1 - \frac{1}{|w|^2}\right).$$
 (4.16)

If $\lambda \in \mathbb{C} \setminus \mathbb{R}$, then using |z| > 1 and |w| > 1 implies that corresponding Im(z) and Im(w) are of the same sign.

Remark 4.1.9. Let $\lambda \in \mathbb{R}$. Then Theorem 4.1.7(b) immediately follows from (4.16). In addition, we have from (4.9) and (4.10) that

$$z = \frac{\lambda - c \pm \sqrt{(\lambda - c)^2 - 4}}{2}, \qquad w = \frac{\lambda + c \pm \sqrt{(\lambda + c)^2 - 4}}{2}.$$
 (4.17)

It follows from Lemma 3.1.1 that $|\lambda| \ge 2 + |c|$ is impossible. Also it can be seen from (4.17) with account of Theorem 4.1.7(b) that

$$\lambda \in (-2 - c, -2 + c] \quad \Leftrightarrow \quad z \in \mathbb{R}, \qquad w \in \partial \mathcal{D}(0, 1), \tag{4.18}$$

$$\lambda \in (-2+c, 2-c) \qquad \Leftrightarrow \quad z \in \partial \mathcal{D}(0, 1), \quad w \in \partial \mathcal{D}(0, 1), \tag{4.19}$$

$$\lambda \in [2 - c, 2 + c) \qquad \Leftrightarrow \quad z \in \partial \mathcal{D}(0, 1), \quad w \in \mathbb{R}.$$
(4.20)

In addition, Theorem 4.1.7(b) states that if $\lambda \in \mathbb{R}$, and z and/or w are non-real, then they lie on the unit circle $\partial \mathcal{D}(0, 1)$.

4.1.3 A_c in the basis of eigenfunctions of H_0

Let the matrix Ψ have the entries $(\psi_{j,k})_{j,k=1}^n = \left(\sqrt{2/(n+1)}\sin(\pi jk/(n+1))\right)_{j,k=1}^n$, i.e.

$$\Psi := \sqrt{\frac{2}{n+1}} \begin{pmatrix} \sin\left(\frac{\pi}{n+1}\right) & \sin\left(\frac{2\pi}{n+1}\right) & \cdots & \sin\left(\frac{\pi n}{n+1}\right) \\ \sin\left(\frac{2\pi}{n+1}\right) & \sin\left(\frac{4\pi}{n+1}\right) & & \vdots \\ \vdots & & \ddots & \\ \sin\left(\frac{\pi n}{n+1}\right) & \cdots & \sin\left(\frac{\pi n^2}{n+1}\right) \end{pmatrix}.$$
(4.21)

Since the columns of Ψ are normalised eigenfunctions of H_0 which is Hermitian, Ψ is a unitary matrix, i.e.

$$\Psi = \Psi^* = \Psi^T = \Psi^{-1}.$$

In the orthonormal basis

$$\left\{ egin{pmatrix} oldsymbol{\psi}_1 \ oldsymbol{0} \end{pmatrix}, \dots, egin{pmatrix} oldsymbol{\psi}_n \ oldsymbol{0} \end{pmatrix}, egin{pmatrix} oldsymbol{0} \ oldsymbol{\psi}_1 \end{pmatrix}, \dots, egin{pmatrix} oldsymbol{0} \ oldsymbol{\psi}_n \end{pmatrix}
ight\},$$

where 0 represents the zero vector in \mathbb{R}^n , the problem (3.5) can be re-written as follows.

Lemma 4.1.10. Using the change of basis matrix $diag(\Psi, \Psi)$, the spectral problem (3.5) is equivalent to the problem

$$\begin{pmatrix} \mu_{1} - \sigma & \psi_{1,n}\psi_{n,n} & \cdots & \psi_{1,n}\psi_{1,n} \\ & \ddots & \vdots & & \vdots \\ & & \mu_{n} - \sigma & \psi_{n,n}\psi_{n,n} & \cdots & \psi_{n,n}\psi_{1,n} \\ \psi_{1,n}\psi_{n,n} & \cdots & \psi_{1,n}\psi_{1,n} & \mu_{1} + \tau & \\ \vdots & & \vdots & & \ddots & \\ \psi_{n,n}\psi_{n,n} & \cdots & \psi_{n,n}\psi_{1,n} & & & \mu_{n} + \tau \end{pmatrix} \begin{pmatrix} u_{1} \\ \vdots \\ u_{n} \\ v_{n} \\ \vdots \\ v_{1} \end{pmatrix} = \mathbf{0}.$$
(4.22)

Proof. Note that $(\operatorname{diag}(\Psi, \Psi))^{-1} = \operatorname{diag}(\Psi, \Psi)$. Consider the problem (3.5). Since for any non-singular matrix X, the spectrum of A and $X^{-1}AX$ are the same, the matrix on the left-hand side of (3.5) can be written in the following alternative form:

$$\begin{pmatrix} \Psi \\ \Psi \end{pmatrix} \begin{pmatrix} H_0 - \sigma I & B \\ B & H_0 + \tau I \end{pmatrix} \begin{pmatrix} \Psi \\ \Psi \end{pmatrix} = \begin{pmatrix} \Psi H_0 \Psi - \sigma I & \Psi B \Psi \\ \Psi B \Psi & \Psi H_0 \Psi + \tau I \end{pmatrix}.$$
 (4.23)

Calculating

$$\widetilde{H}_0 := \Psi H_0 \Psi = \operatorname{diag}(\mu_1, \dots, \mu_n), \tag{4.24}$$

$$\tilde{B} := \Psi B \Psi = (\psi_{n,j} \psi_{k,n})_{j,k=1}^n$$
(4.25)

Substituting these into (4.23) and recalling that $\psi_{j,k} = \psi_{k,j}$, we arrive at (4.22).

We will apply Gershgorin Theorem to the problem (4.22) and deduce the corresponding Gershgorin set in Section 4.5.3.3.

4.2 Reformulating Conjecture 1.3.3 in different forms

As can be seen in Theorem 1.3.4, there are several ways to reformulate Conjecture 1.3.3. Our aim in this section is to re-write the conjecture by using the different forms of characteristic equation.

4.2.1 Ratios of orthogonal polynomials

Define the family of meromorphic function $\widetilde{F}_n : \mathbb{C} \to \mathbb{C}$ by

$$\widetilde{F}_n(\zeta) := \frac{F_n(\zeta)}{F_{n-1}(\zeta)}.$$
(4.26)

In fact, the function $\widetilde{F}_n(\zeta)$ is equal to the ratios of Chebyshev polynomials of the second kind $U_n(\zeta/2) = F_n(\zeta)$. Then the following recurrence relation is immediate.

Proposition 4.2.1. Let $n \ge 1$. Then $\widetilde{F}_n(\zeta)$ satisfies the recurrence relation

$$\widetilde{F}_{n+1}(\zeta) = \zeta - \frac{1}{\widetilde{F}_n(\zeta)}, \qquad \widetilde{F}_1(\zeta) = \zeta.$$
(4.27)

Proof. Consider the recurrence relation (2.35). Then for n = 1

$$\widetilde{F}_1(\zeta) = \frac{F_1(\zeta)}{F_0(\zeta)} = \zeta$$

For $n \ge 2$, dividing the last equation of (2.35) by $F_{j-1}(\zeta)$ we arrive at (4.27).

One can see that functions \widetilde{F}_n and G_n are related to each other in the following manner. **Proposition 4.2.2.** Let $\xi \in \mathbb{C} \setminus \{0\}$ and $n \ge 1$. Then

$$G_n(\xi) = \widetilde{F}_n\left(\xi + \frac{1}{\xi}\right).$$
(4.28)

Proof. We proceed by induction. For n = 1, we have

$$G_1(\xi) = \frac{\xi^2 - \xi^{-2}}{\xi - \xi} = \xi + \frac{1}{\xi} = \widetilde{F}_1\left(\xi + \frac{1}{\xi}\right)$$

so (4.28) holds for n = 1. Now assume that (4.28) is true for some $n \in \mathbb{N}$. Then

$$G_{n+1}(\xi) = \frac{\xi^{n+2} - \xi^{-n-2}}{\xi^{n+1} - \xi^{-n-1}}$$

= $\frac{(\xi^{n+1} - \xi^{-n-1})(\xi - \xi^{-1}) - \xi^n + \xi^{-n}}{\xi^{n+1} - \xi^{-n-1}}$
= $\xi + \xi^{-1} - \frac{1}{G_n(\xi)} = \widetilde{F}_{n+1}\left(\xi + \frac{1}{\xi}\right).$

Thus, by induction, (4.28) holds for all $n \in \mathbb{N}$.

Using Proposition 4.2.1, the characteristic equation (4.5) can be reformulated in terms of \tilde{F}_n as follows.

Corollary 4.2.3. Let λ , σ , τ be related by (3.6), let \mathbf{u} , \mathbf{v} , σ , τ solve (4.1)-(4.2), and let σ , $\tau \notin \{\mu_n, \ldots, \mu_1\}$. Then $\lambda = (\sigma + \tau)/2$ is a simple eigenvalue of \mathcal{A}_c , and

$$\widetilde{F}_n(\sigma)\widetilde{F}_n(\tau) = -1.$$
 (4.29)

The equation (4.29) can also be deduced from Theorem 4.1.7 and Proposition 4.2.2. We now verify the reformulation of Conjecture 1.3.3 given as in Theorem 1.3.4(iii).

Proof of Theorem 1.3.4(iii). Let λ, σ, τ be related by (3.6). Then, by Corollary 4.2.3, λ is a non-real eigenvalue of \mathcal{A}_c if and only if $\sigma, \tau \in \mathbb{C} \setminus \mathbb{R}$ satisfy (4.29). We have by (3.6) that $\operatorname{Im}(\lambda) = \operatorname{Im}(\sigma) = \operatorname{Im}(\tau)$. Moreover, by Lemma 3.1.3 we have the symmetry $\lambda \to \overline{\lambda}$, which implies $\sigma \to \overline{\sigma}$ and $\tau \to \overline{\tau}$. Therefore it is sufficient to consider the non-real solutions of (4.29) with $\operatorname{Im}(\sigma) = \operatorname{Im}(\tau) > 0$.

We omit the proof of part (ii) since its reformulation is immediate by (4.26).

It is easy to see that if $|\sigma| \ge 2$ and $|\tau| \ge 2$, then there is no such $(\sigma, \tau) \in \mathbb{C}^2$ which satisfies (4.29). This is due to the following lemma.

Lemma 4.2.4. If $|\zeta| \ge 2$, then $\left|\widetilde{F}_n(\zeta)\right| > 1$ for all $n \ge 1$.

Proof. We proceed by induction. For n = 1, the statement is obvious. Now assume that $\left|\widetilde{F}_{n}(\zeta)\right| > 1$ holds for some $n \in \mathbb{N}$. Then

$$\left| |\zeta| - \frac{1}{\left| \widetilde{F}_n(\zeta) \right|} \right| \le \left| \zeta - \frac{1}{\widetilde{F}_n(\zeta)} \right| = \left| \widetilde{F}_{n+1}(\zeta) \right|.$$

Since $|\zeta| \ge 2$ and $\frac{1}{\left|\widetilde{F}_n(\zeta)\right|} < 1$, the left-hand side of the inequality is greater than one, and therefore $\left|\widetilde{F}_{n+1}(\zeta)\right| > 1$. Then, by induction, the statement holds for all $n \in \mathbb{N}$.

Using the three term recurrence relation (4.27), the function \tilde{F}_n can be written in a continued fraction expansion

$$\widetilde{F}_n(\zeta) = \zeta - \frac{1}{\zeta - \frac{1}{\ddots - \frac{1}{\zeta}}}$$

However, the main problem in here is that ζ is a complex number. In addition, since the functions $\widetilde{F}_n(\zeta)$ can be expressed in terms of the ratios of Chebyshev polynomials of the second kind, we can treat Conjecture 1.3.3 as a ratio of orthogonal polynomials. In Section 4.5.2, we will apply the ratio asymptotics, which were mentioned in Section 2.3.4, and obtain some bounds on the function $\widetilde{F}_n(\zeta)$.

4.2.2 The area between |z| = 1 and $|z \pm i|^2 < 2$

In Section 4.1.2, we used the change of variables (4.8) to re-write the characteristic equation in terms of z and w, as in Lemma 4.1.7(a) and (c). Therefore another way of expressing Conjecture 1.3.3 is looking at the equivalent statements of λ in terms of z and w, which was stated in the introduction. Our goal in this subsection is to prove and clarify Theorem 1.3.4(iv). We will then give some numerical examples to support this result. We will illustrate z and w values correspond to the non-real eigenvalues of A_c on the same complex plane. We also give the Gershgorin-type localisation result and its illustration in Section 4.5.3.3.

Remark 4.2.5. The characteristic equation of the pencil problem for non-real eigenvalues was given in Theorem 4.1.7(c). In fact, the equation (4.15) holds also for the real eigenvalues of A_c as long as $z^n \neq z^{-n}$ and $w^n \neq w^{-n}$. This is due to Theorem 4.1.7(a), and that

$$\frac{f_n(z,w)}{(z^n - z^{-n})(w^n - w^{-n})} = G_n(z)G_n(w) - 1.$$

It is also easy to see that if λ is non-real, then by Theorem 4.1.7(b), we have $z^n \neq z^{-n}$ and $w^n \neq w^{-n}$.

A direct calculation leads to the following auxiliary result.

Lemma 4.2.6. The roots of |z + 1/z| = 2 are exactly $z = \pm i + \sqrt{2}e^{i\varphi}$ where $\varphi \in [0, 2\pi]$, and therefore

$$\left|z + \frac{1}{z}\right| < 2 \iff |z \pm i| < \sqrt{2}. \tag{4.30}$$

We can now clarify the reformulation of Conjecture 1.3.3 in terms of corresponding z and w.

Proof of Theorem 1.3.4(iv). Recall that if λ is a non-real eigenvalue of the pencil A_c , we then define z, w to be the unique solutions of (4.9) and (4.10), respectively, which satisfy |z| > 1 and |w| > 1. Then by Theorem 4.1.7(b), λ is non-real if and only if $z, w \in \mathbb{C} \setminus \mathbb{R}$ with

|z| > 1 and |w| > 1. By Lemma 4.2.6, we can re-write the statement of the conjecture as any solutions z, w of (4.15) should lie in the set $\mathcal{Z}_1 \cup \mathcal{Z}_2$ where

$$\mathcal{Z}_1 = \mathcal{D}(i, \sqrt{2}) \setminus \overline{\mathcal{D}}(0, 1), \qquad \mathcal{Z}_2 = \mathcal{D}(-i, \sqrt{2}) \setminus \overline{\mathcal{D}}(0, 1),$$

where $\overline{\mathcal{D}}(o, r)$ denotes the closure of the disk $\mathcal{D}(o, r)$.

We present some numerical experiments in Figure 4.2. For simplicity, we omit the values of z and w which lie on the unit circle $\partial D(0,1)$ and on the real line. It is clear from (4.18)-(4.20) that these values correspond to the real eigenvalues of A_c . We take the set of non-real eigenvalues in the λ -plane and plot them in a different complex plane using (4.8). We superimpose corresponding values of z and w in the same complex plane by solving the quadratic equations (4.9) and (4.10) on a finely spaced grid of c values. The black dashed circle represents the unit circle $\partial D(0,1)$, the orange circle represents $\partial D(i, \sqrt{2})$ and the light-orange shaded area represents the set Z_1 . The green circle represents $\partial D(-i, \sqrt{2})$ and the light-green shaded area represents the set Z_2 . The red and blue curves, which correspond to values of z and w respectively, are exactly the non-real eigenvalue curves of the pencil A_c under the change of variables (4.8). For presentation purposes, in the four illustrations we zoom in on an area near the unit circle.

In Figure 4.2 we illustrate all non-real eigenvalue curves, however we would like to know only whether (1.19) is satisfied. Without seeing the dynamics, it is difficult to understand which two values (z and w) correspond to the same c. The only thing we know from Remark 4.1.8 is that $Im\lambda > 0$ implies Im(z) > 0 and Im(w) > 0. According to the figure, the eigenvalues (in terms of z and w) always appear from the unit circle and disappear to the unit circle. Conjecture 1.3.3 will be broken if one eigenvalue escapes from the unit circle and then travels to the real axis. In this case, it would have to cross the forbidden region. The important thing in this figure is that they do not cross the boundary of Z_1 and Z_2 . Conjecture 1.3.3 appear to be true since whatever escapes from the unit circle, returns to the unit circle without going through the forbidden area. Hence, both z and w, which correspond to the non-real eigenvalues, seem to lie in $Z_1 \cup Z_2$.

4.2.3 Intersection of Gershgorin disks

Recall that $\mathcal{G}(A)$ represents the Gershgorin set of A. Our main aim was to enhance the Gershgorin Theorem when applied to the non-real eigenvalues of the pencil \mathcal{A}_c (or the non-self-adjoint matrix \mathbf{A}_c), but we could not. In this section, we apply the Gershgorin Theorem to our problem. In Section 4.5.3, we will discuss the application of known Gershgorin-type results mentioned in Section 2.2 to our problem in more detail.



Figure 4.2: \mathcal{Z}_1 (orange dashed region), $\partial \mathcal{D}(i, \sqrt{2})$ (orange circle), \mathcal{Z}_2 (green dashed region), $\partial \mathcal{D}(-i, \sqrt{2})$ (green circle), $\partial \mathcal{D}(0, 1)$ (black dashed circle), drawn in the complex plane, together with the superimposition of z (red curves) and w (blue curves) values, which correspond to the non-real eigenvalues only, for c values between 0 and 2 with step-size of 10^{-3} . Left top: n = 3. Right top: n = 4. Left bottom: n = 7. Right bottom: n = 14.

The Gershgorin set $\mathcal{G}(\mathbf{A}_c)$ consists of four different disks defined by

$$\mathcal{G}_1(\mathbf{A}_c) = \overline{\mathcal{D}}(c, 1), \qquad \qquad \mathcal{G}_i(\mathbf{A}_c) = \overline{\mathcal{D}}(c, 2), \\ \mathcal{G}_j(\mathbf{A}_c) = \overline{\mathcal{D}}(-c, 2), \qquad \qquad \mathcal{G}_{2n}(\mathbf{A}_c) = \overline{\mathcal{D}}(-c, 1)$$

where i = 2, ..., n and j = n + 1, ..., N - 1. Then $\mathcal{G}_1(\mathbf{A}_c) \subset \mathcal{G}_i(\mathbf{A}_c)$ and $\mathcal{G}_{2n}(\mathbf{A}_c) \subset \mathcal{G}_j(\mathbf{A}_c)$, and therefore

$$\bigcup_{i=1}^{n} \mathcal{G}_{i}(\mathbf{A}_{c}) = \overline{\mathcal{D}}(c, 2), \qquad \bigcup_{j=n+1}^{2n} \mathcal{G}_{j}(\mathbf{A}_{c}) = \overline{\mathcal{D}}(-c, 2).$$

Hence, we obtain

$$\operatorname{Spec}(\mathbf{A}_c) \subseteq \mathcal{G}(\mathbf{A}_c) = \overline{\mathcal{D}}(c,2) \cup \overline{\mathcal{D}}(-c,2).$$

We illustrate the union in Figure 4.3.



Figure 4.3: For n = 11, red dots represent the eigenvalues of \mathbf{A}_c and the blue shaded region is the Gershgorin set $\mathcal{G}(\mathbf{A}_c)$, i.e. the union $\overline{\mathcal{D}}(c, 2) \cup \overline{\mathcal{D}}(-c, 2)$. The boundaries of each disk are represented by the blue circles. Left: c = 0.3. Right: c = 1.4.

However, when we focus on the non-real eigenvalues of A_c , we observe that they all seem to lie in the intersection of the Gershgorin disks. Note that

$$\mathcal{D}(c,2) \cap \mathcal{D}(-c,2) = \{\lambda \in \mathbb{C} : |\lambda - c| < 2 \text{ and } |\lambda + c| < 2\}.$$

Then this leads us to another reformulation of the conjecture as in Theorem 1.3.4(v), i.e. if $\lambda \in \text{Spec}(\mathbf{A}_c) \setminus \mathbb{R}$, then $\lambda \in \mathcal{D}(c, 2) \cap \mathcal{D}(-c, 2)$.

As shown in Section 2.2, there exists no Gershgorin-type localisation result which gives a bound only for the non-real eigenvalues of a matrix. All existing results give a bound for the whole spectrum of a matrix. In this case, we see that Conjecture 1.3.3 is somewhat stronger than Gershgorin Theorem. Unfortunately, we could not obtain any such region which contains only the non-real spectrum of A_c . Nevertheless, we have the following result.

Lemma 4.2.7. Conjecture 1.3.3 is satisfied for those $\lambda \in \text{Spec}(\mathcal{A}_c) \setminus \mathbb{R}$ for which $|\sigma| = |\tau|$ with λ, σ, τ related by (3.6).

Proof. The case $|\sigma| = |\tau|$ implies that

$$|\lambda - c| = |\lambda + c| \iff c = 0 \text{ or } \operatorname{Re}(\lambda) = 0,$$

i.e. λ is equidistant from both c and -c. In other words, these eigenvalues lie on the imaginary axis in the complex plane. Since the Gershgorin disks $\overline{\mathcal{D}}(c,2)$ and $\overline{\mathcal{D}}(-c,2)$ are symmetric with respect to the imaginary axis, the intersection as well as the union of two Gershgorin disks always cover the imaginary axis by Gershgorin Theorem. Therefore these purely imaginary eigenvalues always lie in the intersection of two Gershgorin disks which implies that both $|\sigma| = |\lambda - c|$ and $|\tau| = |\lambda + c|$ are smaller than two. Note that the boundaries does not play a role since the characteristic equation (4.29) is not satisfied by Lemma 4.2.4 if $|\sigma| = |\tau| = 2$.

4.3 Partial cases

The goal of this section is to prove that the Conjecture 1.3.3 holds for n = 2 and for n = 3. We will prove the equivalent statement of the conjecture given in Theorem 1.3.4(iii). The purpose of this section is to see that even for small size, the result is non-trivial, computations get messy and there is no pattern.

When n = 1, it is straightforward to see that Conjecture 1.3.3 is satisfied:

$$\sigma \tau = -1 \quad \Rightarrow \quad \tau = -\frac{1}{\sigma} = -\frac{\operatorname{Re}(\sigma) - i\operatorname{Im}(\sigma)}{|\sigma|^2}$$

and since $Im(\sigma) = Im(\tau)$, we obtain

$$\operatorname{Im}(\tau) = \frac{\operatorname{Im}(\sigma)}{|\sigma|^2} \quad \Rightarrow \quad |\sigma| = 1 \quad \Rightarrow \quad |\tau| = \frac{1}{|\sigma|} = 1,$$

as required.

Now, set

$$\sigma = x + iy, \qquad \tau = t + iy, \tag{4.31}$$

where $x, t \in \mathbb{R}$ and $y \in \mathbb{R} \setminus \{0\}$, and look for solutions of the characteristic equation (4.29) when n = 2 and n = 3. We will see in each situation that there are two cases: either $|\sigma| = |\tau|$ or $|\sigma| \neq |\tau|$. We know by Lemma 4.2.7 that Conjecture 1.3.3 is satisfied in the case $|\sigma| = |\tau|$ for all $n \in \mathbb{N}$. Nevertheless, we emphasise that we will include the proof of this case for the sake of completeness. In addition, recall from Corollary 3.1.8 that $|y| \leq 1$.

4.3.1 The case n = 2

For n = 2, the equation becomes

$$\left(\sigma - \frac{1}{\sigma}\right)\left(\tau - \frac{1}{\tau}\right) = -1 \quad \Leftrightarrow \quad -\sigma\tau = (\sigma^2 - 1)(\tau^2 - 1),$$

substituting (4.31) into (4.29) gives

$$-(x+iy)(t+iy) = (x^2 - y^2 - 1 + i2xy)(t^2 - y^2 - 1 + i2ty).$$

Separating the real and imaginary parts of the equation gives the system of two equations

$$y^{2} - xt = (x^{2} - y^{2} - 1)(t^{2} - y^{2} - 1) - 4xty^{2},$$
(4.32)

$$-(xy+yt) = 2ty(x^2 - y^2 - 1) + 2xy(t^2 - y^2 - 1).$$
(4.33)

Re-arranging equation (4.33) gives

$$(x+t)(1+2y^2-2xt) = 0.$$
 (4.34)

We now have two cases, either t = -x or $1 + 2y^2 - 2xt = 0$.

The first case: $|\sigma| = |\tau|$

Let t = -x which also implies that $|\sigma| = |\tau|$. Then using this in (4.32), we have, after simplification, that

$$4x^2 - 1 = (x^2 + y^2)^2 + (x^2 + y^2)$$

and since $4x^2 - 1 < 4|\sigma|^2 - 1$, the above equation reduces to

$$\sigma|^4 + |\sigma|^2 < 4|\sigma|^2 - 1$$

which implies

$$\left(|\sigma|^2 - \frac{3}{2}\right)^2 - \frac{5}{4} < 0,$$

and so

$$\frac{3}{2} - \sqrt{\frac{5}{4}} < |\sigma|^2 < \frac{3}{2} + \sqrt{\frac{5}{4}},$$

and since $|\sigma| = |\tau|$, we obtain $|\sigma|, |\tau| < 2$.

The second case: $|\sigma| \neq |\tau|$

Take $t \neq -x$. Re-arranging (4.32), we have

$$0 = (x^{2} - y^{2} - 1)(t^{2} - y^{2} - 1) - 4xty^{2} - y^{2} + xt$$

= $x^{2}t^{2} - x^{2}y^{2} - x^{2} - y^{2}t^{2} + y^{4} + y^{2} - t^{2} + y^{2} + 1 - 4xty^{2} - y^{2} + xt$
= $(xt)^{2} + xt - y^{2}(x^{2} + t^{2}) - (x^{2} + t^{2}) + y^{4} + y^{2} + 1 - 4y^{2}(xt)$

Then using $xt = y^2 + 1/2$ into the above equation, we obtain

$$\begin{split} 0 &= (y^4 + y^2 + \frac{1}{4}) - (y^2 + 1)(x^2 + t^2) + y^4 + y^2 + 1 - 4y^4 - 2y^2 + y^2 + \frac{1}{2} \\ &= -(y^2 + 1)(x^2 + t^2) - 2y^4 + 1 + \frac{1}{4} + y^2 + \frac{1}{2} \\ &= -(y^2 + 1)(x^2 + t^2 + 2y^2) + 2y^4 + 2y^2 - 2y^4 + y^2 + \frac{7}{4} \\ &= -(y^2 + 1)(|\sigma|^2 + |\tau|^2) + 3y^2 + \frac{7}{4} \end{split}$$

which implies

$$|\sigma|^2 + |\tau|^2 = \frac{3y^2 + \frac{7}{4}}{y^2 + 1} = 3 - \frac{5}{4(y^2 + 1)} < 3$$

for all y. Thus, $|\sigma|, |\tau| < 2$. Hence the conjecture is true for n = 2.

4.3.2 The case n = 3

When n = 3, equation (4.29) becomes

$$\left(\sigma - \frac{1}{\sigma - \frac{1}{\sigma}}\right) \left(\tau - \frac{1}{\tau - \frac{1}{\tau}}\right) = -1 \quad \Leftrightarrow \quad \left(\sigma - \frac{\sigma}{\sigma^2 - 1}\right) \left(\tau - \frac{\tau}{\tau^2 - 1}\right) = -1 \Leftrightarrow \quad (\sigma^3 - 2\sigma)(\tau^3 - 2\tau) = -(\sigma^2 - 1)(\tau^2 - 1).$$
(4.35)

First, re-arranging the imaginary part of LHS of (4.35), we have

$$\begin{split} \operatorname{Im}[(\sigma^{3} - 2\sigma)(\tau^{3} - 2\tau)] \\ &= \operatorname{Im}[(x^{3} - 3xy^{2} - 2x + i(3x^{2}y - 2y - y^{3}))(t^{3} - 3ty^{2} - 2t + i(3t^{2}y - 2y - y^{3}))] \\ &= (x^{3} - 3xy^{2} - 2x)(3t^{2}y - 2y - y^{3}) + (t^{3} - 3ty^{2} - 2t)(3x^{2}y - 2y - y^{3}) \\ &= -x^{3}y^{3} + 3x^{3}t^{2}y - 2x^{3}y + 3xy^{5} - 9xt^{2}y^{3} + 8xy^{3} - 6xt^{2}y + 4xy \\ &\quad -t^{3}y^{3} + 3x^{2}t^{3}y - 2t^{3}y + 3ty^{5} - 9x^{2}ty^{3} + 8ty^{3} - 6x^{2}ty + 4ty \\ &= -y^{3}(x^{3} + t^{3}) + 3x^{2}t^{2}y(x + t) - 2y(x^{3} + t^{3}) + 3y^{5}(x + t) \end{split}$$

$$-9xty^{3}(x+t) + 8y^{3}(x+t) - 6xty(x+t) + 4y(x+t)$$

= $(x+t)y[(-y^{2}-2)(x^{2}+t^{2}-xt) + 3x^{2}t^{2} + 3y^{4} - 9xty^{2} + 8y^{2} - 6xt + 4],$ (4.36)

and re-arranging the imaginary part of RHS of (4.35), we have

$$Im[-(\sigma^{2} - 1)(\tau^{2} - 1)] = Im[-(x^{2} - y^{2} - 1 + i2xy)(t^{2} - y^{2} - 1 + i2ty)]$$

$$= -(2ty(x^{2} - y^{2} - 1) + 2xy(t^{2} - y^{2} - 1))$$

$$= -2y(tx^{2} - ty^{2} - t + xt^{2} - xy^{2} - x)$$

$$= -2y[(x + t)(xt - y^{2} - 1)].$$
 (4.37)

Equating (4.36) and (4.37) we obtain

$$(x+t)y\left[(-y^2-2)(x^2+t^2-xt)+3x^2t^2+3y^4-9xty^2+6y^2-4xt+2\right]=0.$$
 (4.38)

Since $y \neq 0$, we again have two cases; either $|\sigma| = |\tau|$ or $|\sigma| \neq |\tau|$.

The first case: $|\sigma| = |\tau|$

The first case; let t = -x, i.e. $|\sigma| = |\tau|$. Then from the real part of LHS of (4.35), we have

$$\begin{aligned} \operatorname{Re}[(\sigma^{3} - 2\sigma)(\tau^{3} - 2\tau)] \\ &= \operatorname{Re}[(x^{3} - 3xy^{2} - 2x + \mathrm{i}(3x^{2}y - 2y - y^{3}))(t^{3} - 3ty^{2} - 2t + \mathrm{i}(3t^{2}y - 2y - y^{3}))] \\ &= (x^{3} - 3xy^{2} - 2x)(t^{3} - 3ty^{2} - 2t) - (3x^{2}y - 2y - y^{3})(3t^{2}y - 2y - y^{3}) \\ &= -(x^{3} - 3xy^{2} - 2x)^{2} - (3x^{2}y - 2y - y^{3})^{2} \\ &= -(x^{6} + 3x^{4}y^{2} + 3x^{2}y^{4} + y^{6} - 4x^{4} + 4x^{2} + 4y^{4} + 4y^{2}) \\ &= -[(x^{2} + y^{2})^{3} - 4x^{4} + 4y^{4} + 4(x^{2} + y^{2})] \\ &= -[\sigma|^{6} - 4|\sigma|^{2} + 4x^{4} - 4y^{4} \\ &= -|\sigma|^{6} - 4|\sigma|^{2} + 4(x^{2} + y^{2})^{2} - 8x^{2}y^{2} - 8y^{4} \\ &= -|\sigma|^{6} - 4|\sigma|^{2} + 4|\sigma|^{4} - 8y^{2}|\sigma|^{2}, \end{aligned}$$

$$(4.39)$$

and from the real part of RHS of (4.35),

$$\operatorname{Re}[-(\sigma^{2}-1)(\tau^{2}-1)] = \operatorname{Re}[-(x^{2}-y^{2}-1+i2xy)(z^{2}-y^{2}-1+i2zy)]$$

$$= -(x^{2}-y^{2}-1)(z^{2}-y^{2}-1)+2xy2zy$$

$$= -(x^{4}+y^{4}+1-2x^{2}y^{2}-2x^{2}+2y^{2}+4x^{2}y^{2})$$

$$= -[(x^{2}+y^{2})^{2}-2x^{2}+2y^{2}+1]$$

$$= -|\sigma|^{4}+2|\sigma|^{2}-4y^{2}-1.$$
(4.40)

Equating (4.39) and (4.40) yields

$$|\sigma|^{6} - 5|\sigma|^{4} + |\sigma|^{2}(6 + 8y^{2}) - 4y^{2} - 1 = 0.$$
(4.41)

Let $s := |\sigma|^2$ and denote

$$\widetilde{f}_y(s) := s^3 - 5s^2 + (6 + 8y^2)s - 4y^2 - 1,$$

then the equation (4.41) becomes

$$\widetilde{f}_y(s) = 0.$$

We now need to show that if s_0 is a root of $\tilde{f}_y(s)$ for some y > 0, then $s_0 < 4$. First, note that when s = 4, the function is positive for all y, i.e.

$$\widetilde{f}_y(4) = 7 + 28y^2 > 0.$$

Now, we want to show that the function is increasing for all y > 0 and s > 4. Taking the derivative with respect to *s* gives

$$\widetilde{f}'_y(s) = 3s^2 - 10s + 6 + 8y^2.$$

 $\widetilde{f}'_y(s)$ is increasing in s for $s\geq 5/3,$ therefore for $s\geq 4,$

$$\widetilde{f}'_y(s) \ge \widetilde{f}'_y(4) = 14 + 8y^2 > 0.$$

Thus, $\tilde{f}'_y(s) > 0$ for $s \ge 4$, and if there exists s_0 such that $\tilde{f}_y(s_0) = 0$, then $s_0 < 4$. Hence we prove that if $|\sigma| = |\tau|$, then $|\sigma|, |\tau| < 2$.

The second case: $|\sigma| \neq |\tau|$

The second case is to show if $|\sigma| \neq |\tau|$ then $|\sigma|, |\tau| < 2$. Proving this case is not as easy as the first case. We will see that all calculations get messy and the problem involves finding zeros of high degree polynomials.

Dividing (4.38) by y(x+t) and re-arranging it gives

$$3x^{2}t^{2} - 2xt(4y^{2} + 1) - (x^{2} + t^{2})(y^{2} + 2) + 3y^{4} + 6y^{2} + 2 = 0,$$

which implies

$$3x^{2}t^{2} = 2xt(4y^{2}+1) + (|\sigma|^{2}+|\tau|^{2})(y^{2}+2) - 2y^{4} - 4y^{2} - 3y^{4} - 6y^{2} - 2.$$

Let p := xt and $A := |\sigma|^2 + |\tau|^2$, then

$$3p^{2} - p(8y^{2} + 2) - A(y^{2} + 2) + 5y^{4} + 10y^{2} + 2 = 0.$$
 (4.42)

Now, the real part of (4.35), after simplification, is

$$\begin{aligned} x^3t^3 + x^2t^2(-9y^2+1) + xt(-(3y^2+2)(x^2+t^2) + 9y^4 + 8y^2 + 4) \\ + (x^2+t^2)(3y^4+5y^2-1) - y^6 - 3y^4 - 2y^2 + 1 &= 0. \end{aligned}$$

Using p = xt and $A = |\sigma|^2 + |\tau|^2$ yields

$$p^{3} + p^{2}(-9y^{2} + 1) + p(-A(3y^{2} + 2) + 15y^{4} + 12y^{2} + 4) + A(3y^{4} + 5y^{2} - 1) - 7y^{6} - 13y^{4} + 1 = 0.$$
(4.43)

The equation (4.42) implies that

$$3p^{2} = p\left(8y^{2} + 2\right) + A\left(y^{2} + 2\right) - \left(5y^{4} + 10y^{2} + 2\right).$$
(4.44)

Multiplying (4.44) by p and substituting this into (4.43) gives

$$p^{2} \left(\frac{8y^{2}+2}{3} - 9y^{2} + 1 \right) + p \left(A \frac{y^{2}+2}{3} - \frac{5y^{4} + 10y^{2}+2}{3} - A(3y^{2}+2) + 15y^{4} + 12y^{2} + 4 \right) + A(3y^{4} + 5y^{2} - 1) - 7y^{6} - 13y^{4} + 1 = 0,$$

which implies

$$p^{2} (-19y^{2} + 5) + p (A(-8y^{2} - 4) + 40y^{4} + 26y^{2} + 10) + 3A(3y^{4} + 5y^{2} - 1) - 3(7y^{6} + 13y^{4} - 1) = 0.$$

Substituting equation (4.44) into the above equation yields

$$(-19y^{2}+5) \left[p \left(8y^{2}+2 \right) + A \left(y^{2}+2 \right) - \left(5y^{4}+10y^{2}+2 \right) \right] + 3p \left(A (-8y^{2}-4) + 40y^{4}+26y^{2}+10 \right) + 9A (3y^{4}+5y^{2}-1) - 9(7y^{6}+13y^{4}-1) = 0,$$

which implies

$$p\left(\left(-19y^{2}+5\right)\left(8y^{2}+2\right)-3A(8y^{2}+4)+120y^{4}+78y^{2}+30\right)\right)$$

+ $A\left(\left(-19y^{2}+5\right)\left(y^{2}+2\right)+27y^{4}+45y^{2}-9\right)$

+
$$(19y^2 - 5)(5y^4 + 10y^2 + 2) - 9(7y^6 + 13y^4 - 1) = 0,$$

and so

$$p\left(3A(8y^2+4)+32y^4-80y^2-40\right) = A(8y^4+12y^2+1)+32y^6+48y^4-12y^2-1.$$

Resolving the last equation with respect to p and substituting back into (4.42) gives

$$\begin{split} \widehat{f}_{y}(A) &:= \frac{-144A^{3}(y^{2}+2)(2y^{2}+1)^{2}+9A^{2}(640y^{6}+1776y^{4}+1208y^{2}+243)}{16(A(6y^{2}+3)+8y^{4}-20y^{2}-10)^{2}} \\ &- 9A\frac{(512y^{8}+3584y^{6}+6256y^{4}+3328y^{2}+558)}{16(A(6y^{2}+3)+8y^{4}-20y^{2}-10)^{2}} \\ &+ \frac{20736y^{8}+75456y^{6}+77328y^{4}+27432y^{2}+3123}{16(A(6y^{2}+3)+8y^{4}-20y^{2}-10)^{2}} = 0. \end{split}$$

We want to show that if A_0 is a root of $\hat{f}_y(A)$ for some $0 < y \le 1$, then $A_0 < 4$. First, we show that $\hat{f}_y(4)$ is negative for all $y \in (0, 1]$:

$$\widehat{f}_y(4) = \frac{9(256y^8 + 192y^6 - 304y^4 - 152y^2 - 45)}{64(4y^4 + 2y^2 + 1)^2}.$$

The denominator of $\widehat{f_y}(4)$ is positive. By using $y^8 < y^6 < y^4$ and $-y^2 < -y^4$ we obtain

$$64(4y^4 + 2y^2 + 1)^2 \hat{f}_y(4) = 2304y^8 + 1728y^6 - 2736y^4 - 1368y^2 - 405$$

$$< 2304y^4 + 1728y^4 - 2736y^4 - 1368y^4 - 405$$

$$= -72y^4 - 405 < 0$$

which implies

$$\widehat{f}_{y}(4) < 0, \qquad \forall y \in (0, 1].$$
 (4.45)

Second, taking the derivative with respect to A, we will show that $\widehat{f_y}(A)$ is decreasing for all $0 < y \le 1$ and A > 4. A direct calculation gives

$$\hat{f}_{y}(A) = \frac{k_{3}A^{3} + k_{2}A^{2} + k_{1}A + k_{0}}{k_{-1}},$$

where

$$\begin{aligned} k_3(y) &= -1728y^8 - 6048y^6 - 6480y^4 - 2808y^2 - 432, \\ k_2(y) &= -6912y^{10} - 3456y^8 + 44928y^6 + 61344y^4 + 28080y^2 + 4320, \\ k_1(y) &= 59904y^{10} + 116352y^8 - 73008y^6 - 185472y^4 - 92466y^2 - 14337, \\ k_0(y) &= -18432y^{12} - 207360y^{10} - 394560y^8 - 85824y^6 + 164376y^4 + 98946y^2 + 15741 \end{aligned}$$

$$k_{-1}(A, y) = 8(A(6y^2 + 3) + 8y^4 - 20y^2 - 10)^3.$$

It is easy to see that the denominator $k_{-1}(A, y)$ is always positive when A > 4, and therefore we will ignore it since it does not play a role to the sign of $\hat{f}'_y(A)$. To understand the sign of the numerator of $\hat{f}'_y(A)$, we need the following result.

Lemma 4.3.1. For any $0 < y \le 1$,

(i) $4^{3}k_{3} + 4^{2}k_{2} + 4k_{1} + k_{0} < 0.$ (ii) $k_{1} + k_{2} + k_{3} < 0.$ (iii) $k_{1} + k_{3} < 0.$

Proof.

(i) We have

$$4^{3}k_{3} + 4^{2}k_{2} + 4k_{1} + k_{0} = -18432y^{12} - 78336y^{10} - 95040y^{8} - 46080y^{6} - 10728y^{4} - 1350y^{2} - 135$$

which is always negative.

(ii) We have

$$k_1 + k_2 + k_3 = 52992y^{10} + 111168y^8 - 34128y^6 - 130608y^4 - 67194y^2 - 10449.$$

By using $y^{10} < y^8 < y^4$ and $-y^2 < -y^4$, we obtain

$$k_1 + k_2 + k_3 < 52992y^4 + 111168y^4 - 34128y^6 - 130608y^4 - 67194y^4 - 10449$$
$$= -34128y^6 - 33642y^4 - 10449$$

which is also negative.

(iii) We have

$$k_1 + k_3 = 59904y^{10} + 114624y^8 - 79056y^6 - 191952y^4 - 95274y^2 - 14769y^6 - 191952y^6 -$$

By using $y^{10} < y^8 < y^6$ and $-y^4 < -y^6$, we get

$$k_1 + k_3 < 59904y^6 + 114624y^6 - 79056y^6 - 191952y^6 - 95274y^2 - 14769$$
$$= -96480y^6 - 95274y^2 - 14769 < 0$$

as required.

Now, consider the numerator of $\hat{f}'_y(A)$. Using, consecutively, Lemma 4.3.1(i), (ii) and then (iii), we have

$$k_{3}A^{3} + k_{2}A^{2} + k_{1}A + k_{0} < k_{3}(A^{3} - 4^{3}) + k_{2}(A^{2} - 4^{2}) + k_{1}(A - 4)$$

$$< k_{3}(A^{3} - 4^{3} - A^{2} + 4^{2}) + k_{1}(-A^{2} + 4^{2} + A - 4)$$

$$< k_{3}(A^{3} - 4^{3} - A^{2} + 4^{2} + A^{2} - 4^{2} - A + 4)$$

$$= k_{3}(A - 4)(A^{2} + 4A + 15) < 0,$$

for all $0 < y \le 1$ and all $A \ge 4$. Thus,

$$\hat{f}'_{y}(A) < 0 \qquad \forall A \ge 4, \forall y \in (0, 1].$$
 (4.46)

Combining (4.45) and (4.46), we conclude that if there exists A_0 such that $\hat{f}_y(A_0) = 0$, then $A_0 < 4$.

4.4 Formal asymptotic expansion as $c \rightarrow 0$

As we mentioned in Theorem 3.1.2, all eigenvalues of A_0 (i.e. A_c when c = 0) are nonreal, and Davies and Levitin [19] determined the asymptotic behaviour of the eigenvalues of the pencil A_c for large n. When c increases, each complex conjugate pair moves towards each other and meets on the real line. Numerical experiments show that eigenvalues move along a curve until each pair collide for the first time on the real line (see Level-1 in Table 5.1). Our aim in this section is to understand and describe the curve corresponding to the non-real eigenvalues of A_c as c goes to zero. The investigation will be carried out by using a formal asymptotic expansion method.

Recall the substitution (4.8) and note that each non-real eigenvalue λ corresponds to one non-real z and one non-real w value, which lie outside of the unit disk. When c = 0, the relation (4.8) reduces to

$$\lambda_0 = z_0 + \frac{1}{z_0} = w_0 + \frac{1}{w_0},\tag{4.47}$$

where λ_0 denotes the eigenvalues of \mathcal{A}_0 . In addition, $z_0, w_0 \in \mathbb{C} \setminus \mathbb{R}$ such that $|z_0|, |w_0| > 1$. Then, for simplicity, we can take $z_0 = w_0$. Therefore, when c = 0 the reduced characteristic equation (4.15) becomes

$$(G_n(z_0))^2 = -1 \quad \Rightarrow \quad G_n(z_0) = \pm i. \tag{4.48}$$
Davies and Levitin [19] found that as $n \to \infty$ any solutions of (4.48), which lie in the first quadrant, satisfy

$$z_0 = r e^{i\theta},\tag{4.49}$$

where

$$r = \left(\tan\left(\frac{\pi}{4} + \frac{\theta}{2}\right) \right)^{1/2n}, \qquad \theta = \frac{2\pi k}{2n+1}, \qquad k = 1, \dots, \left\lfloor \frac{n}{2} \right\rfloor,$$

up to order $O(n^{-1})$. We then extend the result to the other quadrants by using symmetries described in Lemma 3.1.3 and the relation (4.47). If *n* is odd, there are additionally two purely z_0 values which satisfy

$$z_0 = \pm i \left(1 + \frac{\log(n)}{n} (1 + o(1)) \right).$$
(4.50)

The idea is to fix the size of matrix $n \in \mathbb{N}$ and seek a solution of the form of a power series in c, i.e.

$$\lambda = \lambda_0 + c\lambda_1 + c^2\lambda_2 + O\left(c^3\right), \qquad (4.51)$$

$$z = z_0 + cz_1 + c^2 z_2 + O(c^3), \qquad (4.52)$$

$$w = w_0 + cw_1 + c^2 w_2 + O(c^3).$$
(4.53)

We will then find $\lambda_1, \lambda_2, z_1, z_2, w_1, w_2$ in terms of z_0 . Afterwards, using the asymptotics for λ_0 and $z_0 = w_0$, i.e. (4.49), we will find an approximate solution to the behaviour of the eigenvalues of A_c when c is small. We will mainly use the relation (4.8) and the characteristic equation (4.15).

Before stating our main result, we introduce some functions. Let $\zeta \in \mathbb{C}$, then define

$$g_n = g_n(\zeta) := \zeta^n + \zeta^{-n},$$
 (4.54)

$$b_n = b_n(\zeta) := \zeta^n - \zeta^{-n},$$
 (4.55)

$$Q_n(\zeta) = \frac{(n+1)g_{n+1} - ng_n G_n(\zeta)}{\zeta b_n},$$
(4.56)

$$\widehat{Q}_{n}(\zeta) = \zeta^{-2} \left(G_{n}(\zeta) \left(n \frac{\zeta^{n}}{b_{n}} + n^{2} \frac{g_{n}^{2}}{b_{n}^{2}} \right) - \frac{n+1}{b_{n}} \left(\zeta^{-n-1} + n \frac{g_{n}g_{n+1}}{b_{n}} \right) \right).$$
(4.57)

Our main result in this section is the following.

Theorem 4.4.1. Let $\lambda_0 \notin \mathbb{R}$ be an eigenvalue of \mathcal{A}_0 , let z_0 be the unique solution of (4.47) outside the unit disk. Then there is an eigenvalue λ of \mathcal{A}_c such that

$$\lambda = z_0 + \frac{1}{z_0} + c^2 \left(\frac{z_0^2}{2(z_0^2 - 1)} \left(\frac{Q_n^2(z_0) - 2G_n(z_0)\widehat{Q}_n(z_0)}{G_n(z_0)Q_n(z_0)} \right) + \frac{z_0}{(z_0^2 - 1)^2} \right) + O\left(c^3\right), \quad (4.58)$$

and moreover

$$z = z_0 + c \left(\frac{z_0^2}{1 - z_0^2}\right) + c^2 \left(\frac{z_0^4}{2(z_0^2 - 1)^2} \left(\frac{Q_n^2(z_0) - 2G_n(z_0)\widehat{Q}_n(z_0)}{G_n(z_0)Q_n(z_0)}\right)\right) + O(c^3),$$

$$w = z_0 - c \left(\frac{z_0^2}{1 - z_0^2}\right) + c^2 \left(\frac{z_0^4}{2(z_0^2 - 1)^2} \left(\frac{Q_n^2(z_0) - 2G_n(z_0)\widehat{Q}_n(z_0)}{G_n(z_0)Q_n(z_0)}\right)\right) + O(c^3),$$

where λ, z, w are related by (4.8).

Remark 4.4.2. Combining (4.58) and (4.51), we can easily deduce that $\lambda_1 = 0$, thus the perturbation of eigenvalues is quadratic in *c*. Note that our asymptotic procedure uses the fact that the eigenvalues are simple. Therefore the results will not work after the first collisions, because when we have a double eigenvalue, the characteristic equation is no longer an analytic function of a parameter, and therefore we cannot expand it into a series.

Before proving the result, we start with some numerical experiments. The results are illustrated in the $(\text{Re}\lambda, 2n\text{Im}\lambda)$ -plane in Figure 4.4, and in terms of z and w values in the complex plane in Figure 4.5. We know that all eigenvalues of A_c are non-real when c = 0. As c increases, the figures show that our asymptotics catch the eigenvalues until they become real for the first time (or until z and w hit the unit circle for the first time). We can see that our asymptotics, as expected, do not catch the eigenvalues after they become real. The corresponding z and w values move along the unit circle whereas the asymptotic expansion that we found keep moving as c increases.



Figure 4.4: Spec(A_c) in the (Re λ , 2nIm λ)-plane with λ values (red stars) and their asymptotic values (white circles) given by Theorem 4.4.1. Left: for n = 20 and c = 0.015. Right: for n = 50 and c = 0.01.

In order to prove Theorem 4.4.1, we need some auxiliary results. The next result provides a formula which converts a ratio of two polynomials to a geometric series.



Figure 4.5: Spec(A_c) in the complex plane with z values (red stars), w values (blue triangles), asymptotic values of z (white circles), asymptotic values of w (white diamonds), the unit circle $\partial D(0,1)$ (dotted circle). Left: for n = 20 and c = 0.015. Right: for n = 50 and c = 0.01.

Lemma 4.4.3. Let $a, b, \gamma, \delta, \rho, \phi \in \mathbb{C}$. As $c \to 0$,

$$\frac{a+c\gamma+c^2\rho}{b+c\delta+c^2\phi} = \frac{a}{b} + c\left(\frac{\gamma}{b} - a\frac{\delta}{b^2}\right) + c^2\left(\frac{\rho}{b} - \frac{\gamma\delta+a\phi}{b^2} + \frac{a\delta^2}{b^3}\right) + O\left(c^3\right).$$

Proof. As $c \to 0$, expanding the ratio as a geometric series up to terms of order $O(c^3)$ gives

$$\begin{aligned} \frac{a+c\gamma+c^2\rho}{b+c\delta+c^2\phi} &= \frac{a+c\gamma+c^2\rho}{b} \left(\frac{1}{1+c\frac{\delta}{b}+c^2\frac{\phi}{b}}\right) + O(c^3) \\ &= \frac{a+c\gamma+c^2\rho}{b} \left(1-c\frac{\delta}{b}-c^2\frac{\phi}{b}+\left(c\frac{\delta}{b}+c^2\frac{\phi}{b}\right)^2\right) + O(c^3) \\ &= \frac{a+c\gamma+c^2\rho}{b} \left(1-c\frac{\delta}{b}+c^2\left(\frac{\delta^2}{b^2}-\frac{\phi}{b}\right)\right) + O(c^3) \\ &= \left(a+c\gamma+c^2\rho\right) \left(\frac{1}{b}-\frac{\delta}{b^2}c+c^2\left(\frac{\delta^2}{b^3}-\frac{\phi}{b^2}\right)\right) + O(c^3) \\ &= \frac{a}{b}+c\left(\frac{\gamma}{b}-a\frac{\delta}{b^2}\right) + c^2\left(\frac{\rho}{b}-\frac{\gamma\delta+a\phi}{b^2}+\frac{a\delta^2}{b^3}\right) + O(c^3). \end{aligned}$$

In the next result, we derive an asymptotic expansion for the function $G_n(z)$ by using Lemma 4.4.3.

Lemma 4.4.4. Suppose that $c \rightarrow 0$, and $z = z_0 + cz_1 + c^2 z_2 + O(c^3)$. Then

$$G_n(z) = G_n(z_0) + cz_1Q_n(z_0) + c^2\left(z_2Q_n(z_0) + z_1^2\widehat{Q}_n(z_0)\right) + O\left(c^3\right),$$

where the functions Q_n , \hat{Q}_n depend only z_0 and are as given in (4.56)-(4.57).

Proof. Substituting the asymptotic expansion (4.52) into the function $G_n(z)$ we have

$$G_n(z) = G_n(z_0 + cz_1 + c^2 z_2 + O(c^3))$$

= $\frac{(z_0 + cz_1 + c^2 z_2)^{n+1} - (z_0 + cz_1 + c^2 z_2)^{-n-1}}{(z_0 + cz_1 + c^2 z_2)^n - (z_0 + cz_1 + c^2 z_2)^{-n}} + O(c^3).$ (4.59)

Now, calculating each power separately we get

$$z^{n+1} = (z_0 + cz_1 + c^2 z_2)^{n+1} = z_0^{n+1} + c(n+1)z_0^n z_1 + c^2(n+1)z_0^{n-1} \left(z_0 z_2 + \frac{n}{2}z_1^2\right) + O\left(c^3\right)$$

and

$$z^{-n-1} = (z^{n+1})^{-1} = (z_0^{n+1} + c(n+1)z_0^n z_1 + c^2(n+1)z_0^{n-1} (z_0 z_2 + \frac{n}{2}z_1^2))^{-1} + O(c^3)$$

$$= z_0^{-n-1} \left(1 + c(n+1)\frac{z_1}{z_0} + c^2(n+1) \left(\frac{z_2}{z_0} + \frac{n}{2}\frac{z_1^2}{z_0^2}\right) \right)^{-1} + O(c^3)$$

$$= z_0^{-n-1} \left(1 - c(n+1)\frac{z_1}{z_0} - c^2(n+1) \left(\frac{z_2}{z_0} - \frac{n}{2}\frac{z_1^2}{z_0^2}\right) + c^2(n+1)^2\frac{z_1^2}{z_0^2}\right) + O(c^3)$$

$$= z_0^{-n-1} \left(1 - c(n+1)\frac{z_1}{z_0} - c^2(n+1) \left(\frac{z_2}{z_0} - \left(\frac{n}{2} + 1\right)\frac{z_1^2}{z_0^2}\right) \right) + O(c^3)$$

$$= z_0^{-n-1} - c(n+1)\frac{z_1}{z_0^{n+2}} - c^2(n+1) \left(\frac{z_2}{z_0^{n+2}} - \left(\frac{n}{2} + 1\right)\frac{z_1^2}{z_0^{n+3}}\right) + O(c^3)$$

Similarly,

$$z^{n} = z_{0}^{n} + cnz_{0}^{n-1}z_{1} + c^{2}nz_{0}^{n-2}\left(z_{0}z_{2} + \frac{n-1}{2}z_{1}^{2}\right) + O\left(c^{3}\right)$$
$$z^{-n} = z_{0}^{-n} - cn\frac{z_{1}}{z_{0}^{n+1}} - c^{2}n\left(\frac{z_{2}}{z_{0}^{n+1}} - \left(\frac{n+1}{2}\right)\frac{z_{1}^{2}}{z_{0}^{n+2}}\right) + O\left(c^{3}\right).$$

Then

$$z^{n+1} - z^{-n-1} = z_0^{n+1} - z_0^{-n-1} + c(n+1)z_1(z_0^n + z_0^{-n-2}) + c^2(n+1)\left(z_2\left(z_0^n + z_0^{-n-2}\right) + z_1^2 z_0^{-2} \frac{n}{2}\left(z_0^{n+1} - z_0^{-n-1}\right) - z_1^2 z_0^{-n-3}\right) + O(c^3),$$
(4.60)

and

$$z^{n} - z^{-n} = z_{0}^{n} - z_{0}^{-n} + cnz_{1}(z_{0}^{n-1} + z_{0}^{-n-1})$$

$$+ c^{2}n\left(z_{2}\left(z_{0}^{n-1}+z_{0}^{-n-1}\right)+z_{1}^{2}z_{0}^{-2}\frac{n-1}{2}\left(z_{0}^{n}-z_{0}^{-n}\right)-z_{1}^{2}z_{0}^{-n-2}\right)+O(c^{3}).$$
 (4.61)

Substituting (4.60) and (4.61) into the function (4.59), we have a ratio of two polynomials in *c* of degree 2. Therefore we will apply to Lemma 4.4.3 so as to write the function G_n as a polynomial in *c*. The complex constants in Lemma 4.4.3 can be matched as

$$\begin{split} a &= z_0^{n+1} - z_0^{-n-1}, \\ b &= b_n = z_0^n - z_0^{-n}, \\ \gamma &= (n+1)z_1 z_0^{-1} g_{n+1}, \\ \delta &= n z_1 z_0^{-1} g_n, \\ \rho &= (n+1) \left(z_2 z_0^{-1} g_{n+1} + \frac{n}{2} z_1^2 z_0^{-2} \left(z_0^{n+1} - z_0^{-n-1} \right) - z_1^2 z_0^{-n-3} \right), \\ \phi &= n \left(z_2 z_0^{-1} g_n + \frac{n-1}{2} z_1^2 z_0^{-2} \left(z_0^n - z_0^{-n} \right) - z_1^2 z_0^{-n-2} \right). \end{split}$$

Then applying Lemma 4.4.3 gives, at order $O(c^0)$,

$$\frac{a}{b} = \frac{z_0^{n+1} - z_0^{-n-1}}{z_0^n - z_0^{-n}} = G_n(z_0),$$

and at order $O(c^1)$,

$$\frac{\gamma}{b} - \frac{a\delta}{b^2} = z_1 \left(\frac{(n+1)z_0^{-1}g_{n+1} - nz_0^{-1}g_nG_n(z_0)}{z_0^n - z_0^{-n}} \right) = z_1 Q_n(z_0).$$

To find the coefficient of c^2 in (4.59), we have

$$\frac{\rho}{b} - \frac{\gamma\delta + a\phi}{b^2} + \frac{a\delta^2}{b^3} = \frac{\rho}{b} - \frac{\gamma\delta}{b^2} - \frac{\phi}{b}G_n(z_0) + \frac{\delta^2}{b^2}G_n(z_0).$$

Then at order $O(c^2)$,

$$(n+1)\left(z_{2}z_{0}^{-1}\frac{g_{n+1}}{b_{n}} + \frac{n}{2}z_{1}^{2}z_{0}^{-2}G_{n}(z_{0}) - \frac{z_{1}^{2}z_{0}^{-n-3}}{b_{n}}\right) - n(n+1)z_{1}^{2}z_{0}^{-2}\frac{g_{n}g_{n+1}}{b_{n}}$$
$$- nG_{n}(z_{0})\left(z_{2}z_{0}^{-1}\frac{g_{n}}{b_{n}} + \left(\frac{n-1}{2}\right)z_{1}^{2}z_{0}^{-2} - \frac{z_{1}^{2}z_{0}^{-n-2}}{b_{n}}\right) + n^{2}G_{n}(z_{0})z_{1}^{2}z_{0}^{-2}\frac{g_{n}^{2}}{b_{n}^{2}}$$

which can be simplified to

$$z_2 Q_n(z_0) + z_1^2 \widehat{Q}_n(z_0),$$

as required.

The next result will be helpful when determining the terms at order O(c).

Lemma 4.4.5. Suppose $z \in \mathbb{C} \setminus \mathbb{R}$ such that |z| > 1 and $G_n^2(z) = -1$. Then $Q_n(z) \neq 0$. *Proof.* Suppose that $\exists z \in \mathbb{C} \setminus \mathbb{R}$ such that $G_n^2(z) = -1$ and $Q_n(z) = 0$. We have

$$G_n^2(z) = \frac{z^{2n+2} + z^{-2n-2} - 2}{z^{2n} + z^{-2n} - 2} = -1 \iff z^{2n+2} + z^{-2n-2} = -z^{2n} - z^{-2n} + 4.$$

Then

$$Q_n(z) = 0 \Leftrightarrow (n+1)(z^{n+1} + z^{-n-1}) = n(z^n + z^{-n})G_n(z)$$

$$\Rightarrow (n+1)^2(z^{2n+2} + z^{-2n-2} + 2) = n^2(z^{2n} + z^{-2n} + 2)(-1)$$

$$\Leftrightarrow (n^2 + 2n + 1)(-z^{2n} - z^{-2n} + 6) = -n^2(z^{2n} + z^{-2n} + 2).$$

Now let $t = z^{2n} + z^{-2n}$. Then

$$(n^{2} + 2n + 1)(-t + 6) = -n^{2}(t + 2) \iff 6n^{2} - n^{2}t + 12n - 2nt + 6 - t = -n^{2}t - 2n^{2}$$
$$\iff 8n^{2} + 12n + 6 = t(2n + 1).$$

Looking at the imaginary parts of the last equation, we deduce that

$$\operatorname{Im}(t) = 0 \iff \operatorname{Im}(z^{2n} + z^{-2n}) = 0 \iff \operatorname{Im}(z^{2n}) = -\operatorname{Im}\left(\frac{1}{z^{2n}}\right) = \frac{\operatorname{Im}(z^{2n})}{|z^{2n}|^2} \iff |z|^{4n} = 1.$$

Therefore, if $\exists z$ such that $G_n^2(z) = -1$ and $Q_n(z) = 0$, then |z| = 1 which is a contradiction.

We can now prove our main result.

Proof of Theorem 4.4.1. First, recall that z_0 and w_0 correspond to the eigenvalues of A_0 which are all non-real and $|z_0|, |w_0| > 1$. Also, we take $z_0 = w_0$. Note that the functions Q_n, \hat{Q}_n depend only on z_0 and therefore, we have from Lemma 4.4.4 that

$$G_n(z) = G_n(z_0) + cz_1Q_n(z_0) + c^2\left(z_2Q_n(z_0) + z_1^2\widehat{Q}_n(z_0)\right) + O\left(c^3\right),$$
(4.62)

and

$$G_n(w) = G_n(z_0) + cw_1 Q_n(z_0) + c^2 \left(w_2 Q_n(z_0) + w_1^2 \widehat{Q}_n(z_0) \right) + O\left(c^3\right).$$
(4.63)

Then substituting (4.62) and (4.63) into the characteristic equation (4.15), and then equating powers of c, we find at order $O(c^0)$ that

$$(G_n(z_0))^2 = -1 \quad \Rightarrow \quad G_n(z_0) = \pm i$$

Note that $Q_n(z_0) \neq 0$ by Conjecture 4.4.5, then we have at order $O(c^1)$ that

$$(z_1 + w_1)Q_n(z_0)G_n(z_0) = 0 \quad \Rightarrow \quad w_1 = -z_1,$$

and at order $O(c^2)$ that

$$(z_2 + w_2)G_n(z_0)Q_n(z_0) + 2z_1^2G_n(z_0)\widehat{Q}_n(z_0) - z_1^2Q_n^2(z_0) = 0.$$
(4.64)

Now, recall equation (4.8):

$$\lambda = z + \frac{1}{z} + c = w + \frac{1}{w} - c$$

Expanding z + 1/z we have

$$z + \frac{1}{z} = z_0 + cz_1 + c^2 z_2 + \frac{1}{z_0 + cz_1 + c^2 z_2} + O(c^3)$$

$$= z_0 + cz_1 + c^2 z_2 + \frac{1}{z_0} \left(\frac{1}{1 + c\frac{z_1}{z_0} + c^2 \frac{z_2}{z_0^2}} \right) + O(c^3)$$

$$= z_0 + cz_1 + c^2 z_2 + \frac{1}{z_0} \left(1 - c\frac{z_1}{z_0} - c^2 \frac{z_2}{z_0^2} + \left(c\frac{z_1}{z_0} + c^2 \frac{z_2}{z_0^2} \right)^2 \right) + O(c^3)$$

$$= z_0 + \frac{1}{z_0} + cz_1 \left(1 - \frac{1}{z_0^2} \right) + c^2 \left(z_2 \left(1 - \frac{1}{z_0^2} \right) + \frac{z_1^2}{z_0^3} \right) + O(c^3) .$$
(4.65)

Similarly,

$$w + \frac{1}{w} = w_0 + \frac{1}{w_0} + cw_1 \left(1 - \frac{1}{w_0^2}\right) + c^2 \left(w_2 \left(1 - \frac{1}{w_0^2}\right) + \frac{w_1^2}{w_0^3}\right) + O\left(c^3\right).$$
(4.67)

Substituting the asymptotic expansions (4.66) and (4.67) into (4.8), and equating terms at each order we have at order $O(c^0)$ that

$$\lambda_0 = z_0 + \frac{1}{z_0} = w_0 + \frac{1}{w_0}$$

and we take $w_0 = z_0$. At order $O(c^1)$ we get

$$\lambda_1 = z_1 \left(1 - \frac{1}{z_0^2} \right) + 1 = w_1 \left(1 - \frac{1}{w_0^2} \right) - 1 \quad \Rightarrow \quad \lambda_1 = \frac{z_1 + w_1}{2} \left(1 - \frac{1}{z_0^2} \right) = 0,$$

which also implies

$$w_1 = -z_1 = \frac{z_0^2}{z_0^2 - 1}.$$
(4.68)

We have at order $O\left(c^2\right)$ that

$$\lambda_2 = z_2 \left(1 - \frac{1}{z_0^2} \right) + \frac{z_1^2}{z_0^3} = w_2 \left(1 - \frac{1}{w_0^2} \right) + \frac{w_1^2}{w_0^3} \quad \Rightarrow \quad z_2 = w_2.$$
(4.69)

Now, substituting $w_1 = -z_1$ and $w_2 = z_2$ into (4.64) gives

$$2z_2G_n(z_0)Q_n(z_0) = z_1^2 \left(Q_n^2(z_0) - 2G_n(z_0)\widehat{Q}_n(z_0) \right),$$

then substituting (4.68) into above, we can write z_2 in terms of z_0 as

$$z_{2} = \frac{z_{0}^{4}}{2(z_{0}^{2}-1)^{2}} \left(\frac{Q_{n}^{2}(z_{0}) - 2G_{n}(z_{0})\widehat{Q}_{n}(z_{0})}{G_{n}(z_{0})Q_{n}(z_{0})} \right)$$

Now substituting z_1 and z_2 into (4.69), we find λ_2 in terms of z_0 as

$$\lambda_2 = \frac{z_0^2}{2(z_0^2 - 1)} \left(\frac{Q_n^2(z_0) - 2G_n(z_0)\widehat{Q}_n(z_0)}{G_n(z_0)Q_n(z_0)} \right) + \frac{z_0}{(z_0^2 - 1)^2}.$$

4.5 Improved bounds and ways to bound

4.5.1 Block-structure estimates

In this section, we apply Theorem 2.1.6 to our problem immediately. Take $\mathbf{X} = S^{-1}H_c^{(2n)}$. First, remark that Theorem 2.1.6 is most informative if $\delta = |\langle A_{12}\mathbf{v}, \mathbf{u}\rangle|$ is very small, which can happen even if $||A_{12}||$ is not small. Moreover, $|\mathrm{Im}\lambda| \leq |b|$ by Lemma 2.1.1, and Davies and Levitin [19] showed that one cannot hope for a result better than $|b| = |\langle A_{12}\mathbf{v}, \mathbf{u}\rangle| = O(1/N)$, and that even this is only possible if $|\mathrm{Re}\lambda|$ is not too close to 0. Proving such a bound depends on using the explicit expressions for \mathbf{u} , \mathbf{v} and A_{12} .

Now, take $A_{11} = H_0 + cI$, $A_{22} = -H_0 - cI$, and the dimension of **X** is N = 2n. Recall that the eigenvalues μ_j of H_0 are as given in (2.37). Then

$$a_1 = \mu_n + c = 2\cos\left(\frac{\pi n}{n+1}\right) + c = -2\cos\left(\frac{\pi}{n+1}\right) + c = -2 + c + \epsilon,$$

and similarly

$$a_2 = \mu_1 + c = 2\cos\left(\frac{\pi}{n+1}\right) = 2 + c - \epsilon,$$

$$d_1 = -2\cos\left(\frac{\pi}{n+1}\right) - c = -2 - c + \epsilon,$$

$$d_2 = -2\cos\left(\frac{\pi n}{n+1}\right) - c = 2 - c - \epsilon.$$

In all these formulae

$$\epsilon := 2 - \mu_1 = 2 - 2\cos\left(\frac{\pi}{n+1}\right) = O(n^{-2}).$$

The inequality of Theorem 2.1.6 reads then

$$-2 + c + \epsilon - \delta \le \operatorname{Re}\lambda \le 2 - c - \epsilon + \delta.$$

As we want to prove (in the "real parts" version of the conjecture as opposed to the possibly harder "modulus" version)

$$-2 + c \le \operatorname{Re}\lambda \le 2 - c,$$

the comparison shows that we actually need a bound $\delta = |\langle A_{12}\mathbf{v}, \mathbf{u} \rangle| \leq \epsilon = O(n^{-2})$ (and for all *n*, not just asymptotically) which is very difficult, by the remark in the beginning of this subsection without additional careful analysis.

4.5.2 Bounds on the ratios of Chebyshev polynomials of the second kind

The behaviour of the ratio for orthogonal polynomials of finite index is still an open question. As mentioned in Section 2.3.4, there exists a function for the limit on the ratio which is given by Simon [66]. However, we will show in this section that Simon's estimate when applied to our particular problem is not enough to prove the conjecture. In this section, we wish to provide a better understanding of the asymptotic behaviour on the finite ratios of the orthogonal polynomials since Simon's estimate is not sufficient.

Recall that the functions $\widetilde{F}_n(\zeta)$, defined by (4.26), equal the ratio of Chebyshev polynomials of the second kind, i.e.

$$\widetilde{F}_n(\zeta) = \frac{U_n(\zeta/2)}{U_{n-1}(\zeta/2)},$$

where $\zeta \in \mathbb{C} \setminus \text{Spec}\left(H_0^{(n-1)}\right)$. In Corollary 4.2.3, we wrote the characteristic equation of the pencil \mathcal{A}_c in terms of \widetilde{F}_n . We start the investigation by applying Theorem 2.3.8 to our particular problem.

Lemma 4.5.1. Let $\sigma, \tau \in \mathbb{C} \setminus [-2, 2]$. Then

$$\lim_{n \to \infty} \left[\widetilde{F}_n(\sigma) \widetilde{F}_n(\tau) \right] = -1 \implies \sigma = -\tau,$$

which implies $\lambda = 0$.

Proof. Using the formula (2.51), we have

$$\lim_{n \to \infty} \left[\widetilde{F}_n(\sigma) \widetilde{F}_n(\tau) \right] = \lim_{n \to \infty} \widetilde{F}_n(\sigma) \lim_{n \to \infty} \widetilde{F}_n(\tau) = -1$$

$$\Leftrightarrow \frac{1}{2}(\sigma + \sqrt{\sigma^2 - 4})\frac{1}{2}(\tau + \sqrt{\tau^2 - 4}) = -1$$

$$\Leftrightarrow (\sigma + \sqrt{\sigma^2 - 4})(\tau + \sqrt{\tau^2 - 4}) = -4.$$
 (4.70)

Note that $(\tau + \sqrt{\tau^2 - 4})(\tau - \sqrt{\tau^2 - 4}) = 4$, then using this in (4.70) yields

$$\begin{split} \sigma + \sqrt{\sigma^2 - 4} &= \sqrt{\tau^2 - 4} - \tau, \\ \Rightarrow & \sigma + \tau = \sqrt{\tau^2 - 4} - \sqrt{\sigma^2 - 4}, \\ \Rightarrow & \sigma^2 + \tau^2 + 2\sigma\tau = \tau^2 - 4 + \sigma^2 - 4 - 2\sqrt{(\sigma^2 - 4)(\tau^2 - 4)}, \\ \Rightarrow & \sigma\tau + 4 = -\sqrt{(\sigma^2 - 4)(\tau^2 - 4)}, \\ \Rightarrow & \sigma^2\tau^2 + 16 + 8\sigma\tau = \sigma^2\tau^2 + 16 - 4(\sigma^2 + \tau^2), \\ \Rightarrow & 4(\sigma^2 + \tau^2) + 8\sigma\tau = 0, \\ \Rightarrow & 4(\sigma + \tau)^2 = 0, \\ \Rightarrow & \sigma = -\tau, \end{split}$$

as required.

Remark 4.5.2. Note that the formula (2.51) does not depend on n. Indeed, Davies and Levitin [19] showed that the spectrum of the pencil A_c converge to the real line as n tends to infinity.

Unfortunately, Proposition 2.3.10 does not give us much information either. Using part (i), we deduce the next two result.

Corollary 4.5.3. For all $\zeta \in \mathbb{C}$, we have

$$\left|\widetilde{F}_{n}(\zeta)\right| \geq \min_{j} \left|\zeta - \mu_{j}^{(n)}\right|.$$
(4.71)

Proof. Taking the absolute value of the equation (2.52) that

$$\frac{1}{\widetilde{F}_n(\zeta)} \left| \leq \sum_{j=1}^n \left| \frac{\alpha_j^{(n)}}{\zeta - \mu_j^{(n)}} \right| \\ \leq \frac{\sum_{j=1}^n \alpha_j^{(n)}}{\min_j |\zeta - \mu_j^{(n)}|} \\ \leq \frac{1}{\min_j |\zeta - \mu_j^{(n)}|},$$

as required.

Corollary 4.5.4. For all $\zeta \in \mathbb{C}$, we have

$$\left|\widetilde{F}_{n}(\zeta)-\zeta\right|\leq\frac{1}{\min_{k}\left|\zeta-\mu_{k}^{(n-1)}\right|}.$$

Proof. We have

$$\widetilde{F}_n(\zeta) = \frac{\prod_{j=1}^n \left(\zeta - \mu_j^{(n)}\right)}{\prod_{k=1}^{n-1} \left(\zeta - \mu_k^{(n-1)}\right)},$$

then doing long divisions of polynomials

$$\widetilde{F}_{n}(\zeta) = \zeta + \sum_{k=1}^{n-1} \frac{a_{k}^{(n-1)}}{\zeta - \mu_{k}^{(n-1)}} \Rightarrow \left| \widetilde{F}_{n}(\zeta) - \zeta \right| \leq \sum_{k=1}^{n-1} \left| \frac{a_{k}^{(n-1)}}{\zeta - \mu_{k}^{(n-1)}} \right|, \\ \leq \frac{\sum_{k=1}^{n-1} \left| a_{k}^{(n-1)} \right|}{\min_{k} \left| \zeta - \mu_{k}^{(n-1)} \right|}, \\ = \frac{1}{\min_{k} \left| \zeta - \mu_{k}^{(n-1)} \right|},$$

as required.

Proposition 2.3.10(ii) tells us if $Im\zeta > 0$, then

$$\frac{\mathrm{Im}\widetilde{F}_n(\zeta)}{|\widetilde{F}_n(\zeta)|} \le \frac{1}{\mathrm{Im}\zeta}$$

Part (iii), which was $\text{Im}\zeta \leq \text{Im}\widetilde{F}_n(\zeta)$, will be shown in Lemma 4.5.5(ii). We see that the results given in Section 2.3.4 do not give anything new for our case.

Now, we want to investigate the finite ratios of the Chebyshev polynomials of the second kind, i.e. the function \widetilde{F}_n , in more detail. The general idea is to obtain better estimates for the function \widetilde{F}_n , not only as the size n of the matrix $H_0^{(n)}$ goes to infinity but also for finite size matrix. We obtain various new inequalities. The following collection of lemmas reveals sharper estimates on some areas for the finite ratio of orthogonal polynomials and/or the moduli of the real part $|\operatorname{Re}(\widetilde{F}_n)|$. The original purpose of these inequalities was to obtain a contradiction to the case that one of $|\sigma|$ and $|\tau|$ can be greater than two. We have obtained many inequalities, however we failed to obtain a contradiction. The main problem is that we cannot introduce the condition $\operatorname{Im}(\sigma) = \operatorname{Im}(\tau)$ easily. Nonetheless, we still believe that they are new and useful bounds. The results up to the end of this subsection are joint with Sabine Bögli (unpublished).

Before stating the results, first recall from the basic properties of complex numbers that

$$\left|\widetilde{F}_{n}(\zeta)\right| = \left|\zeta - \frac{1}{\widetilde{F}_{n-1}(\zeta)}\right| \ge |\zeta| - \frac{1}{\left|\widetilde{F}_{n-1}(\zeta)\right|}.$$
(4.72)

We now list several bounds related with the function \widetilde{F}_n . In all results in this section, we always assume that $\zeta \in \mathbb{C} \setminus \text{Spec} \left(H_0^{(n-1)} \right)$.

Lemma 4.5.5.

- (i) If $\operatorname{Im} \zeta > 0$, then $\operatorname{Im} \widetilde{F}_n(\zeta) > \operatorname{Im} \zeta > 0$ for all $n \ge 2$.
- (ii) If $|\zeta| > 2$ and $\operatorname{Re}\zeta > 0$, then $0 < \operatorname{Re}\widetilde{F}_n(\zeta) < \operatorname{Re}\zeta$ for all $n \ge 2$.
- (iii) If $|\zeta| > 2$ and $\operatorname{Re}\zeta < 0$, then $\operatorname{Re}\widetilde{F}_n(\zeta) < \operatorname{Re}\zeta < 0$ for all $n \ge 2$.

Proof.

(i) Let $\text{Im}\zeta > 0$. Then we have for n = 2 that

$$\mathrm{Im}\widetilde{F}_2(\zeta) = \mathrm{Im}\zeta + \frac{\mathrm{Im}\zeta}{|\zeta|^2} > \mathrm{Im}\zeta,$$

so (i) holds for n = 2. Now assume that (i) is true for some $n \in \mathbb{N}$, i.e. $\mathrm{Im}\widetilde{F}_n(\zeta) > \mathrm{Im}\zeta$. Then

$$\mathrm{Im}\widetilde{F}_{n+1}(\zeta) = \mathrm{Im}\zeta + \frac{\mathrm{Im}F_n(\zeta)}{\left|\widetilde{F}_n(\zeta)\right|^2} > \mathrm{Im}\zeta + \frac{\mathrm{Im}\zeta}{\left|\widetilde{F}_n(\zeta)\right|^2} > \mathrm{Im}\zeta.$$

Then, by induction, (i) holds for all $n \ge 2$.

(ii) Let $|\zeta| > 2$ and $\operatorname{Re}\zeta > 0$. Then we have for n = 2 that

$$\operatorname{Re}\widetilde{F}_{2}(\zeta) = \operatorname{Re}\zeta - \frac{\operatorname{Re}\overline{\zeta}}{|\zeta|^{2}} = \operatorname{Re}\zeta\left(1 - \frac{1}{|\zeta|^{2}}\right) < \operatorname{Re}\zeta$$

so (ii) is true for n = 2. Now assume that the inequality holds for some $n \in \mathbb{N}$ and consider n + 1;

$$\operatorname{Re}\widetilde{F}_{n+1}(\zeta) = \operatorname{Re}\zeta - \frac{\operatorname{Re}F_n(\zeta)}{\left|\widetilde{F}_n(\zeta)\right|^2} < \operatorname{Re}\zeta,$$

since $0 < \operatorname{Re}\widetilde{F}_n(\zeta)$. Thus, by induction, (ii) holds for all $n \in \mathbb{N}$.

(iii) The proof follows in a similar way to part (ii).

The next result gives a bound on the moduli of the function $\widetilde{F}_n(\zeta)$ without assuming any condition on ζ .

Lemma 4.5.6. For all $n \in \mathbb{N}$ and all $\zeta \in \mathbb{C} \setminus \text{Spec} \left(H_0^{(n)}\right)$, the function $\widetilde{F}_n(\zeta)$ satisfies

$$\left|\widetilde{F}_{n}(\zeta)\right| \geq \widetilde{F}_{n}(|\zeta|).$$
 (4.73)

Proof. For n = 1, (4.73) is trivial. For n = 2, we have

$$\left|\widetilde{F}_{2}(\zeta)\right| = \left|\zeta - \frac{1}{\widetilde{F}_{1}(\zeta)}\right| = \left|\zeta - \frac{1}{\zeta}\right| \ge |\zeta| - \frac{1}{|\zeta|} = \widetilde{F}_{2}(|\zeta|),$$

so (4.73) holds for n = 2. Now assume (4.73) holds for some $n \in \mathbb{N}$ and consider n + 1;

$$\left|\widetilde{F}_{n+1}(\zeta)\right| = \left|\zeta - \frac{1}{\widetilde{F}_n(\zeta)}\right| \ge |\zeta| - \frac{1}{\left|\widetilde{F}_n(\zeta)\right|}$$
$$\ge |\zeta| - \frac{1}{\widetilde{F}_n(|\zeta|)}$$
$$= \widetilde{F}_{n+1}(|\zeta|),$$

i.e. (4.73) holds with n replaced by n + 1. By induction (4.73) holds for all $n \in \mathbb{N}$.

The next result gives monotonicity of the function $\widetilde{F}_n(\zeta)$ on the real line.

Lemma 4.5.7. For every $n \in \mathbb{N}$, $\widetilde{F}_n(\zeta)$ is a strictly increasing function on each real interval $(\mu_{j-1}^{(n-1)}, \mu_j^{(n-1)})$, $j = 1, \ldots, n$, where we set $\mu_0^{(n-1)} := -\infty$ and $\mu_n^{(n-1)} := +\infty$.

Proof. First, it can be seen from the definition of \widetilde{F}_n that the zeros of $\widetilde{F}_n(\zeta)$ occur at the zeros of $F_n(\zeta)$, which are $\zeta = \mu_j^{(n)}$, and $\widetilde{F}_n(\zeta)$ is a continuous function except at the zeros of $\widetilde{F}_{n-1}(\zeta)$, which are $\zeta = \mu_k^{(n-1)}$.

Now, it is easy to see that for n = 1, $\widetilde{F}_1(\zeta) = \zeta$ is strictly increasing on $(-\infty, \infty)$. For n = 2,

$$\widetilde{F}_2(\zeta) = \zeta - \frac{1}{\zeta},$$

which is strictly increasing as z increases on $(-\infty, 0)$ and $(0, \infty)$, and blow-up occurs when $\widetilde{F}_1(\zeta) = \zeta = 0$. Suppose that for $n \in \mathbb{N}$, $\widetilde{F}_n(\zeta)$ is strictly increasing function except at the zeros of F_{n-1} , then $-1/\widetilde{F}_n(\zeta)$ is strictly increasing function except at the zeros of $F_n(\zeta)$. Using $\widetilde{F}_{n+1}(\zeta) = \zeta - 1/\widetilde{F}_n(\zeta)$, we have that $\widetilde{F}_{n+1}(\zeta)$ is strictly increasing function except at the zeros of $F_n(\zeta)$. Then the result follows.

We show in the next result that, for some values of $|\zeta|$, we can actually calculate the precise values of the function $\widetilde{F}_n(|\zeta|)$.

Lemma 4.5.8. For each $n \in \mathbb{N}$, we have

$$\widetilde{F}_n(2) = 1 + \frac{1}{n}.$$
 (4.74)

Proof. For n = 1, we have $\widetilde{F}_1(2) = 2$ and for n = 2, we have $\widetilde{F}_2(2) = 2-1/2 = 3/2 = 1+1/2$ so (4.74) holds. Suppose (4.74) holds for some $n \in \mathbb{N}$. Then

$$\widetilde{F}_{n+1}(2) = 2 - \frac{1}{\widetilde{F}_n(2)} = 2 - \frac{1}{1 + \frac{1}{n}} = 2 - \frac{n}{n+1} = \frac{n+2}{n+1} = 1 + \frac{1}{n+1},$$

i.e. (4.74) holds with n replaced by n + 1. By induction (4.74) holds for all $n \in \mathbb{N}$.

Now, using Lemma 4.5.6, Lemma 4.5.7 and Lemma 4.5.8, we deduce two bounds on the moduli of $\widetilde{F}_n(\zeta)$.

Corollary 4.5.9. We have for all $n \in \mathbb{N}$ and all $|\zeta| > 2$ that

$$\left|\widetilde{F}_{n}(\zeta)\right| > 1 + \frac{1}{n}.$$
(4.75)

Proof. It immediately follows from Lemma 4.5.6, Lemma 4.5.7 and Lemma 4.5.8 that

$$\left|\widetilde{F}_{n}(\zeta)\right| \geq \widetilde{F}_{n}(|\zeta|) > \widetilde{F}_{n}(2) = 1 + \frac{1}{n}.$$

Lemma 4.5.10. For all $n \in \mathbb{N}$ and $|\zeta| > 2$, we have

$$\left|\widetilde{F}_{n}(\zeta)\right| > |\zeta| - 1 + \frac{1}{n}.$$
(4.76)

Proof. Using (4.72) and Corollary 4.5.9, we have

$$\begin{aligned} \left| \widetilde{F}_n(\zeta) \right| &\geq |\zeta| - \frac{1}{\left| \widetilde{F}_{n-1}(\zeta) \right|} \\ &> |\zeta| - \frac{1}{\widetilde{F}_{n-1}(2)} \\ &= |\zeta| - \frac{1}{1 + \frac{1}{n-1}} \\ &= |\zeta| - \frac{n-1}{n} \\ &= |\zeta| - 1 + \frac{1}{n}, \end{aligned}$$

as required.

Using Lemma 4.5.10, we deduce an enclosure for the moduli of $\widetilde{F}_n(\zeta)$.

Lemma 4.5.11. For all $n \in \mathbb{N}$ and $|\zeta| > 2$, we have

$$\left| |\widetilde{F}_{n}(\zeta)| - |\zeta| \right| < \frac{1}{|\zeta| - 1 + \frac{1}{n-1}}.$$
 (4.77)

Proof. Using equation (4.72) and Lemma 4.5.10 together, we have for $|\zeta| > 2$:

$$\left|\widetilde{F}_{n}(\zeta)\right| \ge |\zeta| - \frac{1}{\left|\widetilde{F}_{n-1}(\zeta)\right|} > |\zeta| - \frac{1}{|\zeta| - 1 + \frac{1}{n-1}}.$$
(4.78)

On the other hand, we have

$$\begin{aligned} |\widetilde{F}_{n}(\zeta)| - |\zeta| &= \left| \zeta - \frac{1}{\widetilde{F}_{n-1}(\zeta)} \right| - |\zeta| \\ &\leq |\zeta| + \left| \frac{1}{\widetilde{F}_{n-1}(\zeta)} \right| - |\zeta| \\ &= \frac{1}{\left| \widetilde{F}_{n-1}(\zeta) \right|} \\ &< \frac{1}{|\zeta| - 1 + \frac{1}{n-1}}, \end{aligned}$$
(4.79)

where we used Lemma 4.5.10 to obtain the last inequality. Combining (4.78) and (4.79) we arrive at (4.77). $\hfill\square$

Next, we prove an enclosure for the imaginary part of $\widetilde{F}_n(\zeta)$.

Lemma 4.5.12. Assume that $\text{Im}\zeta > 0$. Then we have for all $n \in \mathbb{N}$ and $\zeta \in \mathbb{C}$ that

$$\operatorname{Im}\zeta \le \operatorname{Im}\left(\widetilde{F}_n(\zeta)\right) < \operatorname{Im}\zeta + 1 \tag{4.80}$$

with equality only for n = 1.

Proof. For the lower bound, we showed in Lemma 4.5.5(i) that $\text{Im}\zeta < \text{Im}\left(\widetilde{F}_n(\zeta)\right)$ for all $n \ge 2$. In addition, the equality holds clearly only for n = 1. For the upper bound, we have

$$\operatorname{Im}\left(\widetilde{F}_{n}(\zeta)\right) = \operatorname{Im}\zeta - \operatorname{Im}\left(\frac{1}{\widetilde{F}_{n-1}(\zeta)}\right),$$
$$= \operatorname{Im}\zeta + \frac{\operatorname{Im}\left(\widetilde{F}_{n-1}(\zeta)\right)}{\left|\widetilde{F}_{n-1}(\zeta)\right|^{2}}$$
$$\leq \operatorname{Im}\zeta + 1.$$

_	_	
		н
		L
		н

The following easy result gives a bound on the imaginary part of $\widetilde{F}_n(\zeta)$.

Corollary 4.5.13. If $|\zeta| > 2$, then for all $n \in \mathbb{N}$, we have

$$\operatorname{Im}\left(\widetilde{F}_{n}(\zeta)\right) < \operatorname{Im}\zeta + \frac{\operatorname{Im}\left(\widetilde{F}_{n-1}(\zeta)\right)}{|\zeta| - 1 + \frac{1}{n-1}}$$

Proof. Using Lemma 4.5.10 we have

$$\operatorname{Im}\left(\widetilde{F}_{n}(\zeta)\right) = \operatorname{Im}\zeta + \frac{\operatorname{Im}\left(\widetilde{F}_{n-1}(\zeta)\right)}{\left|\widetilde{F}_{n-1}(\zeta)\right|^{2}} \leq \operatorname{Im}\zeta + \frac{\operatorname{Im}\left(\widetilde{F}_{n-1}(\zeta)\right)}{\left|\zeta\right| - 1 + \frac{1}{n-1}}.$$

Now, we prove an enclosure on the real part of $\widetilde{F}_n(\zeta)$.

Lemma 4.5.14. *If* $|\zeta| > 2$ *, then*

$$\left|\operatorname{Re}\left(\widetilde{F}_{n}(\zeta)\right) - \operatorname{Re}\zeta\right| < \left(|\zeta| - 1 + \frac{1}{n-1}\right)^{-1} < 1.$$

Proof. Using Lemma 4.5.10 we have

$$\left|\operatorname{Re}\left(\widetilde{F}_{n}(\zeta)\right) - \operatorname{Re}\zeta\right| = \left|\operatorname{Re}\left(\frac{1}{\widetilde{F}_{n-1}(\zeta)}\right)\right| < \frac{1}{\left|\widetilde{F}_{n-1}(\zeta)\right|} < \left(|\zeta| - 1 + \frac{1}{n-1}\right)^{-1} < 1.$$

Now, we give a bound on the moduli and on the real part of the function $\widetilde{F}_n(\zeta)$ assuming the real part of the parameter ζ is greater than two.

Lemma 4.5.15. *If* $|\text{Re}\zeta| > 2$ *, then*

$$\widetilde{F}_n(\zeta)| > \sqrt{4 + (\mathrm{Im}\zeta)^2} - \frac{n-1}{n} > 1.$$
 (4.81)

Proof. Using equation (4.78) and |Re(z)| > 2, we obtain

$$|\widetilde{F}_n(\zeta)| > \sqrt{4 + (\mathrm{Im}\zeta)^2} - \frac{1}{\sqrt{4 + (\mathrm{Im}\zeta)^2} - 1 + \frac{1}{n-1}} > 1.$$

Let

$$a := \sqrt{4 + (\mathrm{Im}\zeta)^2} \ge 2,$$

 $b(a) := a - \frac{1}{a - 1 + \frac{1}{n - 1}},$

then we have

$$b(2) = 2 - \frac{1}{1 + \frac{1}{n-1}} = 2 - \frac{n-1}{n} = \frac{n+1}{n},$$

$$b'(a) = 1 + \frac{1}{\left(a - 1 + \frac{1}{n-1}\right)^2} > 1,$$

so that using the Taylor series for b(a) at 2, we obtain

$$b(a) = b(2) + b'(2)(a-2) + \dots \ge \frac{n+1}{n} + (a-2) + \dots = a - \frac{n-1}{n} + \dots,$$

i.e. we obtain

$$\left|\widetilde{F}_{n}(\zeta)\right| > b(a) \ge \sqrt{4 + (\mathrm{Im}\zeta)^{2}} - \frac{n-1}{n}.$$

Lemma 4.5	.16. If	$ \text{Re}\zeta $	> 2,	then
-----------	---------	--------------------	------	------

$$\operatorname{Re}\left(\widetilde{F}_{n}(\zeta)\right) > 2 - \left(\sqrt{4 + (\operatorname{Im}\zeta)^{2}} - \frac{n-2}{n-1}\right)^{-1} > 1.$$

Proof. We have

$$\left|\operatorname{Re}\left(\widetilde{F}_{n}(\zeta)\right) - \operatorname{Re}\zeta\right| < \frac{1}{\left|\widetilde{F}_{n-1}(\zeta)\right|}$$

which implies using equation (4.78)

$$-\left(|\zeta| - \frac{1}{|\zeta| - 1 + \frac{1}{n-2}}\right)^{-1} < \operatorname{Re}\left(\widetilde{F}_n(\zeta)\right) - \operatorname{Re}\zeta < \left(|\zeta| - \frac{1}{|\zeta| - 1 + \frac{1}{n-2}}\right)^{-1},$$

which then implies

$$\operatorname{Re}\left(\widetilde{F}_{n}(\zeta)\right) > \operatorname{Re}\zeta - \left(|\zeta| - \frac{1}{|\zeta| - 1 + \frac{1}{n-2}}\right)^{-1}$$
$$> 2 - \left(|\zeta| - \frac{1}{|\zeta| - 1 + \frac{1}{n-2}}\right)^{-1}$$
$$> 2 - \left(\sqrt{4 + (\operatorname{Im}\zeta)^{2}} - \frac{1}{\sqrt{4 + (\operatorname{Im}\zeta)^{2}} - 1 + \frac{1}{n-2}}\right)^{-1} > 1.$$

Let

$$a := \sqrt{4 + (\operatorname{Im}(\zeta))^2},$$

$$d(a) := a - \frac{1}{a - 1 + \frac{1}{n - 2}}$$

then we have

$$d(2) = 2 - \frac{1}{1 + \frac{1}{n-2}} = 2 - \frac{n-2}{n-1} = \frac{n}{n-1},$$

$$d'(a) = 1 + \frac{1}{\left(a - 1 + \frac{1}{n-2}\right)^2} > 1,$$

so that using the Taylor series expansion for d(a) at 2

$$d(a) = d(2) + d'(2)(a-2) + \dots \ge \frac{n}{n-1} + a - 2,$$

which implies

$$-(d(a))^{-1} \ge -\left(\frac{n}{n-1} + a - 2\right)^{-1} = -\left(a - \frac{n-2}{n-1}\right)^{-1},$$

i.e. we have

$$\operatorname{Re}\left(\widetilde{F}_{n}(\zeta)\right) > 2 - \left(\sqrt{4 + (\operatorname{Im}\zeta)^{2}} - \frac{n-2}{n-1}\right)^{-1}$$

4.5.3 Gershgorin-type localisation

4.5.3.1 Application of known results to the pencil problem

Recall that $\mathcal{A}_c = H_c^{(N)} + \lambda S$ is the linear pencil and $\mathbf{A}_c = S^{-1}H_c^{(N)}$ is the non-self-adjoint block matrix, and that $\operatorname{Spec}(\mathcal{A}_c) \equiv \operatorname{Spec}(\mathbf{A}_c)$. We already mentioned the Gershgorin set $\mathcal{G}(\mathbf{A}_c)$ of \mathbf{A}_c in Section 4.2.3. In this section, we apply the well-known Gershgorin-type eigenvalue localisations, mentioned in Section 2.2, to our problem. Note that, in this section, the letters *i*, *j*, *p*, *q* may represent different number sequences in each of the bullet points.

(i) The Brauer set $\mathcal{K}(\mathbf{A}_c)$ consists of seven different Brauer Cassini ovals given by

$$\mathcal{K}_{1,j}(\mathbf{A}_c) = \mathcal{D}(c,\sqrt{2}), \qquad \mathcal{K}_{i,j}(\mathbf{A}_c) = \mathcal{D}(c,2),$$
$$\mathcal{K}_{p,N}(\mathbf{A}_c) = \mathcal{D}(-c,\sqrt{2}), \qquad \mathcal{K}_{p,q}(\mathbf{A}_c) = \mathcal{D}(-c,2),$$
$$\mathcal{K}_{1,N}(\mathbf{A}_c) = \{z \in \mathbb{C} : |z^2 - c^2| \le 1\},$$
$$\mathcal{K}_{i,N}(\mathbf{A}_c) = \mathcal{K}_{1,q}(\mathcal{A}_c) = \{z \in \mathbb{C} : |z^2 - c^2| \le 2\},$$
$$\mathcal{K}_{p,j}(\mathcal{A}_c) = \mathcal{K}_{i,q}(\mathcal{A}_c) = \{z \in \mathbb{C} : |z^2 - c^2| \le 4\},$$

where i, j = 2, ..., n with $i \neq j$, and p, q = n + 1, ..., N - 1 with $p \neq q$. Note that $\mathcal{K}_{1,j}(\mathbf{A}_c) \subset \mathcal{K}_{i,j}(\mathcal{A}_c), \mathcal{K}_{p,N}(\mathbf{A}_c) \subset \mathcal{K}_{p,q}(\mathbf{A}_c), \mathcal{K}_{1,N}(\mathbf{A}_c) \subset \mathcal{K}_{i,N}(\mathbf{A}_c) \subset \mathcal{K}_{p,j}(\mathbf{A}_c)$ and also $\mathcal{K}_{p,j}(\mathbf{A}_c) \subset \mathcal{K}_{i,j}(\mathbf{A}_c) \cup \mathcal{K}_{p,q}(\mathbf{A}_c)$. Then we obtain

$$\operatorname{Spec}(\mathbf{A}_c) \subseteq \mathcal{K}(\mathbf{A}_c) = \mathcal{D}(c,2) \cup \mathcal{D}(-c,2),$$

which is the same region as the Gershgorin set.

(ii) We have from Theorem 2.2.5 that

$$\sum_{i=1}^{N} \frac{r_i(\mathbf{A}_c)}{\max_{j \in J_i} |a_{i,j}|} = 2 + 2(N-2) = 2N - 2 \le \alpha(1+\alpha).$$

However, for any $n \ge 2$ (or $N \ge 4$), we have $\alpha \ge 2$, and so it gives a bigger region than Gershgorin set for our case.

- (iv) The set $\Omega(\mathbf{A}_c)$ given in Theorem 2.2.6 produces a complicated expression and there is a freedom of choice of the set \tilde{J}_i . In addition, it takes a large amount of time to find the sets as *n* increases. Nevertheless, it gives a smaller region than the Gershgorin set for our case. First, we discuss this set in more detail at the end of this section. We then illustrate the set along with its comparison with other sets in Figure 4.7.
- (v) Numerical experiments suggest that the set given in Theorem 2.2.7 gives the same region as the Gershgorin set when c = 0. However, the region gets much bigger as c increases. Therefore, we omit the mathematical argument for this set.
- (vi) When we consider the sets given by Melman [51] and Li et al. [44], which were stated in Theorem 2.2.8 and Theorem 2.2.10 respectively, they both give the same region as the Gershgorin union.
- (vii) Consider the union $\bigcup_{i=1}^{N} S_i(H_c^{(N)}, S)$ given in Theorem 2.2.12. In our case, since $d_{i,j} = 0$ when $i \neq j$, the union yields the same region as the Gershgorin set.
- (viii) The union $\bigcup_{i=1}^{N} \widetilde{S}_i(H_c^{(N)}, S)$ given in Theorem 2.2.13 consists of four different regions that are

$$\widetilde{S}_{1}(H_{c}^{(N)},S) = \left\{ z \in \mathbb{C} : \frac{|z-c|}{\sqrt{1+|z|^{2}}} \le 1 \right\}; \ \widetilde{S}_{j}(H_{c}^{(N)},S) = \left\{ z \in \mathbb{C} : \frac{|z-c|}{\sqrt{1+|z|^{2}}} \le 2 \right\},\\ \widetilde{S}_{k}(H_{c}^{(N)},S) = \left\{ z \in \mathbb{C} : \frac{|z+c|}{\sqrt{1+|z|^{2}}} \le 1 \right\}; \ \widetilde{S}_{N}(H_{c}^{(N)},S) = \left\{ z \in \mathbb{C} : \frac{|z+c|}{\sqrt{1+|z|^{2}}} \le 2 \right\},$$

where j = 2, ..., n and k = n + 1, ..., N - 1. Note that $\widetilde{S}_1\left(H_c^{(N)}, S\right) \subset \widetilde{S}_j\left(H_c^{(N)}, S\right)$ and $\widetilde{S}_N\left(H_c^{(N)}, S\right) \subset \widetilde{S}_k\left(H_c^{(N)}, S\right)$. Therefore we obtain

Spec
$$(H_c^{(N)} - \lambda S) \subset \bigcup_{i=1}^N \widetilde{\mathcal{S}}_i (H_c^{(N)}, S) \equiv \widetilde{\mathcal{S}}_j (H_c^{(N)}, S) \cup \widetilde{\mathcal{S}}_k (H_c^{(N)}, S)$$

which covers the whole complex plane when $|c| \le 2$ and excludes a region when |c| > 2.

(ix) Applying Gershgorin set for block matrices (b.m.) (Theorem 2.2.14) with $A_{11} = H_c$, $A_{22} = -H_c$ and $||A_{12}|| = 1$ gives

$$\operatorname{Spec}(\mathbf{A}_c) \subset \{\zeta \in \mathbb{C} : \operatorname{dist}(\zeta, \operatorname{Spec}(H_c) \cup \operatorname{Spec}(-H_c)) \leq 1\},\$$

which means that in the complex plane, there are n disks with centre $(\mu_j + c)$ and radius one; n disks with centre $(\mu_j - c)$ and radius one; and eigenvalues that lie within the union of 2n disks. In addition, we get

$$\operatorname{Spec}(\mathbf{A}_c) \cap \mathbb{R} \subset [\mu_n - c, \mu_1 + c] \subset [-2 - c, 2 + c],$$

which was already known. Nevertheless, this result also supports that $Im\lambda \leq 1$ for all $n \in \mathbb{N}$.

(x) Applying Brauer set for b.m. (Theorem 2.2.15) to our problem, we obtain

$$\operatorname{Spec}(\mathbf{A}_{\mathbf{c}}) \subset \{\zeta \in \mathbb{C} : \operatorname{dist}(\zeta, \operatorname{Spec}(H_c)) \operatorname{dist}(\zeta, \operatorname{Spec}(-H_c)) \leq 1\}$$

As discussed above, some results do not give better bounds than Gershgorin Theorem when applied to the pencil A_c problem. Namely, Fan and Hoffmann (Theorem 2.2.5) is worse for $n \ge 2$, Theorem 2.2.7 produces a bigger region when $c \ne 0$, and weighted Stewart-Sun (Theorem 2.2.13) gives nothing. On the other hand, some results do not improve classical Gershgorin, however they produce the same region as Gershgorin theorem. Namely, these are classical Brauer, Theorem 2.2.8 and Theorem 2.2.10, Unweighted Stewart-Sun (Theorem 2.2.12).

In addition to the ones which were mentioned in Section 2.2, there are many other Gershgorin-type regions which were obtained, for instance, by Ostrowski [79], Fan and Hoffman [79], Dashnic and Zusmanovich [18], Nakatsukasa [58], Li et al. [44], Anna et al. [17], Herzog and Schmoeger [29]. We, however, exclude these as they do not give a better spectral enclosure than the Gershgorin union for our particular case.

We note that we do not attempt to find the minimal Gershgorin set of A_c as the algorithms are suitable for small or medium size matrices. This result was reviewed due to its importance.

In our case, there are three results which improve classical Gershgorin Theorem. Gershgorin set for b.m. given by Tretter in Theorem 2.2.14, Brauer set for b.m. given by Freingold-Varga in Theorem 2.2.15, and the set given by Melman in Theorem 2.2.6. We first turn our attention to the set $\Omega(\mathbf{A}_c)$ given by Melman in Theorem 2.2.6. We will compare these sets at the end of this subsection.

We choose $\widetilde{J}_i \equiv J_i$ to obtain the smallest region. Let

$$\Omega(\mathbf{A}_c) = \widetilde{\Omega}(\mathbf{A}_c) \cup \widehat{\Omega}(\mathbf{A}_c)$$

where

$$\widetilde{\Omega}(\mathbf{A}_c) := \bigcup_{i=1}^n \widetilde{\Omega}_i(\mathbf{A}_c) = \bigcup_{i=1}^n \left\{ \bigcap_{j \in \widetilde{J}_i} \Omega_{i,j}(\mathbf{A}_c) \right\},$$
$$\widehat{\Omega}(\mathbf{A}_c) := \bigcup_{k=n+1}^{2n} \widehat{\Omega}_k(\mathbf{A}_c) = \bigcup_{k=n+1}^{2n} \left\{ \bigcap_{j \in \widetilde{J}_k} \Omega_{k,j}(\mathbf{A}_c) \right\}.$$

There are 2n sets in the union $\Omega(\mathbf{A}_c)$ and we divided the union into two; those whose centre are at c called $\widetilde{\Omega}(\mathbf{A}_c)$ and those whose centre at -c called $\widehat{\Omega}(\mathbf{A}_c)$. In other words, $\widetilde{\Omega}(\mathbf{A}_c)$ and $\widehat{\Omega}(\mathbf{A}_c)$ are symmetric with respect to imaginary axis. Note that each set $\widetilde{\Omega}_i(\mathbf{A}_c)$ consists of intersection of more than one regions, however we omit the mathematical expression of these sets for clarity and illustrate them instead. Numerical experiments suggest that when $n \ge 4$, the union $\widetilde{\Omega}(\mathbf{A}_c)$ contains four possible geometric shapes and the union $\widehat{\Omega}(\mathbf{A}_c)$ contains their symmetric regions with respect to the Im λ -axis. We illustrate these regions in Figure 4.6, without specifying their centre. Nevertheless, when we take the union of these sets, we then see that

$$\bigcup_{i\in J_3}\widetilde{\Omega}_i(\mathbf{A}_c)\subset\widetilde{\Omega}_3(\mathbf{A}_c),$$

i.e. $\widetilde{\Omega}(\mathbf{A}_c) \equiv \widetilde{\Omega}_3(\mathbf{A}_c)$. Similarly, we have $\widehat{\Omega}(\mathbf{A}_c) \equiv \widehat{\Omega}_{2n-3}(\mathbf{A}_c)$.

Remark 4.5.17. The illustration of the set $\widetilde{\Omega}_n(\mathbf{A}_c)$ in Figure 4.6 may change as c changes, however $\widetilde{\Omega}_n(\mathbf{A}_c) \subset \widetilde{\Omega}_3(\mathbf{A}_c)$ will always hold.

Looking at $\widetilde{\Omega}_3(\mathbf{A}_c)$ and $\widehat{\Omega}_{2n-3}(\mathbf{A}_c)$, we see that

$$\widetilde{\Omega}_3(\mathbf{A}_c) = \bigcap_{j \in \widetilde{J}_3} \Omega_{3,j}(\mathbf{A}_c) \equiv \{ z \in \mathbb{C} : |z - c|^2 \le 2|z - c| \text{ and } |z^2 - c^2| \le 2|z + c| \}$$



Figure 4.6: Without specifying their coordinates, four possible regions are illustrated from the union $\widetilde{\Omega}(\mathbf{A}_c)$. Dark orange shaded regions from left to right: $\widetilde{\Omega}_1(\mathbf{A}_c)$, $\widetilde{\Omega}_2(\mathbf{A}_c)$, $\widetilde{\Omega}_3(\mathbf{A}_c)$ and $\widetilde{\Omega}_n(\mathbf{A}_c)$. Other coloured regions represent their corresponding intersected sets.

$$\cap \{ z \in \mathbb{C} : |(z-c)^2 - 1| \le 1 + |z-c| \}$$

$$\equiv \{ z \in \mathbb{C} : |(z-c)^2 - 1| \le 1 + |z-c| \},$$

and similarly

$$\widehat{\Omega}_{2n-3}(\mathbf{A}_c) = \bigcap_{j \in \widetilde{J}_{2n-3}} \Omega_{2n-3,j}(\mathbf{A}_c) \equiv \{ z \in \mathbb{C} : |(z+c)^2 - 1| \le 1 + |z+c| \}.$$

Hence

$$\operatorname{Spec}(\mathbf{A}_c) \subset \Omega(\mathbf{A}_c) = \widetilde{\Omega}_3(\mathbf{A}_c) \cup \widehat{\Omega}_{2n-3}(\mathbf{A}_c).$$

We now compare these there sets in Figure 4.7. It seems that Gershgorin set for b.m is contained in Brauer set for b.m., and they both support that $Im(\lambda) \leq 1$. On the one hand Tretter's result improves away on $Re(\lambda)$, on the other hand Freingold-Varga's result is better for small *n*, however it seems that the set is asymptotically the same as Gershgorin set for b.m. for large *n*. Moreover, the set $\Omega(\mathbf{A}_c)$ gives the same bound on $Re(\lambda)$ as classical Gershgorin and very close bound to the Gershgorin for b.m. on $Im(\lambda)$. However, when *c* is close to 2, it gives a better bound on $Im(\lambda)$ for the eigenvalues whose real parts are close to zero.

4.5.3.2 Gershgorin-type region for the two-parameter eigenvalue problem

Now, let us apply Corollary 2.2.16 to our problem. Let $A = H_0^{(n)}$, $D = H_0^{(m)}$ and $(C)_{n,1} = \kappa$, then we have

$$\operatorname{Spec}_p(\mathbf{M}) \subset \mathcal{M}\left(H_0^{(n)}, H_0^{(m)}\right)$$



Figure 4.7: Spec(\mathbf{A}_c) (red dots), the classical Gershgorin disks $\mathcal{G}(\mathbf{A}_c)$ (black dotted circles), Gershgorin sets for b.m. (brown shaded region), Brauer set for b.m. (blue shaded region) and Melman's set $\Omega(\mathbf{A}_c)$ (orange shaded region) are drawn for the non-self-adjoint matrix \mathbf{A}_c . Top-left: c = 0.3 and n = 7. Top-right: $c = \sqrt{2}$ and n = 8. Bottom-left: c = 1.85 and n = 25.

where

$$\mathcal{M}\left(H_0^{(n)}, H_0^{(m)}\right) = \left\{ (\alpha, \beta) \in \mathbb{C}^2 : \left(\inf_{a \in \operatorname{Spec}\left(H_0^{(n)}\right)} |\alpha - a| \right) \left(\inf_{d \in \operatorname{Spec}\left(H_0^{(m)}\right)} |\beta - d| \right) \le \|C\|^2 \right\}$$

which can be re-written as

$$\mathcal{M}\left(H_0^{(n)}, H_0^{(m)}\right) = \bigcup_{i=1}^n \bigcup_{j=1}^m \left\{ (\alpha, \beta) \in \mathbb{C}^2 : \left| \alpha - \mu_i^{(n)} \right| \left| \beta - \mu_j^{(m)} \right| \le \kappa^2 \right\},$$

where $\mu_i^{(n)} \in \text{Spec}(H_0^{(n)})$. Illustration of Gershgorin-type localisation of the two-parameter eigenvalue problem in the $(\text{Re}(\alpha), \text{Re}(\beta))$ -plane can be seen for n = m in Figure 4.8 and for $n \neq m$ in Figure 4.9.

4.5.3.3 Gershgorin circles under the mapping z + 1/z

In this section, we want to apply the Gershgorin Theorem to the problem (4.22) to obtain equivalent statements in terms of z and w values and we illustrate the region in the complex plane.

Let

$$u_j = 2\cos\theta_j, \qquad \theta_j = \pi j/(n+1), \quad j = 1, \dots, n.$$
 (4.82)



Figure 4.8: $\widetilde{\text{Spec}}(\mathbf{M})$ with $A = D = H_0^{(n)}$ and $\kappa = 1$, and blue shaded region is the union $\mathcal{M}\left(H_0^{(n)}, H_0^{(n)}\right)$. Left: n = 4. Right: n = 5.



Figure 4.9: $\widetilde{\text{Spec}}(\mathbf{M})$ with $A = H_0^{(3)}$ and $D = H_0^{(7)}$ and blue shaded region is the union $\mathcal{M}\left(H_0^{(3)}, H_0^{(7)}\right)$. Left: $\kappa = 0.3$. Right: $\kappa = \sqrt{3}$.

Recall that $\psi_{j,k} = \sqrt{\frac{2}{n+1}} \sin\left(\frac{\pi jk}{n+1}\right)$. Define

$$\widehat{r}_j := |\psi_{j,n}| \sum_{k=1}^n |\psi_{k,n}|.$$
(4.83)

Now, consider the problem (4.22):

$$\widetilde{\mathbf{A}}_{c}\begin{pmatrix}\mathbf{u}\\\mathbf{v}\end{pmatrix} = \begin{pmatrix} \mu_{1}+c & \psi_{1,n}\psi_{n,n} & \cdots & \psi_{1,n}\psi_{1,n} \\ & \ddots & \vdots & & \vdots \\ & & \mu_{n}+c & \psi_{n,n}\psi_{n,n} & \cdots & \psi_{n,n}\psi_{1,n} \\ & & & -\psi_{1,n}\psi_{n,n} & \cdots & -\psi_{1,n}\psi_{1,n} & -(\mu_{1}+c) & & \\ & \vdots & & \vdots & & \ddots & \\ & & & -\psi_{n,n}\psi_{n,n} & \cdots & -\psi_{n,n}\psi_{1,n} & & -(\mu_{n}+c) \end{pmatrix} \begin{pmatrix} u_{1} \\ \vdots \\ u_{n} \\ v_{1} \\ \vdots \\ v_{n} \end{pmatrix} = \lambda \begin{pmatrix} \mathbf{u} \\ \mathbf{v} \end{pmatrix}.$$

Note that μ_j are symmetric. Then applying Gershgorin Theorem to $\widetilde{\mathbf{A}}_c$ gives that

Spec
$$\left(\widetilde{\mathbf{A}}_{c}\right) \subset \mathcal{G}\left(\widetilde{\mathbf{A}}_{c}\right) = \bigcup_{j=1}^{n} \mathcal{G}_{j}\left(\widetilde{\mathbf{A}}_{c}\right),$$

where

$$\mathcal{G}_{j}\left(\widetilde{\mathbf{A}}_{c}\right) = \mathcal{D}\left(\mu_{j} + c, \widehat{r}_{j}\right) \cup \mathcal{D}\left(-\mu_{j} - c, \widehat{r}_{j}\right).$$

To understand the geometry of $\mathcal{G}\left(\widetilde{\mathbf{A}}_{c}\right)$, we illustrate each set in the union $\bigcup_{j} \mathcal{D}\left(\mu_{j}+c,\widehat{r}_{j}\right)$ in Figure 4.10, without specifying its coordinates. Again, the union $\bigcup_{j} \mathcal{D}\left(\mu_{j}-c,\widehat{r}_{j}\right)$ and $\bigcup_{j} \mathcal{D}\left(\mu_{j}+c,\widehat{r}_{j}\right)$ are symmetric with respect to the imaginary axis. We then compare $\mathcal{G}\left(\widetilde{\mathbf{A}}_{c}\right)$ and $\mathcal{G}\left(\mathbf{A}_{c}\right)$ in Figure 4.11.



Figure 4.10: Without specifying the coordinates, each set in the union $\cup_j \mathcal{D}(\mu_j + c, \hat{r}_j)$ are drawn as blue dashed lines in $(\text{Re}(\lambda), \text{Im}(\lambda))$ -plane. Left: for n = 5. Right: for n = 25.



Figure 4.11: Spec(\mathbf{A}_c) (red dots), the union $\mathcal{G}\left(\widetilde{\mathbf{A}}_c\right)$ (blue shaded region) and the Gerhsgorin set $\mathcal{G}(\mathbf{A}_c)$ (black dotted circles) are drawn in (Re λ , Im λ)-plane. Left: for n = 5 and c = 0.4. Right: for n = 25 and c = 1.6.

In order to plot the Gershgorin set $\mathcal{G}(\widetilde{\mathbf{A}}_c)$ using the mapping $\lambda \mapsto z$ and $\lambda \mapsto w$ with (4.8), we need the following auxiliary result.

Lemma 4.5.18. Let $\zeta \in \mathbb{C} \setminus \{0\}$, and μ_j and θ_j are as in (4.82). Then

$$\left|\zeta + \frac{1}{\zeta} - \mu_j\right| = \frac{\left|\zeta - e^{i\theta_j}\right| \left|\zeta - e^{-i\theta_j}\right|}{\left|\zeta\right|}, \qquad \left|\zeta + \frac{1}{\zeta} + \mu_j\right| = \frac{\left|\zeta + e^{i\theta_j}\right| \left|\zeta + e^{-i\theta_j}\right|}{\left|\zeta\right|}.$$

Proof. We prove the first of these equations, the second is similar. We have

$$\begin{split} \left| \zeta + \frac{1}{\zeta} - \mu_j \right| &= \frac{|\zeta^2 - \mu_j \zeta + 1|}{|\zeta|} \\ &= \frac{|\zeta^2 - 2\zeta \cos \theta_j + \cos^2 \theta_j + \sin^2 \theta_j|}{|\zeta|} \\ &= \frac{|(\zeta - \cos \theta_j)^2 - i^2 \sin^2 \theta_j|}{|\zeta|} \\ &= \frac{|\zeta - \cos \theta_j - i \sin \theta_j| |\zeta - \cos \theta_j + i \sin \theta_j|}{|\zeta|} \\ &= \frac{|\zeta - e^{i\theta_j}| |\zeta - e^{-i\theta_j}|}{|\zeta|}. \end{split}$$

Lemma 4.5.19. Let λ , z, w be related by (4.8). Then λ is an eigenvalue of $\text{Spec}(\mathcal{A}_c)$ if and only if z and w lies in the union

$$\mathcal{G}_{\zeta} := \bigcup_{j=1}^{n} \left\{ \zeta \in \mathbb{C} \setminus \{0\} : \frac{|\zeta - e^{i\theta_j}| |\zeta - e^{-i\theta_j}|}{|\zeta|} \le \widehat{r}_j \right\},\tag{4.84}$$

where θ_j and \hat{r}_j are as in (4.82) and (4.83) respectively.

Proof. We have

$$|\lambda - \mu_j - c| = |\sigma - \mu_j| = \left|z + \frac{1}{z} - \mu_j\right|, \qquad |\lambda + \mu_j + c| = |\tau + \mu_j| = \left|w + \frac{1}{w} + \mu_j\right|.$$

Applying Gershgorin Theorem to \tilde{A}_c and then using Lemma 4.5.18 gives the union

$$\left\{\zeta \in \mathbb{C} \setminus \{0\}: \ \frac{|\zeta - e^{i\theta_j}||\zeta - e^{-i\theta_j}|}{|\zeta|} \leq \widehat{r}_j\right\} \cup \left\{\zeta \in \mathbb{C} \setminus \{0\}: \ \frac{|\zeta + e^{i\theta_j}||\zeta + e^{-i\theta_j}|}{|\zeta|} \leq \widehat{r}_j\right\}.$$

Note that there is a symmetry: $|\zeta - e^{i\theta_j}| = |\zeta + e^{i\theta_{n+1-j}}|$ and $|\psi_{j,k}| = |\psi_{n+1-j,k}|$. Therefore, when taking the union for all j = 1, ..., n, the union reduces to \mathcal{G}_{ζ} due to the symmetries. We then have that $\lambda \in \text{Spec}(\mathcal{A}_c)$ iff $z, w \in \mathcal{G}_{\zeta}$ where \mathcal{G}_{ζ} is as given in (4.84).

We illustrate each set in the union \mathcal{G}_{ζ} in Figure 4.12. Recall that if $\lambda \notin \mathbb{R}$, then |z|, |w| > 1. Note that the union consists of n sets and the boundary of each set is drawn with blue dashed lines. It can be seen that both the inner boundary and the outer boundary of the union \mathcal{G}_{ζ} get smoother as n increases. In Figure 4.13, we compare the region \mathcal{G}_{ζ} with the region $\mathcal{Z}_1 \cup \mathcal{Z}_2$, which was defined in Theorem 1.3.4(iv).



Figure 4.12: The union \mathcal{G}_{ζ} (blue shaded region) and the boundary of each set in the union \mathcal{G}_{ζ} (blue dashed lines) are drawn in the complex plane. Left: for n = 5. Right: for n = 10.



Figure 4.13: Superimposition of \mathcal{G}_{ζ} (blue shaded area), \mathcal{Z}_1 (orange shaded area), \mathcal{Z}_2 (green shaded area), z values (red dots) and w values (blue dots). Left: for n = 5. Right: for n = 10.

Chapter 5

Numerics and heuristics

In this chapter, we present some heuristics related to the non-real eigenvalues of a twoparameter eigenvalue problem and the pencil problem. There is a complicated interplay between the eigenvalues of the pencil A_c . We start by describing the dynamics of the eigenvalues of A_c as the parameter c changes in Section 5.1. In Section 5.2 we give illustrative examples related to a two-parameter eigenvalue problem. We examine the non-real pair-eigenvalues in different coordinate systems and we illustrate some interesting patterns in their location and behaviour. In Section 5.3, we conclude the chapter with a brief discussion of the double eigenvalues of A_c . We propose several conjectures about the location and the quantity of double eigenvalues based on numerical experiments.

5.1 Dynamics of the eigenvalues of the pencil A_c

When c = 0, all the eigenvalues of A_c are non-real and Davies and Levitin [19] were able to determine the asymptotics of complex eigenvalues of A_c as $n \to \infty$. However, as cchanges, tracing the behaviour of the eigenvalues of A_c remains difficult. In this section, we describe the dynamics of the eigenvalues of A_c as c increases from 0 to 2. Since the spectrum $\text{Spec}(A_c)$ is invariant under reflections about the real and imaginary axes, we are able to describe the behaviour that occurs in the right half plane, i.e. $\text{Re}(\lambda) > 0$. First, we recall the collision types from Figure 3.12; two real eigenvalues collide and produce a complex conjugate pair, called Type-A, and a complex conjugate pair collides and produces real eigenvalues, called Type-B.

Numerical experiments demonstrate that as c goes from 0 to 2, for arbitrary n each complex conjugate pair approach to each other along a curve, as opposed to a straight line, until they collide for the first time on the real axis. We refer to this as Level-1 which consists only of Type-B collisions that occur for the first time. All collisions in Level-1 occur

in order, i.e. the pair with the greatest real part collide first and the pair who has the least real part collide last.

After the first collisions occur, each pair travels along the real axis, where one heads to the right and the other to the left. If two real eigenvalues meet, they become a double eigenvalue at the point at which they meet. Subsequently they split and escape from the real line as a conjugate pair. In the complex plane, they make a small jump (in the sense that the eigenvalues travels a half-loop) in directions opposite to the origin. Then this jump is followed by another collision on the real line. This is what we refer to as Level-2, where eigenvalues have two collisions in total; first Type-A collisions occur and each complex conjugate pair makes a small jump in the complex plane and then Type-B collisions occur. The process in Level-2 is repeated in the following levels.

These collisions and jumps in each level are illustrated in Table 5.1 for n = 7. For simplicity, we plot jumps that occur only in the first quadrant. There are some collisions which takes place at the origin, i.e. $\lambda = 0$. When Type-A collisions occur at the origin, they then escape to the complex plane as a purely imaginary pair and they never leave the imaginary axis. They come back and collide on the real line at the origin again. Note that in Table 5.1 we do not show the jumps that occur at the origin in any level as the eigenvalues move along the imaginary axis only. However, the final collision always occurs at the origin.

We observe the following generic behaviour from numerics:

- (i) Each pair always collide, separately and independently of other pairs.
- (ii) All jumps occur in the direction opposite to the origin.
- (iii) At each Level-j, j = 2, ..., n 1, the eigenvalue whose real part is closest to the origin reaches higher point while jumping in the complex plane.
- (iv) At Level-1, the collisions are monotonic in c. This, however, is not true for any other levels. The collisions occur without order, that is for sufficiently large n, some eigenvalues at Level-j may be produced earlier than those which are still in Level-(j 1) or Level-(j 2). Nevertheless, collisions and jumps occur with sufficient speed such that the eigenvalues are unable to overtake each other on the real line.

5.2 Heuristics for the non-real pair-eigenvalues of ${f M}$

In this section we discuss some heuristic results and conjectures for the behaviour of the non-real pair-eigenvalues of a two-parameter eigenvalue problem. The pair-eigenvalues



Table 5.1: For n = 7, blue dots represents the dynamics of the non-real eigenvalues of \mathcal{A}_c in the first quadrant of the complex plane ($\operatorname{Re}(\lambda), \operatorname{Im}(\lambda)$), as c increases from 0 to 2.

provide a better perspective in the $(c, \operatorname{Re}(\lambda))$ -plane than they do in the $(\operatorname{Re}(\alpha), \operatorname{Re}(\beta))$ plane. We therefore present them in $(c, \operatorname{Re}(\lambda))$ coordinates. Recall the substitution (3.17),

$$\alpha - \beta = 2\lambda; \qquad \alpha + \beta = -2c. \tag{5.1}$$

This substitution allows us to change the coordinate system from $(c, \text{Re}(\lambda))$ to (α, β) . Therefore, we can equivalently write all results in Section 3.2 in the $(c, \text{Re}(\lambda))$ -plane.

It appears from the numerical experiments that the patterns we show in here occur only for our specific example, that is, $\operatorname{Spec}_p(\mathbf{M})$ when $A = D = H_0$. Recall the two-parameter eigenvalue problem:

$$\begin{pmatrix} H_0 - \alpha & \kappa B \\ \kappa B & H_0 - \beta \end{pmatrix} \begin{pmatrix} \mathbf{u} \\ \mathbf{v} \end{pmatrix} = \mathbf{0},$$
(5.2)

or equivalently

$$\mathbf{A}_{c,\kappa} \begin{pmatrix} \mathbf{u} \\ \mathbf{v} \end{pmatrix} = \begin{pmatrix} H_c & \kappa B \\ -\kappa B & -H_c \end{pmatrix} \begin{pmatrix} \mathbf{u} \\ \mathbf{v} \end{pmatrix} = \lambda \begin{pmatrix} \mathbf{u} \\ \mathbf{v} \end{pmatrix},$$

where $\kappa \in \mathbb{R}$ and $B = B^*$ with $B_{n,n} = 1$ and all other entries of B are equal to zero. Note that when $\kappa = 1$, the problem is equivalent to the pencil problem considered in [19], i.e. $\mathbf{A}_c = \mathbf{A}_{c,1}$. In the following, we present some illustrations about the non-real-eigenvalues of $\mathbf{A}_{c,\kappa}$.

When $\kappa \ge 1$, the proportion of non-real eigenvalues of $\mathbf{A}_{c,\kappa}$ tends to usually decrease with κ , whereas $\kappa \le 1$ it tends to increase with κ . We note that we show a typical picture for some $\kappa \ge 1$ in Figure 5.2(left). On the other hand, if we fix κ and change the size of the matrix n, then proportion of non-real eigenvalues becomes more stable, see Figure 5.2(right). Although the proportion of non-real eigenvalues become more stable for a specific value of c and κ as n increases, it remains difficult to estimate the range of possible values.

Another example is given in Figure 5.3 where we illustrate the eigenvalues which escape from the real line. In Figure 5.3, we take n = 10 and superimpose the real parts of the non-real eigenvalues of $A_{c,\kappa}$, with respect to c, for different values of κ between 0.001 and 5. One can see a similar pattern to Chess Board structure in this figure, however, not in the real eigenvalues but in the non-real eigenvalues.

In the top panel of Figure 5.4, the non-real eigenvalues of $A_{c,\kappa}$ is superimposed by taking *n* from 2 to 50. If one takes κ to be either closer to zero or large, then this pattern somehow gets clearer, see the bottom Figure 5.4 when $\kappa = 4$. In the next section, we will look at the escape points (i.e. collision points) in more details.



Figure 5.2: Proportion of the non-real eigenvalues of $A_{c,\kappa}$ as c goes from 0 to 2.2. Left: for n = 500, with $\kappa = 1$ (red line), $\kappa = 2$ (purple line), $\kappa = 5$ (orange line). Right: for $\kappa = 1$, with n = 500 (red line), n = 1000 (blue line).



Figure 5.3: For n = 10, this figure shows the superimposition of the non-real eigenvalues of $\mathbf{A}_{c,\kappa}$ in the $(c, \operatorname{Re}(\lambda))$ -plane, when the values of κ ranges from 0.001 to 5 with the stepsize of 0.04.

5.3 Double eigenvalues of A_c

In this section we provide illustrative numerical examples for the location and the number of double eigenvalues of the pencil A_c (or, equivalently $A_{c,1}$). Recall that the pencil A_c has



Figure 5.4: In the $(c, \operatorname{Re}(\lambda))$ -plane, the superimposition of the non-real eigenvalues of $A_{c,\kappa}$ by taking n from 2 to 50. Top: for $\kappa = 1$. Bottom: for $\kappa = 4$.

dimension 2n. Most conjectures in this section are supported by numerical calculations for n = 1, ..., 30 and for a wide range of c. As n gets bigger, it takes a relatively large amount of time to find the location of the double eigenvalues of A_c . First, we note the following:

Conjecture 5.3.1. For each given value of $n \in \mathbb{N}$ and $c \in (0,2)$ there are no non-real double eigenvalues of \mathcal{A}_c .

We provide two examples as illustrations. First, we plot all the double eigenvalues in the (c, λ) -plane for a fixed size, n = 20, in Figure 5.5. Although some double eigenvalues appear to lie on the same line, it is difficult to see a pattern. However, if one superimposes the double eigenvalues by taking the values of n from 2 to 30, then one would see that they form an interesting pattern in the (c, λ) -plane, as shown in Figure 5.6.



Figure 5.5: The double eigenvalues of A_c in the (c, λ) -plane when n = 20.

As mentioned in Lemma 3.7.3, the collisions of the eigenvalues (or, equivalently, double eigenvalues) of the pencil A_c occur when $dc/d\lambda = 0$ (or, equivalently $d\beta/d\alpha = -1$). These are the critical points along eigenvalue curves.

Assume that we are in the situation of the simple Chess Board Theorem. Recall from Chapter 3 that the real pair-eigenvalues of (5.2) when $\kappa = 1$ consists of curves $(\alpha, \beta(\alpha))$ which are continuous except at $\alpha \in \text{Spec}(H_0)$. We denote by Λ_i , i = 1, ..., n, both branches of the curve $(\alpha, \beta_i(\alpha))$, such that $\beta_i(\alpha_i \pm 0) \rightarrow \pm \infty$. A typical enumeration of the curves Λ_i can be seen in Figure 5.7 for n = 6. Note that the eigenvalue curves



Figure 5.6: The superimposition of the double eigenvalues of A_c in the (c, λ) -plane when n is between 2 and 30.



Figure 5.7: For n = 6, $\operatorname{Spec}(\mathbf{A}_{c,1})$ when $c \in (-2,2)$ in the $(\operatorname{Re}(\alpha), \operatorname{Re}(\beta))$ -plane. Black dotted lines represent $\alpha = \alpha_i \in \operatorname{Spec}(H_0)$ in the vertical direction and $\beta = \alpha_i \in \operatorname{Spec}(H_0)$ in the horizontal direction.

in Figure 3.11 look a bit different than they do in Figure 5.7. This is because we take $c \in (-2, 2)$ in Figure 5.7 whereas we considered $c \in \mathbb{R}$ in Figure 3.11.
From numerical experiments we observe the following:

Conjecture 5.3.2. Let $c \in (-2, 2)$. Then each curve Λ_i , i = 1, ..., n has exactly 2n critical points such that

$$\frac{d\beta_i(\alpha)}{d\alpha} = -1.$$

As can be seen from Figure 5.7 each curve has exactly *n* Type-A collisions and *n* Type-B collisions. We may be able to determine the location of some double eigenvalues of the pencil A_c . Indeed, we can count the number of double eigenvalues of A_c which occur at $\lambda = 0$. Recall the pencil problem

$$\mathcal{A}_{c}\mathbf{f} = \left(H_{c}^{(2n)} - \lambda S\right)\mathbf{f} = \left(H_{0}^{(2n)} + cI - \lambda S\right)\mathbf{f} = 0.$$

We observe that if $\lambda = 0$, then

$$H_0^{(2n)}\mathbf{f} = -c\mathbf{f},$$

which implies that $c = -\mu_j^{(2n)}$, and since $\mu_j^{(2n)}$ are symmetric, the double eigenvalues λ^* of \mathcal{A}_c that occur at the origin, i.e. at $\lambda = 0$, occur when

$$c = \mu_j^{(2n)} = 2\cos\frac{\pi j}{2n+1}, \qquad j = 1,\dots,2n,$$

and therefore

$$\# \bigcup_{c \in (-2,2)} \left\{ \lambda^* \in \operatorname{Spec} \left(\mathcal{A}_c \right) : \lambda^* = 0 \right\} = 2n.$$
(5.3)

One should note that the real pair-eigenvalue spectrum viewed in $(\text{Re}(\alpha), \text{Re}(\beta))$ -plane covers the spectra of the corresponding pencils \mathcal{A}_c for all real c. To find the real eigenvalue use of \mathcal{A}_{c_0} for a particular $c_0 \in \mathbb{R}$, one has to look at the intersection of the real eigenvalue curves (blue curves) with the straight line $\alpha + \beta = -2c$, see (5.1). Every line will intersect with the eigenvalue curves at most 2n points, cf. Figure 5.7. If, additionally, at each intersection point, the curve has gradient -1, then it is a double eigenvalue intersection point. Note that for some particular c's, there may be or may not be any double eigenvalues of \mathcal{A}_c since it is rare. So the number of all critical points as a union of all c's and all λ 's is the number of double eigenvalues. Therefore λ^* do not depend continuously on c and they form a discrete set. If we look at the union of all c's, then we look at the whole picture, c.f. Figure 5.5 for n = 20. On the other hand, in Figure 5.5, if one wants to look at the picture or count the double eigenvalues of \mathcal{A}_c for a given c, then one needs to take a straight vertical line in the (c, λ) -plane, and count how many of them are on the line, which will give the number of double eigenvalues.

Lemma 5.3.3. Subject to Conjecture 5.3.2, for a fixed n, the double real eigenvalues λ^* of A_c satisfy

(i)
$$\# \bigcup_{c \in (-2,2)} \{\lambda^* \in \text{Spec}(\mathcal{A}_c) : -4 \le \lambda^* \le 4\} = 2n^2.$$

(ii) $\# \bigcup_{c \in (-2,2)} \{\lambda^* \in \text{Spec}(\mathcal{A}_c) : 0 < \lambda^*\} = n(n-1).$

Proof. The statement (i) follows by multiplying the number of curves, which is n, and the number of critical points in each curve, which is 2n by Conjecture 5.3.2. To prove (ii), using (5.3) we see that there are $2n^2 - 2n$ critical points which do not lie at $\lambda = 0$. Then the statement follows from dividing by 2, as we consider the upper-half plane $\lambda > 0$.

In the next result, we conjecture that all double eigenvalues of the pencil A_c are localized to a specific area. Numerical experiments suggest that double eigenvalues of A_c occur only when $\lambda < 2 - |c|$. In fact, if the only time when the eigenvalues of A_c escape from the real line and rejoin the real line are after the collisions, then Conjecture 1.3.3 is a consequence of the following conjecture.

Conjecture 5.3.4. The double eigenvalues λ^* of the pencil A_c satisfy

(i)
$$\# \bigcup_{c \in (-2,2)} \{\lambda^* \in \text{Spec}(\mathcal{A}_c) : 0 < \lambda^* + |c| < 2\} = n(n-1),$$

(ii) $\# \bigcup_{c \in (-2,2)} \{\lambda^* \in \text{Spec}(\mathcal{A}_c) : 2 \le \lambda^* + |c| \le 4\} = 0.$

Remark 5.3.5. Note that as c increases from 0 to 2, the double eigenvalues do not overtake each other when they collide and while they jump. Thus, Conjecture 5.3.4 implies that the rate at which eigenvalues become double eigenvalues should be lower than the rate at which c decreases.

Bibliography

- Y. A. Alpin, M.-T. Chien, and L. Yeh. The numerical radius and bounds for zeros of a polynomial. *Proceedings of the American Mathematical Society*, 131(3):725–730, 2003.
- [2] F. V. Atkinson. Multiparameter spectral theory. *Bulletin of the American Mathematical Society*, 74(1):1–27, 1968.
- [3] F. V. Atkinson. *Multiparameter eigenvalue problems*, volume 1. Academic Press, New York, 1972.
- [4] F. V. Atkinson and A. B. Mingarelli. *Multiparameter eigenvalue problems: Sturm-Liouville theory*. CRC Press, Boca Raton, FL, 2011.
- [5] T. Y. Azizov and I. S. lokhvidov. *Linear operators in spaces with an indefinite metric*. John Wiley & Sons Ltd., Chichester, 1989.
- [6] F. Bagarello, J. Gazeau, F. Szafraniec, and M. Znojil. *Non-selfadjoint Operators in Quantum Physics: Mathematical Aspects*. John Wiley & Sons, 2015.
- [7] C. Bender, A. Fring, U. Günther, and H. Jones. Quantum physics with non-hermitian operators. *Journal of Physics A: Mathematical and Theoretical*, 45, 440301, 2012.
- [8] M. V. Berry. Physics of nonhermitian degeneracies. Czechoslovak Journal of Physics, 54(10):1039–1047, 2004.
- [9] M. V. Berry and M. R. Dennis. The optical singularities of birefringent dichroic chiral crystals. *Proc. R. Soc. A*, 459(2033):1261–1292, 2003.
- [10] P. Binding and P. J. Browne. Spectral properties of two-parameter eigenvalue problems. *Proc. Roy. Soc. Edinburgh*, 89A(1-2):157–173, 1981.
- [11] P. Binding and P. J. Browne. Spectral properties of two-parameter eigenvalue problems. II. *Proc. Roy. Soc. Edinburgh*, 106A, 106A(1-2):39–51, 1987.

- [12] P. Binding and P. J. Browne. Applications of two parameter spectral theory to symmetric generalised eigenvalue problems. *Applicable Analysis*, 29(1-2):107–142, 1988.
- [13] M. S. Birman and M. Z. Solomjak. Spectral theory of selfadjoint operators in Hilbert space. Mathematics and its Applications (Soviet Series). D. Reidel Publishing Co., Dordrecht, 1987.
- [14] J. Bognár. Indefinite inner product spaces. Springer-Verlag, New York, 1974.
- [15] S. Bora and V. Mehrmann. Linear perturbation theory for structured matrix pencils arising in control theory. SIAM Journal on Matrix Analysis and Applications, 28(1):148–169, 2006.
- [16] N. Bourbaki. *Elements of the history of mathematics*. Springer-Verlag, Berlin, 1994. Translated from the 1984 French original by John Meldrum.
- [17] A. Dall'Acqua, D. Mugnolo, and M. Schelling. A new gershgorin-type result for the localisation of the spectrum of matrices. *Mathematische Nachrichten*, 288(17-18):1981–1994, 2015.
- [18] L. Dashnits and M. Zusmanovich. Criteria for matrix regularity and spectrum localization. USSR Computational Mathematics and Mathematical Physics, 10(5):34 – 41, 1970.
- [19] E. B. Davies and M. Levitin. Spectra of a class of non-self-adjoint matrices. *Linear Algebra and its Applications*, 448:55–84, 2014.
- [20] J. Dieudonné. *History of functional analysis*, volume 49 of *North-Holland Mathematics Studies*. North-Holland Publishing Co., Amsterdam-New York, 1981.
- [21] N. Dunford and J. T. Schwartz. *Linear operators. Part II: Spectral theory. Self adjoint operators in Hilbert space.* New York : Interscience Publishers, 1963.
- [22] D. M. Elton, M. Levitin, and I. Polterovich. Eigenvalues of a one-dimensional Dirac operator pencil. Ann. Henri Poincaré, 15(12):2321–2377, 2014.
- [23] D. G. Feingold and R. S. Varga. Block diagonally dominant matrices and generalizations of the Gerschgorin circle theorem. *Pacific Journal of Mathematics*, 12(4):1241– 1250, 1962.

- [24] N. H. Fletcher and T. D. Rossing. *The physics of musical instruments*. Springer-Verlag, New York, second edition, 1998.
- [25] S. Geršgorin. Über die abgrenzung der eigenwerte einer matrix. *Izv. Akad. Nauk.* USSR Otd.Fiz.-Mat.Nauk, 6(6):749–754, 1931.
- [26] M. I. Gil'. Operator functions and localization of spectra, volume 1830 of Lecture Notes in Mathematics. Springer-Verlag, Berlin, 2003.
- [27] I. Gohberg, P. Lancaster, and L. Rodman. *Matrices and indefinite scalar products*, volume 8 of *Operator Theory: Advances and Applications*. Birkhäuser, Basel, 1983.
- [28] R. Hartmann, N. Robinson, and M. Portnoi. Smooth electron waveguides in graphene. *Physical Review B*, 81(24):245431, 2010.
- [29] G. Herzog and C. Schmoeger. The Brauer-Ostrowski theorem for matrices of operators. Integral Equations and Operator Theory, 57(4):513–520, 2007.
- [30] T. Hey and P. Walters. *The New Quantum Universe*. Cambridge University Press, 2003.
- [31] R. A. Horn and C. R. Johnson. *Matrix analysis*. Cambridge University Press, second edition, 2013.
- [32] Y. V. Hote, D. R. Choudhury, and J. R. P. Gupta. Gerschgorin Theorem and its Applications in Control System Problems. In 2006 IEEE International Conference on Industrial Technology, pages 2438–2443. IEEE, 2006.
- [33] V. E. Howle and L. N. Trefethen. Eigenvalues and musical instruments. *Journal of Computational and Applied Mathematics*, 135(1):23–40, 2001.
- [34] I. S. lohvidov, M. G. Krein, and H. Langer. Introduction to the spectral theory of operators in spaces with an indefinite metric, volume 9. Akademie-Verlag, Berlin, 1982.
- [35] T. Kailath. *Linear systems*, volume 156. Prentice-Hall, Englewood Cliffs, N.J., 1980.
- [36] N. D. Kopachevsky and S. G. Krein. Operator Approach to Linear Problems of Hydrodynamics. Vol. 2. Nonself-adjoint Problems for Viscous Fluids, volume 146. Birkhäuser Verlag, Basel, 2003.

- [37] V. Kostić. On general principles of eigenvalue localizations via diagonal dominance. *Advances in Computational Mathematics*, 41(1):55–75, 2015.
- [38] V. R. Kostić, A. Miedlar, and L. Cvetković. An algorithm for computing minimal geršgorin sets. *Numerical Linear Algebra with Applications*, 23(2):272–290, 2016.
- [39] L. D. Landau and E. M. Lifshitz. *Quantum mechanics: non-relativistic theory*. Pergamon Press Ltd., 3rd edition, 1977. Translated from the Russian by J. B. Sykes and J. S. Bell.
- [40] H. Langer. Spectral functions of definitizable operators in Krein spaces. In D. Butković, H. Kraljević, and S. Kurepa, editors, *Functional analysis (Dubrovnik, 1981)*, Lecture Notes in Mathematics, vol. 948, pages 1–46. Springer, Berlin, Heidelberg, 1982.
- [41] M. Levitin and H. M. Oztürk. A two-parameter eigenvalue problem for a class of blockoperator matrices. In A. Böttcher, D. Potts, P. Stollmann, and D. Wenzel, editors, *The Diversity and Beauty of Applied Operator Theory*, volume 268, pages 367–380. Birkhäuser, Cham, 2018.
- [42] M. Levitin and M. Seri. Accumulation of complex eigenvalues of an indefinite Sturm-Liouville operator with a shifted Coulomb potential. *Operators and Matrices*, 10(1):223–245, 2016.
- [43] C.-K. Li and L. Rodman. Numerical range of matrix polynomials. *SIAM Journal on Matrix Analysis and Applications*, 15(4):1256–1265, 1994.
- [44] S. Li, C. Li, and Y. Li. Exclusion sets for eigenvalues of matrices. *eprint arXiv:1705.01758v2*, 2017.
- [45] A. S. Markus. Introduction to the spectral theory of polynomial operator pencils, volume 71. American Mathematical Society, Providence, RI, 1988.
- [46] M. Marletta and C. Tretter. Spectral bounds and basis results for non-self-adjoint pencils, with an application to Hagen-Poiseuille flow. *Journal of Functional Analysis*, 264(9):2136–2176, 2013.
- [47] J. Maroulas and P. Psarrakos. The boundary of the numerical range of matrix polynomials. *Linear Algebra and its Applications*, 267:101–111, 1997.

- [48] J. C. Mason and D. C. Handscomb. *Chebyshev polynomials*. Chapman & Hall/CRC, 2003.
- [49] A. Melman. Spectral inclusion sets for structured matrices. Linear Algebra and its Applications, 431(5-7):633–656, 2009.
- [50] A. Melman. Generalizations of gershgorin disks and polynomial zeros. *Proceedings* of the American Mathematical Society, 138(7):2349–2364, 2010.
- [51] A. Melman. Gershgorin disk fragments. *Mathematics Magazine*, 83(2):123–129, 2010.
- [52] A. Melman. Modified Gershgorin disks for companion matrices. *SIAM Review*, 54(2):355–373, 2012.
- [53] A. Melman. Ovals of cassini for toeplitz matrices. *Linear and Multilinear Algebra*, 60(2):189–199, 2012.
- [54] A. Melman. Comment on a result by Alpin, Chien, and Yeh. *Proceedings of the American Mathematical Society*, 141(3):775–777, 2013.
- [55] A. Melman. A single oval of Cassini for the zeros of a polynomial. *Linear and Multilinear Algebra*, 61(2):183–195, 2013.
- [56] N. Moiseyev. *Non-Hermitian Quantum Mechanics*. Cambridge University Press, 2011.
- [57] T. K. Moon and W. C. Stirling. Mathematical methods and algorithms for signal processing, volume 1. New Jersey : Prentice Hall, 2000.
- [58] Y. Nakatsukasa. Gerschgorin's theorem for generalized eigenvalue problems in the Euclidean metric. *Mathematics of Computation*, 80(276):2127–2142, 2011.
- [59] D. R. Nelson and N. M. Shnerb. Non-Hermitian localization and population biology. *Phys. Rev. E*, 58:1383–1403, 1998.
- [60] B. N. Parlett and H. C. Chen. Use of indefinite pencils for computing damped natural modes. *Linear Algebra and its Applications*, 140:53–88, 1990.
- [61] P. J. Psarrakos. Numerical range of linear pencils. *Linear Algebra and its Applications*, 317(1-3):127–141, 2000.

- [62] E. Recami, V. S. Olkhovsky, and S. P. Maydanyuk. On non-self-adjoint operators for observables in quantum mechanics and quantum field theory. *International Journal* of Modern Physics A, 25(9):1785–1818, 2010.
- [63] L. Rodman. An introduction to operator polynomials. Operator Theory: Advances and Applications Vol. 38. Birkhäuser Verlag, Basel, 1989.
- [64] D. Rosinová, N. Thuan, V. Veselỳ, and L. Marko. Robust decentralized controller design: Subsystem approach. *Journal of Electrical Engineering*, 63(1):28–34, 2012.
- [65] H. N. Salas. Gershgorin's theorem for matrices of operators. *Linear Algebra and its Applications*, 291(1-3):15–36, 1999.
- [66] B. Simon. Ratio asymptotics and weak asymptotic measures for orthogonal polynomials on the real line. *Journal of Approximation Theory*, 126(2):198–217, 2004.
- [67] B. D. Sleeman. Multiparameter spectral theory in Hilbert space. J. Math. Anal. Appl., 65(3):511–530, 1978.
- [68] Y. Smagina, O. Nekhamkina, and M. Sheintuch. Stabilization of Fronts in a Reaction
 Diffusion System: Application of the Gershgorin Theorem. *Industrial and Engineering Chemistry Research*, 41(8):2023–2032, 2002.
- [69] L. A. Steen. Highlights in the history of spectral theory. *The American Mathematical Monthly*, 80(4):359–381, 1973.
- [70] G. W. Stewart. Gershgorin theory for the generalized eigenvalue problem $Ax = \lambda Bx$. Mathematics of Computation, 29:600–606, 1975.
- [71] G. W. Stewart and J. G. Sun. *Matrix perturbation theory*. Computer Science and Scientific Computing. Academic Press, Inc., Boston, MA, 1990.
- [72] F. Tisseur and N. J. Higham. Structured pseudospectra for polynomial eigenvalue problems, with applications. *SIAM Journal on Matrix Analysis and Applications*, 23(1):187–208, 2001.
- [73] F. Tisseur and K. Meerbergen. The quadratic eigenvalue problem. *SIAM Review*, 43(2):235–286, 2001.
- [74] L. N. Trefethen and M. Embree. Spectra and Pseudospectra: The Behavior of Nonnormal Matrices and Operators. Princeton University Press, 2005.

- [75] C. Tretter. *Spectral theory of block operator matrices and applications*. Imperial College Press, London, 2008.
- [76] C. Tretter and M. Wagenhofer. The block numerical range of an $n \times n$ block operator matrix. *SIAM Journal on Matrix Analysis and Applications*, 24(4):1003–1017, 2003.
- [77] W. Van Assche. Orthogonal polynomials in the complex plane and on the real line. In Special functions, q-series and related topics (M.E.H. Ismail et al., eds.), volume 14 of Fields Inst. Commun., pages 211–245. Amer. Math. Soc., Providence, RI, 1997.
- [78] R. S. Varga. Minimal Gerschgorin sets. *Pacific Journal of Mathematics*, 15(2):719– 729, 1965.
- [79] R. S. Varga. Geršgorin and his circles, volume 36 of Springer Series in Computational Mathematics. Springer-Verlag, Berlin, 2004.
- [80] R. S. Varga, L. Cvetkovic, and V. Kostic. Approximation of the minimal Geršgorin set of a square complex matrix. *Electronic Transactions on Numerical Analysis*, 30:398– 405, 2008.
- [81] J. H. Wilkinson. *The algebraic eigenvalue problem*, volume 662. Clarendon Press, Oxford, 1965.